UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

Mark One)	
I	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the Quarterly Period Ended April 3, 2011
	OR
0	TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the transition period from to

LSI CORPORATION

Commission File Number: 1-10317

(Exact name of registrant as specified in its charter)

Delaware (State of Incorporation)

94-2712976 (I.R.S. Employer Identification Number)

1621 Barber Lane Milpitas, California 95035 (Address of principal executive offices) (Zip code)

(408) 433-8000 (Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \square No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer ☑ Accelerated filer o Non-accelerated filer o Smaller reporting company o (Do not check if a smaller reporting company.)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🗵

As of May 6, 2011, there were 605,650,469 shares of the registrant's Common Stock, \$.01 par value, outstanding.

LSI CORPORATION FORM 10-Q For the Quarter Ended April 3, 2011 INDEX

	Page No.
PART I. FINANCIAL INFORMATION	
<u>Item 1. Financial Statements (Unaudited)</u>	3
Condensed Consolidated Balance Sheets as of April 3, 2011 and December 31, 2010	3
Condensed Consolidated Statements of Operations for the three months ended April 3, 2011 and April 4, 2010	4
Condensed Consolidated Statements of Cash Flows for the three months ended April 3, 2011 and April 4, 2010	5
Notes to Condensed Consolidated Financial Statements	6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	19
Item 3. Quantitative and Qualitative Disclosures about Market Risk	26
Item 4. Controls and Procedures	26
PART II. OTHER INFORMATION	
Item 1. Legal Proceedings	26
Item 1A. Risk Factors	26
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	28
Item 6. Exhibits	28
<u>Signatures</u>	29
<u>EX-2.1</u>	
<u>EX-31.1</u>	
<u>EX-31.2</u>	
<u>EX-32.1</u>	
<u>EX-32.2</u>	
EX-101 INSTANCE DOCUMENT	
EX-101 SCHEMA DOCUMENT	
EX-101 CALCULATION LINKBASE DOCUMENT	
EX-101 LABELS LINKBASE DOCUMENT	
EX-101 PRESENTATION LINKBASE DOCUMENT	

FORWARD-LOOKING STATEMENTS

This Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. The words "estimate," "plan," "intend," "expect," "anticipate," "believe" and similar words are intended to identify forward-looking statements. Although we believe our expectations are based on reasonable assumptions, our actual results could differ materially from those projected in the forward-looking statements. We have described in Part II, "Item 1A. Risk Factors" a number of factors that could cause our actual results to differ from our projections or estimates. Except where otherwise indicated, the statements made in this report are made as of the date we filed this report with the Securities and Exchange Commission and should not be relied upon as of any subsequent date. We expressly disclaim any obligation to update the information in this report, except as may otherwise be required by law.

PART I — FINANCIAL INFORMATION

Item 1. Financial Statements

LSI CORPORATION CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, except per share amounts) (Unaudited)

	April 3, 2011	December 31, 2010
ASSETS		
Cash and cash equivalents	\$ 526,528	\$ 521,786
Short-term investments	155,786	154,880
Accounts receivable, less allowances of \$7,418 and \$9,701, respectively	286,133	326,604
Inventories	155,023	186,772
Prepaid expenses and other current assets	68,369	73,314
Assets held for sale	236,296	464
Total current assets	1,428,135	1,263,820
Property and equipment, net	188,002	223,181
Identified intangible assets, net	519,588	561,137
Goodwill	72,377	188,698
Other assets	146,772	188,076
Total assets	\$ 2,354,874	\$ 2,424,912
LIABILITIES AND STOCKHOLDERS' EQUITY		·
Accounts payable	\$ 198,836	\$ 173,919
Accrued salaries, wages and benefits	104,198	126,307
Other accrued liabilities	181,217	184,402
Total current liabilities	484,251	484,628
Pension and post-retirement benefit obligations	454,750	463,119
Income taxes payable — non-current	81,478	85,717
Other non-current liabilities	74,746	73,946
Total liabilities	1,095,225	1,107,410
Commitments and contingencies (Note 13)		
Stockholders' equity:		
Preferred stock, \$.01 par value: 2,000 shares authorized; none outstanding	_	_
Common stock, \$.01 par value: 1,300,000 shares authorized; 606,150 and 615,191 shares outstanding, respectively	6,062	6,152
Additional paid-in capital	5,926,306	5,998,137
Accumulated deficit	(4,358,368)	(4,368,522)
Accumulated other comprehensive loss	(314,351)	(318,265)
Total stockholders' equity	1,259,649	1,317,502
Total liabilities and stockholders' equity	\$ 2,354,874	\$ 2,424,912

The accompanying notes are an integral part of these Condensed Consolidated Financial Statements.

LSI CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (In thousands, except per share amounts) (Unaudited)

	Three M April 3, 2011	onths Ended April 4, 2010
Revenues	\$ 473,264	\$ 472,672
Cost of revenues	249,090	257,878
Gross profit	224,174	214,794
Research and development	142,347	138,862
Selling, general and administrative	68,867	70,365
Restructuring of operations and other items, net	2,806	1,620
Income from operations	10,154	3,947
Interest expense	_	(3,894)
Interest income and other, net	4,288	(8,807)
Income/(loss) from continuing operation before income taxes	14,442	(8,754)
Benefit from income taxes	(4,104)	(23,102)
Income from continuing operations	18,546	14,348
(Loss)/income from discontinued operations, net of tax	(8,392)	8,172
Net income	\$ 10,154	\$ 22,520
Basic income/(loss) per share:		
Income from continuing operations	\$ 0.03	\$ 0.02
(Loss)/income from discontinued operations	\$ (0.01)	\$ 0.01
Net income	\$ 0.02	\$ 0.03
Diluted income/(loss) per share:		
Income from continuing operations	\$ 0.03	\$ 0.02
(Loss)/income from discontinued operations	\$ (0.01)	\$ 0.01
Net income	\$ 0.02	\$ 0.03
Shares used in computing per share amounts:		
Basic	615,450	656,528
Diluted	629,733	664,315

The accompanying notes are an integral part of these Condensed Consolidated Financial Statements.

LSI CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands) (Unaudited)

		onths Ended
Operating activities:	<u>April 3, 2011</u>	<u>April 4, 2010</u>
Net income	\$ 10,154	\$ 22,520
Adjustments:	4 = 3 ,= 3 .	4,=_,
Depreciation and amortization	56,007	67,017
Stock-based compensation expense	13,986	16,431
Non-cash restructuring of operations and other items, net	10,824	(10)
Write-down of investments	_	11,600
(Gain)/loss on sale of property and equipment	(239)	3
Unrealized foreign exchange loss/(gain)	1,379	(2,215)
Deferred taxes	(43)	98
Changes in assets and liabilities:		
Accounts receivable, net	40,471	40,396
Inventories	(12,651)	(16,441)
Prepaid expenses, assets held for sale and other assets	(1,066)	(8,095)
Accounts payable	24,273	(8,547)
Accrued and other liabilities	(35,066)	(16,979)
Net cash provided by operating activities	108,029	105,778
Investing activities:		·
Purchases of debt securities available-for-sale	(15,530)	_
Proceeds from maturities and sales of debt securities available-for-sale	12,958	11,254
Purchases of property, equipment and software	(21,542)	(27,276)
Proceeds from sale of property and equipment	310	22
Net cash used in investing activities	(23,804)	(16,000)
Financing activities:		
Issuances of common stock	17,319	3,635
Purchase of common stock under repurchase programs	(96,791)	(26,208)
Net cash used in financing activities	(79,472)	(22,573)
Effect of exchange rate changes on cash and cash equivalents	(11)	(2,117)
Net change in cash and cash equivalents	4,742	65,088
Cash and cash equivalents at beginning of period	521,786	778,291
Cash and cash equivalents at end of period	\$ 526,528	\$ 843,379
1	+,	+

The accompanying notes are an integral part of these Condensed Consolidated Financial Statements.

LSI CORPORATION NOTES TO UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Note 1 — Basis of Presentation

For financial reporting purposes, LSI Corporation ("LSI" or the "Company") reports on a 13- or 14-week quarter with the year ending December 31. The first quarter of 2011 and 2010 consisted of approximately 13 weeks each and ended on April 3, 2011 and on April 4, 2010, respectively. The results of operations for the quarter ended April 3, 2011 are not necessarily indicative of the results to be expected for the full year.

On March 9, 2011, the Company entered into a definitive agreement to sell its external storage systems business to NetApp, Inc. ("NetApp") for \$480.0 million in cash. The strategic decision to divest the external storage systems business was based on the Company's expectation that long-term shareholder value can be maximized by becoming a pure-play semiconductor company. Under the terms of the agreement, NetApp will purchase substantially all the assets of the Company's external storage systems business, which develops and delivers external storage systems products and technology to a wide range of partners that provide storage solutions to end customers. Most of LSI external storage systems employees are expected to join NetApp. In accordance with the applicable accounting guidance for the disposal of long-lived assets, the results of the external storage systems business are presented as discontinued operations and, as such, have been excluded from both continuing operations and segment results for all periods presented. In the first quarter of 2011, the Company operates in one reportable segment. The external storage systems business was part of the Storage Systems segment. The results of the redundant array of independent disks ("RAID") adapter business, which were formerly included in the Storage Systems segment, are now included in the Company's remaining reportable segment. Additionally, the assets of the external storage systems business as of April 3, 2011 are presented as held for sale in the condensed consolidated balance sheets.

The preparation of financial statements in conformity with generally accepted accounting principles in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting periods. Actual results could differ significantly from these estimates.

In management's opinion, the accompanying unaudited condensed consolidated financial statements contain all normal recurring adjustments necessary for a fair presentation of the Company's financial position, results of operations, and cash flows for the interim periods presented. While the Company believes that the disclosures are adequate to make the information not misleading, these financial statements should be read in conjunction with the audited consolidated financial statements and accompanying notes included in the Company's Annual Report on Form 10-K for the year ended December 31, 2010.

Recent Accounting Pronouncements

Pronouncements adopted during the three months ended April 3, 2011:

In October 2009, the Financial Accounting Standards Board ("FASB") amended revenue recognition guidance on multiple-deliverable arrangements to address how to separate deliverables and how to measure and allocate arrangement consideration. The new guidance requires the use of management's best estimate of selling price for the deliverables in an arrangement when a vendor does not have specific objective evidence of selling price or third party evidence of selling price. In addition, excluding specific software revenue guidance, the residual method of allocating arrangement consideration is no longer permitted, and an entity is required to allocate arrangement consideration using the relative selling price method. This guidance also expands the disclosure requirements to include both quantitative and qualitative information. The Company adopted this guidance in the first quarter of 2011. The adoption did not impact the Company's results of operations or financial position.

In October 2009, the FASB issued guidance to clarify that tangible products containing software components and non-software components that function together to deliver a product's essential functionality will be considered non-software deliverables and will be scoped out of the software revenue recognition guidance. The Company adopted this guidance in the first quarter of 2011. The adoption did not impact the Company's results of operations or financial position.

In December 2010, the FASB issued guidance to clarify that, when presenting comparative financial statements for business combinations that occurred during the current year, a public entity should disclose revenue and earnings of the combined entity as

though the business combinations had occurred as of the beginning of the comparable prior annual reporting period. The update also expands the supplemental pro forma disclosures to include a description of the nature and amount of material, nonrecurring pro forma adjustments directly attributable to the business combination included in the reported pro forma revenue and earnings. The Company adopted this guidance in the first quarter of 2011. The adoption did not impact the Company's results of operations or financial position.

Note 2 — Stock-Based Compensation and Common Stock

Stock-Based Compensation Expense

The following table summarizes stock-based compensation expense, net of estimated forfeitures, related to the Company's stock options, Employee Stock Purchase Plan ("ESPP") and restricted stock unit awards.

	Three Months Ended			ed
Stock-Based Compensation Expense Included In:	April 3, 2011 Apri		ril 4, 2010	
		(In thousands)		
Cost of revenues	\$	1,813	\$	1,416
Research and development		6,223		6,020
Selling, general and administrative		5,631		5,687
Total stock-based compensation expense	\$	13,667	\$	13,123

Stock Options:

The fair value of each option grant is estimated as of the date of grant using a reduced-form calibrated binomial lattice model (the "lattice model"). The following table summarizes the weighted-average assumptions that the Company applied in the lattice model:

		Three Months Ended		
	April 3, 2011		1 April	
Estimated grant date fair value per share	\$	2.08	\$	1.97
Expected life (years)		4.45		4.28
Risk-free interest rate		2%		2%
Volatility		47%		51%

The following table summarizes changes in stock options outstanding:

	Number of Shares (In thousands)	Weighted-Average Exercise Price Per Share		Weighted-Average Remaining <u>Contractual Term</u> (In years)	Aggregate Intrinsic Value (In thousands)
Options outstanding at December 31, 2010	69,997	\$	6.99	—	_
Options granted	7,433		6.18	_	_
Options exercised	(2,587)		4.79	_	_
Options canceled	(3,725)		18.01	_	_
Options outstanding at April 3, 2011	71,118	\$	6.41	3.74	\$ 82,958
Options exercisable at April 3, 2011	45,093	\$	7.10	2.88	\$ 42,216

Employee Stock Purchase Plan:

Compensation expense for the Company's ESPP is calculated using the fair value of the employees' purchase rights under the Black-Scholes model. Under the ESPP, rights to purchase shares are granted during the second and fourth quarters of each year. No shares related to the ESPP were issued during the three months ended April 3, 2011 or April 4, 2010.

Restricted Stock Awards:

The cost of restricted stock unit awards is determined using the fair value of the Company's common stock on the date of grant.

Service-based:

The vesting requirements for service-based restricted stock units are determined at the time of grant and require that the employee remain employed by the Company for a specified period of time. As of April 3, 2011, the total unrecognized compensation expense related to these restricted stock units, net of estimated forfeitures, was \$57.6 million and is expected to be recognized over the next 3.5 years on a weighted-average basis. The fair value of the shares that were issued upon the vesting of service-based restricted stock units during the three months ended April 3, 2011 was \$10.3 million.

The following table summarizes changes in service-based restricted stock units outstanding:

	Number of Units (In thousands)
Unvested service-based restricted stock units at December 31, 2010	6,713
Granted	6,297
Vested	(1,640)
Forfeited	(212)
Unvested service-based restricted stock units at April 3, 2011	11,158

Performance-based:

The vesting of performance-based restricted stock units is contingent upon the Company meeting specified performance criteria and requires that the employee remain employed by the Company for a specified period of time. As of April 3, 2011, the total unrecognized compensation expense related to performance-based restricted stock units was \$18.9 million and, if the contingencies are fully met, is expected to be recognized over the next 1 to 3 years.

The following table summarizes changes in performance-based restricted stock units outstanding:

	Number of Units
	(In thousands)
Unvested performance-based restricted stock units at December 31, 2010	2,277
Granted	3,476
Vested	(796)
Forfeited	(54)
Unvested performance-based restricted stock units at April 3, 2011	4,903

Common Stock

Stock Repurchase Program:

On March 9, 2011, the Company announced that its Board of Directors had authorized a new stock repurchase program of up to \$750.0 million of its common stock. Purchases under this program are expected to be funded from the proceeds of the sale of the external storage systems business, available cash and short-term investments. The Company repurchased 14.7 million shares for \$96.8 million under this program during the three months ended April 3, 2011. The repurchased shares were retired immediately after the repurchases were completed. Retirement of the repurchased shares is recorded as a reduction of common stock and additional paid-in capital.

Note 3 — Restructuring, Asset Impairment Charges and Other Items

The following table summarizes items included in restructuring of operations and other items, net from continuing operations:

		Three Months Ended			
	Apr	April 3, 2011		April 4, 2010	
		(In thou	sands)		
Lease and contract terminations	\$	1,688(a)	\$	846	
Employee severance and benefits		1,643(b)		525	
Total restructuring expenses		3,331		1,371	
Other items		(525)		249	
Total	\$	2,806	\$	1,620	

- (a) The amount primarily relates to changes in estimates and changes in time value of accruals for previously accrued facility lease exit costs.
- (b) The amount primarily relates to restructuring actions taken during the first quarter of 2011 as the Company continues to streamline operations.

In connection with the strategic decision to divest the external storage systems business, the Company initiated certain restructuring actions. The results of those actions are included in discontinued operations and are summarized below:

	7	Three Months Ended		
	April 3, 2	<u>A</u>	pril 4, 2010	
		(In thousands)		
Lease and contract terminations	\$ 1,7	711 \$		
Employee severance and benefits	11,0)20(a)		
Asset impairment and other exit charges	11,0	080(b)	_	
Total	\$ 23,8		_	

- (a) The amount includes severance accruals for the restructuring actions taken related to the decision to sell the external storage systems business.
- (b) The amount includes \$9.0 million and \$1.9 million for the write-down of certain identified intangible assets and certain software and hardware assets, respectively, related to the decision to sell the external storage systems business.

The following table summarizes the significant activity within, and components of, the Company's restructuring obligations:

	Write-downs ther Exit Costs	and Contract minations (In thousands)	S	mployee everance d Benefits	Total
Beginning balance at December 31, 2010	\$ _	\$ 20,905	\$	4,951	\$ 25,856
Expense	11,080	3,399		12,663	27,142
Utilized	(11,080)	(4,467)(a)		(7,473)(a)	(23,020)
Ending balance at April 3, 2011	\$ 	\$ 19,837(b)	\$	10,141(b)	\$ 29,978

- (a) The amounts utilized represent cash payments.
- (b) The balance remaining for lease and contract terminations is expected to be paid during the remaining terms of the leases, which extend through 2013. The majority of the balance remaining for severance is expected to be paid by the third quarter of 2011.

Note 4 — Benefit Obligations

The Company has pension plans covering substantially all former Agere Systems Inc. ("Agere") U.S. employees, excluding management employees hired after June 30, 2003. Retirement benefits are offered under defined benefit pension plans, which include a management plan and a represented plan. The payments under the management plan are based on an adjusted career-average-pay formula or a cash-balance program. The cash-balance program provides for annual company contributions based on a participant's age, compensation and interest on existing balances. It covers employees of certain companies acquired by Agere since 1996 and management employees hired after January 1, 1999 and before July 1, 2003. The payments under the represented plan are based on a dollar-per-month formula. Since February 2009, there have been no active participants under the represented plan. The Company also has a non-qualified supplemental pension plan in the U.S. that principally provides benefits based on compensation in excess of amounts that can be considered under a tax qualified plan. The Company also provides post-retirement life insurance coverage under a group life insurance plan for former Agere employees excluding participants in the cash-balance program and management employees hired after June 30, 2003. The Company also has pension plans covering certain international employees.

Effective April 6, 2009, the Company froze the U.S. management defined benefit pension plan. Participants in the adjusted career-average-pay program will not earn any future service accruals after that date. Participants in the cash-balance program will not earn any future service accruals, but will continue to earn 4% interest per year on their cash-balance accounts.

The following table summarizes the components of the net periodic benefit cost/(credit):

	Three Months Ended						
	Apr	il 3, 2011	April 4, 2010				
	Pension Post-retirement Benefits Benefits					Post-retirement Benefits	
Service cost	\$ 134	\$ 17	\$ 118	\$ 20			
Interest cost	16,850	625	17,617	608			
Expected return on plan assets	(16,998)	(1,032)	(17,823)	(1,148)			
Amortization of transition asset	(5)	_	_	_			
Amortization of prior service cost	10	_	10	_			
Amortization of net actuarial loss	1,681	92	547				
Total benefit cost/(credit)	\$ 1,672	\$ (298)	\$ 469	\$ (520)			

During the three months ended April 3, 2011, the Company contributed \$9.0 million to its pension plans. The Company expects to contribute an additional \$56.2 million to its pension plans for the remainder of 2011. The Company does not expect to contribute to its post-retirement benefit plan in 2011.

Note 5 — Inventories

Inventories were comprised of the following:

	April 3, 2011	December 31, 2010
	(In thou	ısands)
Raw materials	\$ 31,706	\$ 30,691
Work-in-process	45,930	33,513
Finished goods	77,387	122,568
Total inventories	\$ 155,023	\$ 186,772

Note 6 — Cash Equivalents and Investments

The following tables summarize the Company's cash equivalents and investments measured at fair value:

		air Value Measurement		
	Level 1	<u>Level 2</u> (In thousa	Level 3_ unds)	<u>Total</u>
Cash equivalents:		(III IIIOIIO		
Money-market funds	\$402,953(a)	\$ —	\$ —	\$402,953
Available-for-sale debt securities:				
Asset-backed and mortgage-backed securities	\$ —	\$ 114,312(b)	\$ —	\$ 114,312
U.S. government and agency securities	1,497(a)	25,346(b)	_	26,843
Corporate debt securities		14,631(b)		14,631
Total short-term investments	\$ 1,497	\$154,289	<u> </u>	\$155,786
Long-term investments in equity securities:				
Marketable available-for-sale equity securities	\$ 1,802(c)	\$ —	\$ —	\$ 1,802
	Fair Level 1	· Value Measurements a Level 2	s of December 31, 2010 Level 3) Total
	Level 1	Level 2 (In thousa		10tai
Cash equivalents:		•	·	
Money-market funds	\$378,382(a)	\$ —	\$ —	\$378,382
U.S. government and agency securities	2,000(a)		<u></u>	2,000
Total cash equivalents	\$380,382	<u> </u>	<u>\$</u>	\$380,382
Available-for-sale debt securities:				
Asset-backed and mortgage-backed securities	\$ —	\$ 116,552(b)	\$ —	\$ 116,552
U.S. government and agency securities	1,496(a)	24,502(b)	_	25,998
Corporate debt securities		12,330(b)		12,330
Total short-term investments	\$ 1,496	\$153,384	<u> </u>	\$154,880
Long-term investments in equity securities:				
Marketable available-for-sale equity securities	\$ 1,681(c)	\$ —	\$ —	\$ 1,681
10				

- The fair value of money-market funds is determined using unadjusted prices in active markets. The fair value of U.S. Treasury securities is determined using quoted prices in active markets.
- The fair value of the short-term investments in debt securities is determined using the market approach and the income approach. These investments are traded less frequently than Level 1 securities and are valued using inputs that include quoted prices for similar assets in active markets and inputs other than quoted prices that are observable for the asset, such as interest rates, yield curves, prepayment speeds, collateral performance, broker/dealer quotes and indices that are observable at commonly quoted intervals.
- The fair value of marketable equity securities is determined using quoted market prices in active markets. These amounts are included within other assets in the condensed consolidated balance sheets.

Investments in Non-Marketable Securities

The Company does not estimate the fair value of non-marketable securities unless there are identified events or changes in circumstances that may have a significant adverse effect on the investment. There were no non-marketable securities fair-valued during the three months ended April 3, 2011. The following table summarizes certain non-marketable securities measured and recorded at fair value on a non-recurring basis during the three months ended April 4, 2010:

	Carrying	Carrying Value Fair Value Measurements				Los	ses for the			
	as (as of		During the Three Months Ended April 4, 2010)10	Three I	Months Ended
	April 4,	April 4, 2010		Level 1		vel 2	Level 3		Ap	ril 4, 2010
				,	(In tho	usands)				
Non-marketable securities	\$	1,900	\$	_	\$	_	\$	1,900	\$	11,600

As of April 3, 2011 and December 31, 2010, the aggregate carrying value of the Company's non-marketable securities was \$39.9 million. There were no sales of non-marketable securities for the three months ended April 3, 2011 and April 4, 2010.

Investments in Available-for-Sale Securities

The following tables summarize the Company's available-for-sale securities:

	April 3, 2011						
	Amortized Cost	Gross Unrealized Gain (In t		Unrealized Loss*	Fair Value		
Short-term debt securities:		,	ĺ				
Asset-backed and mortgage-backed securities	\$105,348	\$ 9,164	\$	(200)	\$ 114,312		
U.S. government and agency securities	26,414	548		(119)	26,843		
Corporate debt securities	14,530	171		(70)	14,631		
Total short-term debt securities	\$146,292	\$ 9,883	\$	(389)	\$155,786		
					 -		
Long-term marketable equity securities	\$ 852	\$ 989	\$	(39)	\$ 1,802		

As of April 3, 2011, there were 57 investments in an unrealized loss position.

	December 31, 2010					
	Amortized Cost	Gross Unrealized Gain (In thou	Gross Unrealized Loss usands)	Fair Value		
Short-term debt securities:		(,			
Asset-backed and mortgage-backed securities	\$107,891	\$ 9,012	\$ (351)	\$ 116,552		
U.S. government and agency securities	25,313	812	(127)	25,998		
Corporate debt securities	12,226	176	(72)	12,330		
Total short-term debt securities	\$145,430	\$ 10,000	\$ (550)	\$154,880		
Long-term marketable equity securities	\$ 852	\$ 868	\$ (39)	\$ 1,681		
	11					

The following tables summarize the gross unrealized losses and fair values of the Company's short-term investments that have been in a continuous unrealized loss position for less than and greater than 12 months, aggregated by investment category:

		April 3, 2011						
	Less th	an 12 Months	Greater	Greater than 12 Month				
	Fair Value	Unrealized Losses	Fair Value thousands)	<u>Unrea</u>	lized Losses			
Asset-backed and mortgage-backed securities	\$ 4,781	\$ (77)	\$ 1,274	\$	(123)			
U.S. government and agency securities	13,535	(119)	_					
Corporate debt securities	7,279	(70)	_		_			
Total	\$ 25,595	\$ (266)	\$ 1,274	\$	(123)			
		Decer	nber 31, 2010					
		an 12 Months	<u>Greater</u> Fair Value	than 12 Mo				
	<u>Fair Value</u>	Fair Value Unrealized Losses (In the			lized Losses			
Asset-backed and mortgage-backed securities	\$ 11,807	\$ (179)	\$ 2,469	\$	(172)			
U.S. government and agency securities	13,969	(127)	_		_			
Corporate debt securities	6,527	(72)	_		_			
Total	\$ 32,303	\$ (378)	\$ 2,469	\$	(172)			

There were no impairment charges for available-for-sale debt or equity securities for the three months ended April 3, 2011 or April 4, 2010. There were no material other than temporary impairment losses recorded in other comprehensive income for the three months ended April 3, 2011 or April 4, 2010. Net realized gain or loss on sales of available-for-sale debt and equity securities for the three months ended April 3, 2011 and April 4, 2010 was not significant.

Contractual maturities of available-for-sale debt securities as of April 3, 2011 were as follows:

	Amount thousands)
Due within one year	\$ 9,394
Due in 1-5 years	37,957
Due in 5-10 years	9,230
Due after 10 years	99,205
Total	\$ 155,786

The maturities of asset-backed and mortgage-backed securities were allocated based on contractual principal maturities assuming no prepayments.

Note 7 — **Derivative Instruments**

The Company has foreign subsidiaries that operate and sell the Company's products in various markets around the world. As a result, the Company is exposed to changes in foreign-currency exchange rates. The Company utilizes forward contracts to manage its exposure associated with net asset and liability positions denominated in non-functional currencies and to reduce the volatility of earnings and cash flows related to forecasted foreign-currency transactions. The Company does not hold derivative financial instruments for speculative or trading purposes.

Cash-Flow Hedges

The Company enters into forward contracts that are designated as foreign-currency cash-flow hedges of selected forecasted payments denominated in currencies other than U.S. dollars. These forward contracts generally mature within 12 months. The Company evaluates and calculates the effectiveness of each hedge at least quarterly. Changes in fair value attributable to changes in time value are excluded from the assessment of effectiveness and are recognized in interest income and other, net. The effective portion of the forward contracts' gain or loss is recorded in other comprehensive income and is subsequently reclassified into earnings when the hedged expense is recognized within the same line item in the statements of operations as the impact of the hedged transaction. The ineffective portion of the gain or loss is reported in earnings immediately. As of April 3, 2011 and December 31,

2010, the total notional value of the Company's outstanding forward contracts, designated as foreign-currency cash-flow hedges, was \$38.2 million and \$41.7 million, respectively. For the three months ended April 3, 2011 and April 4, 2010, the after-tax effect of foreign-exchange forward contract derivatives on other comprehensive income was not material.

Other Foreign-Currency Hedges

The Company enters into foreign-exchange forward contracts that are used to hedge certain foreign-currency-denominated assets or liabilities that do not qualify for hedge accounting. These forward contracts generally mature within three months. Changes in the fair value of these hedges are recorded immediately in earnings to offset the changes in fair value of the assets or liabilities being hedged. As of April 3, 2011 and December 31, 2010, the total notional value of the Company's outstanding forward contracts, not designated as hedges under hedge accounting, was \$53.1 million and \$112.3 million, respectively. For the three months ended April 3, 2011 and April 4, 2010, the gain/(loss) on other foreign-currency hedges of \$1.8 million and \$(5.8) million, respectively, were recognized in interest income and other, net. This gain and loss was substantially offset by the loss and gain on the underlying foreign-currency-denominated assets or liabilities.

Fair Value of Derivative Instruments

As of April 3, 2011 and December 31, 2010, the fair value of derivative instruments included in the condensed consolidated balance sheets was not material.

Note 8 — Reconciliation of Basic and Diluted Income per Share

The following table provides a reconciliation of the numerators and denominators used in the computation of basic and diluted per share amounts:

	Three Months Ended					
	April 3, 2011			April 4, 2010		
Income*	Shares+	Per Share Amount	Income*	Shares+	Per Share Amount	
		(In thousands exc	ept per share amount	is)		
\$10,154	615,450	\$0.02	\$22,520	656,528	\$0.03	
k						
ards —	14,283	_	_	7,787	_	
\$10,154	629,733	\$0.02	\$22,520	664,315	\$0.03	
	\$10,154 k ards —	Income* Shares+	Income* Shares+ Per Share Amount (In thousands excessed	April 3, 2011	Name	

^{*} Numerator

The following table provides information about the weighted-average common share equivalents that were excluded from the computation of diluted shares because their inclusion would have an antidilutive effect on net income per share:

	Three M	Ionths Ended
	April 3, 2011	April 4, 2010
	(Shares	in thousands)
Anti-dilutive securities:		
Stock options	48,331	69,225
Restricted stock unit awards	16,217	399
Convertible notes	_	26,080

Note 9 — Segment and Geographic Information

The Chief Executive Officer has been identified as the Chief Operating Decision Maker ("CODM"). The CODM allocates resources and assesses the Company's performance using information about its revenue and operating income or loss before interest and taxes.

Historically, the Company operated in two reportable segments — the Semiconductor segment and the Storage Systems segment. The Semiconductor segment designs, develops and markets highly complex integrated circuits for storage and networking

⁺ Denominator

applications. These solutions include both custom solutions and standard products. The Storage Systems segment offered external storage systems and RAID adapters for computer servers and associated software for attaching storage devices to computer servers. On March 9, 2011, the Company entered into a definitive agreement to sell its external storage systems business to NetApp and started to operate its RAID adapter business as part of its semiconductor business. Accordingly, the Company now has one reportable segment. The change has been reflected in the Company's segment reporting for all periods presented.

Information about Geographic Areas

The following table summarizes the Company's revenues by geography based on the ordering location of the customer. Because the Company sells its products primarily to other sellers of technology products and not to end-users, the information in the table below may not accurately reflect geographic end-demand for its products.

	Three M	Three Months Ended					
	April 3, 2011	April 4, 2010	\$ Change	% Change			
		(Dollars i	in thousands)				
North America *	\$ 120,918	\$ 89,766	\$ 31,152	34.7%			
Asia **	302,658	332,738	(30,080)	(9.0)%			
Europe and the Middle East	49,688	50,168	(480)	(1.0)%			
Total	\$ 473,264	\$ 472,672	\$ 592	0.1%			

Primarily the United States.

Note 10 — Comprehensive Income

Comprehensive income or loss is defined as a change in equity of a company during a period from transactions and other events and circumstances, excluding transactions resulting from investments by owners and distributions to owners. The following table summarizes the changes in the total comprehensive income, net of taxes:

	Three !	Months Ended
	April 3, 2011	<u> April 4, 2010</u>
	(In	thousands)
Net income	\$ 10,154	\$ 22,520
Net unrealized gain on investments	326	527
Net unrealized gain/(loss) on derivatives	409	(859)
Foreign currency translation adjustments	1,401	(4,002)
Amortization of transition asset, prior-service cost and net actuarial loss	1,778	557
Total comprehensive income	\$ 14,068	\$ 18,743

Note 11 — Income Taxes

The Company recorded income tax benefits from continuing operations of \$4.1 million and \$23.1 million for the three months ended April 3, 2011 and April 4, 2010, respectively.

The income tax benefit from continuing operations for the three months ended April 3, 2011 included a reversal of \$8.2 million in liabilities for uncertain tax positions, which included previously unrecognized tax benefits of \$4.8 million and interest and penalties of \$3.4 million, as a result of the expiration of statutes of limitations in multiple jurisdictions. The income tax benefit from continuing operations for the three months ended April 4, 2010 was primarily a result of a reversal of \$27.9 million in liabilities for uncertain tax positions, which included previously unrecognized tax benefits of \$12.2 million and interest and penalties of \$15.7 million, as a result of the expiration of statutes of limitations in multiple jurisdictions.

In general, the Company computes the tax provision using an estimated annual tax rate. However, the Company excludes certain loss jurisditions from the computation of estimated annual rate when no benefit can be realized on those losses. Excluding certain foreign jurisdictions, the Company believes that it is more likely than not that the future benefit of deferred tax assets will not be realized.

^{**} Primarily Singapore, China and Taiwan.

Note 12 — Related Party Transactions

A member of the Company's board of directors is also a member of the board of directors of Seagate Technology. The Company sells semiconductors used in storage product applications to Seagate Technology for prices comparable to those charged to an unrelated third party. The Company also purchased drives used in its storage systems from Seagate Technology for prices comparable to those paid to other vendors for similar products. Revenues from sales by the Company to Seagate Technology were \$98.6 million and \$96.1 million for the three months ended April 3, 2011 and April 4, 2010, respectively. Purchases from Seagate Technology were \$0.4 million and \$11.6 million for the three months ended April 3, 2011 and April 4, 2010, respectively. The Company had accounts receivable from Seagate Technology of \$53.1 million and \$55.0 million as of April 3, 2011 and December 31, 2010, respectively.

The Company has an equity interest in a joint venture, Silicon Manufacturing Partners Pte Ltd. ("SMP"), with GLOBALFOUNDRIES, a manufacturing foundry for integrated circuits. SMP operates an integrated circuit manufacturing facility in Singapore. The Company owns a 51% equity interest in this joint venture, and GLOBALFOUNDRIES owns the remaining 49% equity interest. The Company's 51% interest in SMP is accounted for under the equity method because the Company is effectively precluded from unilaterally taking any significant action in the management of SMP due to GLOBALFOUNDRIES' significant participatory rights under the joint venture agreement. Because of GLOBALFOUNDRIES' approval rights, the Company cannot make any significant decisions regarding SMP without GLOBALFOUNDRIES' approval, despite the 51% equity interest. In addition, the General Manager, who is responsible for the day-to-day management of SMP, is appointed by GLOBALFOUNDRIES, and GLOBALFOUNDRIES provides day-to-day operational support to SMP.

The Company purchased \$10.8 million and \$12.0 million of inventory from SMP for the three months ended April 3, 2011 and April 4, 2010, respectively. As of April 3, 2011 and December 31, 2010, the amounts of inventory on hand that were purchased from SMP were \$7.1 million and \$5.0 million, respectively, and the amounts payable to SMP were \$4.6 million and \$1.2 million, respectively.

Note 13 — Commitments, Contingencies and Legal Matters

Purchase Commitments

The Company maintains purchase commitments with certain suppliers, primarily for raw materials and manufacturing services and for some non-production items. Purchase commitments for inventory materials are generally restricted to a forecasted time horizon as mutually agreed upon between the parties. This forecasted time horizon can vary for different suppliers. As of April 3, 2011, the Company had purchase commitments of \$371.2 million, which are due through 2015.

The Company has a take-or-pay agreement with SMP under which it has agreed to purchase 51% of the managed wafer capacity from SMP's integrated circuit manufacturing facility, and GLOBALFOUNDRIES has agreed to purchase the remaining 49% of the managed wafer capacity. SMP determines its managed wafer capacity each year based on forecasts provided by the Company and GLOBALFOUNDRIES. If the Company fails to purchase its required commitments, it will be required to pay SMP for the fixed costs associated with the unpurchased wafers. GLOBALFOUNDRIES is similarly obligated with respect to the wafers allotted to it. The agreement may be terminated by either party upon two years written notice. The agreement may also be terminated for material breach, bankruptcy or insolvency.

Guarantees

Product Warranties:

The Company warrants finished goods against defects in material and workmanship under normal use and service for periods of generally one to three years. A liability for estimated future costs under product warranties is recorded when products are shipped.

The following table sets forth a summary of changes in product warranties:

	April	nths Ended 3, 2011 usands)
Balance as of December 31, 2010	\$	17,617
Accruals for warranties issued during the period		2,258
Accruals related to pre-existing warranties (including changes in estimates)		547
Settlements made during the period (in cash or in kind)		(3,169)
Balance as of April 3, 2011	\$	17,253

Standby Letters of Credit:

As of April 3, 2011 and December 31, 2010, the Company had outstanding obligations relating to standby letters of credit of \$3.5 million and \$3.9 million, respectively. Standby letters of credit are financial guarantees provided by third parties for leases, customs and certain self-insured risks. If the guarantees are called, the Company must reimburse the provider of the guarantee. The fair value of the letters of credit approximates the contract amounts. The standby letters of credit generally renew annually.

Uncertain Tax Positions

As of April 3, 2011, the Company had \$150.8 million of unrecognized tax benefits, for which the Company is unable to make a reasonably reliable estimate as to when cash settlement with a taxing authority may occur. It is reasonably possible that the total amount of unrecognized tax benefits will increase or decrease in the next 12 months. Such changes could occur based on the normal expiration of statutes of limitations or the possible conclusion of ongoing tax audits in various jurisdictions around the world. If those events occur within the next 12 months, the Company estimates that the unrecognized tax benefits, plus accrued interest and penalties, could decrease by up to \$18.5 million.

Indemnifications

The Company is a party to a variety of agreements pursuant to which it may be obligated to indemnify the other party. Typically, these obligations arise primarily in connection with sales contracts and license agreements or the sale of assets, under which the Company customarily agrees to hold the other party harmless against losses arising from a breach of warranties, representations and covenants related to such matters as title to assets sold, validity of certain intellectual property rights, non-infringement of third-party rights, and certain income tax-related matters. In each of these circumstances, payment by the Company is typically subject to the other party making a claim to and cooperating with the Company pursuant to the procedures specified in the particular contract. This usually allows the Company to challenge the other party's claims or, in case of breach of intellectual property representations or covenants, to control the defense or settlement of any third-party claims brought against the other party. Further, the Company's obligations under these agreements may be limited in terms of activity (typically to replace or correct the products or terminate the agreement with a refund to the other party), duration and/or amounts. In some instances, the Company may have recourse against third parties covering certain payments made by the Company.

Legal Matters

On December 6, 2006, Sony Ericsson Mobile Communications USA Inc. ("Sony Ericsson") filed a lawsuit against Agere in Wake County Superior Court in North Carolina, alleging unfair and deceptive trade practices, fraud and negligent misrepresentation in connection with Agere's engagement with Sony Ericsson to develop a wireless data card for personal computers. The complaint claims an unspecified amount of damages and seeks compensatory damages, treble damages and attorneys' fees. On February 13, 2007, Agere filed a motion to dismiss for improper venue. On August 27, 2007, the court granted Agere's motion to dismiss for improper venue. Sony Ericsson appealed that ruling. On March 3, 2009, the North Carolina Court of Appeals affirmed the lower court's ruling. On October 22, 2007, Sony Ericsson filed a lawsuit in the Supreme Court of the State of New York, New York County against LSI, raising substantially the same allegations and seeking substantially the same relief as the North Carolina proceeding. In January 2010, Sony Ericsson amended its complaint by adding claims for fraudulent concealment and gross negligence. On September 10, 2010, LSI filed a motion for summary judgment. In January 2011, LSI and Sony Ericsson held an unsuccessful mediation in this matter. The Company is unable to estimate the possible loss or range of loss, if any, that may be incurred with respect to this matter.

On March 23, 2007, CIF Licensing, LLC, d/b/a GE Licensing ("GE") filed a lawsuit against Agere in the United States District Court for the District of Delaware, asserting that Agere products infringe patents in a portfolio of patents GE acquired from Motorola. GE has asserted that four of the patents cover inventions relating to modems. GE is seeking monetary damages. Agere believes it has a number of defenses to the infringement claims in this action, including laches, exhaustion and its belief that it has a license to the patents. The court postponed hearing motions based on these defenses until after the trial, and did not allow Agere to present evidence on these defenses at trial. On February 17, 2009, the jury in this case returned a verdict finding that three of the four patents were invalid and that Agere products infringed the one patent found to be valid and awarding GE \$7.6 million for infringement of that patent. The jury also found Agere's infringement was willful, which means that the judge could increase the amount of damages up to three times its original amount. The court has not scheduled hearings on Agere's post-trial motions related to its defenses. One of these motions seeks to have a mis-trial declared based on Agere's belief that GE withheld evidence in discovery, which affected Agere's ability to present evidence at trial. On October 6, 2010, a special master appointed by the court determined that GE's actions were not

wrongful and that the evidence withheld by GE was not material to the jury's findings. Agere is challenging this determination. If the jury's verdict is entered by the court, Agere would also expect to be required to pay interest from the date of infringing sales. If the verdict is entered, Agere intends to appeal the matter. On February 17, 2010, the court issued an order granting GE's summary judgment motions seeking to bar Agere's defenses of laches, exhaustion, and license and denying Agere's summary judgment motions concerning the same defenses. On July 30, 2010, the court held that one of the patents found invalid by the jury was valid. The court also held that the February 17, 2010 order was not inconsistent with its previous ruling that Agere would be permitted to renew its laches, licensing, and exhaustion defenses, and that Agere has not been precluded from asserting them post-trial. The Company is unable to estimate the possible loss or range of loss, if any, that may be incurred with respect to this matter.

On December 1, 2010, Rambus Inc. ("Rambus") filed a lawsuit against LSI in the United States District Court for the Northern District of California alleging that LSI's products infringe one or more of nineteen Rambus patents. These products contain either DDR-type memory controllers or certain high-speed SerDes peripheral interfaces, such as PCI Express interfaces and certain SATA and SAS interfaces. Rambus is seeking unspecified monetary damages, treble damages and costs, expenses and attorneys' fees due to alleged willfulness, interest, and permanent injunctive relief in this action. In addition, on December 1, 2010, Rambus filed an action with the International Trade Commission ("ITC") against LSI and five of its customers alleging that LSI products infringe six of the nineteen patents in the California case. Rambus also named five other companies and a number of their customers in the ITC action. Rambus is seeking an exclusionary order against LSI and its customers in the ITC action, which, if granted, would preclude LSI and its customers from selling these products in the U.S. The ITC instituted its investigation on December 29, 2010. LSI has filed an answer in the ITC proceedings and has requested a stay in the California case. The Company is unable to estimate the possible loss or range of loss, if any, that may be incurred with respect to this matter.

In addition to the foregoing, the Company and its subsidiaries are parties to other litigation matters and claims in the normal course of business. The Company does not believe, based on currently available facts and circumstances, that the final outcome of these other matters, taken individually or as a whole, will have a material adverse effect on the Company's consolidated results of operations or financial position. However, the pending unsettled lawsuits may involve complex questions of fact and law and may require the expenditure of significant funds and the diversion of other resources to defend. From time to time, the Company may enter into confidential discussions regarding the potential settlement of such lawsuits. However, there can be no assurance that any such discussions will occur or will result in a settlement. Moreover, the settlement of any pending litigation could require the Company to incur substantial costs and, in the case of the settlement of any intellectual property proceeding against the Company, may require the Company to obtain a license to a third-party's intellectual property that could require royalty payments in the future and the Company to grant a license to certain of its intellectual property to a third party under a cross-license agreement. The results of litigation are inherently uncertain, and material adverse outcomes are possible.

The Company has not provided accruals for any legal matters in its financial statements as potential losses for such matters are not considered probable and reasonably estimable. However, because such matters are subject to many uncertainties, the ultimate outcomes are not predictable, and there can be no assurances that the actual amounts required to satisfy alleged liabilities from the matters described above will not have a material adverse effect on its consolidated results of operations, financial position or cash flows.

Note 14 — Discontinued Operations

On March 9, 2011, the Company entered into a definitive agreement to sell its external storage systems business to NetApp for \$480.0 million in cash. The external storage systems business had historically been part of the Company's Storage Systems segment. In accordance with the applicable accounting guidance for the disposal of long-lived assets, the results of the external storage systems business are presented as discontinued operations and, as such, have been excluded from both continuing operations and segment results for all periods presented. Additionally, the assets of the external storage systems business at April 3, 2011 are presented as held for sale in the condensed consolidated balance sheets.

Following is selected financial information included in (loss)/income from discontinued operations for the external storage systems business:

Revenues (Loss)/income before income taxes

	i nree Months Ended				
April 3, 2011 April 4, 2010					
	(In tho	usands)			
\$	155,690	\$	164,510		
\$	(7,910)	\$	12,474		

Assets classified as held for sale as of April 3, 2011 related to the external storage systems business are as follows:

	Amount (In thousands)
Inventories	\$ 44,400
Property and equipment, net	25,952
Goodwill and identified intangible assets, net	117,847
Other assets	47,700
Total assets held for sale of discontinued operations	\$ 235,899

Note 15 — Subsequent Events

On May 6, 2011, the Company completed the sale of its external storage systems business to NetApp pursuant to the terms of the asset purchase agreement and received cash consideration of \$480.0 million.

As part of the asset purchase agreement, certain transitional services will be provided to NetApp for a period of up to eighteen months. The purpose of these services is to provide short-term assistance to the buyer in assuming the operations of the external storage systems business.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This management's discussion and analysis should be read in conjunction with the other sections of this Form 10-Q, including Part 1, "Item 1. Financial Statements."

Where more than one significant factor contributed to changes in results from year to year, we have quantified these factors throughout Management's Discussion and Analysis of Financial Condition and Results of Operations where practicable and material to understanding the discussion.

OVERVIEW

We design, develop and market complex, high-performance storage and networking semiconductors. We provide silicon-to-system solutions that are used at the core of products that create, store, consume and transport digital information. We offer a broad portfolio of capabilities, including custom and standard product integrated circuits used in hard disk drives, solid state drives, high-speed communications systems, computer servers, storage systems and personal computers. We also offer redundant array of independent disks, or RAID, adapters for computer servers and RAID software applications.

We sell our integrated circuits for server and storage applications principally to makers of hard disk drives, solid state drives and computer servers. We sell our integrated circuits for networking applications principally to makers of devices used in computer and telecommunications networks and, to a lesser extent, to makers of personal computers. We also generate revenue by licensing other entities to use our intellectual property.

We derive the majority of our revenue from sales of products for the server, hard disk drive and networking equipment end markets. We believe that these markets offer us attractive opportunities because of the growing demand to create, store, manage and move digital content. We believe that this growth is occurring as a result of a number of trends, including:

- The increasing popularity of mobile devices such as smart phones and media tablets, and the increasing use of the internet for streaming media, such as videos and music, which are driving the need for more network capacity;
- Consumer and business demand for hard disks to store increasing amounts of digital data, including music, video, pictures, and medical and other business records; and
- Enterprises are refreshing their data centers to provide higher levels of business support and analytics, which drives demand for new servers and storage systems and associated equipment.

Our revenues depend on market demand for these types of products and our ability to compete in highly competitive markets. We face competition not only from makers of products similar to ours, but also from competing technologies. For example, we see the development of solid state drives based on flash memory rather than the spinning platters used in hard disk drives as a long-term potential competitor to certain types of hard disk drives, and we are focusing development efforts in that area.

Prior to May 6, 2011, we also sold external storage systems, primarily to original equipment manufacturers, or OEMs, who resell these products to end customers under their own brand name. On May 6, 2011, we completed the sale of our external storage systems business to NetApp for \$480.0 million in cash. Because of our decision to exit the external storage systems business, we have reflected that business as a discontinued operation in our financial statements for all periods presented and have reclassified the assets of that business as held for sale in the balance sheet as of April 3, 2011. Until we made the decision to exit the external storage systems business, our storage systems business included our external storage systems and RAID adapter businesses. We have retained the RAID adapter business, which is now managed by and has been incorporated into our semiconductor business. We now operate in one reportable segment.

During the first quarter of 2011, we reported revenue of \$473.3 million, compared to \$472.7 for the first quarter of 2010, and net income of \$10.2 million, or \$0.02 per share, compared to \$22.5 million, or \$0.03 per share, for the first quarter of 2010.

On March 9, 2011, our Board of Directors authorized a new stock repurchase program of up to \$750.0 million of our common stock. During the first quarter of 2011, we repurchased 14.7 million shares for \$96.8 million under this program and anticipate continuing to repurchase stock under current market conditions. We ended the first quarter of 2011 with cash and cash equivalents,

together with short-term investments, of \$682.3 million, an improvement of \$5.6 million from the end of 2010 and received additional cash from the closing of the sale of our external storage systems business after the end of the quarter.

As we look forward into 2011, we are focused on a number of key objectives, including:

- Working with suppliers and customers to maintain our ability to meet customer demand for products in light of the impact of the recent earthquake and ensuing events in Japan. The packaging we use for some of our semiconductor products uses materials from a supplier in Japan whose operations have been affected. While we believe we have access to sufficient materials for the near term, we are working to ensure adequate supplies of materials acceptable to our customers to fully meet demand anticipated in the second half of 2011 and beyond.
- · Improving our gross margins and controlling operating expenses to drive improved financial performance;
- Meeting or exceeding our development, product quality and delivery commitments to our customers;
- Identifying attractive opportunities for future products, particularly in areas that are adjacent to technologies where we have strong capabilities;
- · Developing leading-edge new technologies; and
- Developing the skills of our workforce.

RESULTS OF OPERATIONS

Revenues

	Three Mo	nths Ended		
	April 3, 2011	April 4, 2010	\$ Change	% Change
		(Dollars in	millions)	
Revenues	\$ 473.3	\$ 472.7	\$ 0.6	0.1%

The increase in revenues was primarily attributable to increases in demand for our server RAID adapters from existing customers, program ramp-ups with new customers, and higher revenues from the licensing of our intellectual property. These increases were offset in part by a decrease in unit sales from semiconductors used in storage product applications, primarily as our customers experienced lower demand for fiber channel and host bus adaptors.

Significant Customers:

The following table provides information about our one customer that accounted for 10% or more of consolidated revenues:

	Three	Months Ended
	April 3, 2011	April 4, 2010
% of revenues	219	6 20%

Revenues by Geography

The following table summarizes our revenues by geography based on the ordering location of the customer. Because we sell our products primarily to other sellers of technology products and not to end-users, the information in the table below may not accurately reflect geographic end-demand for our products.

	Th			
	April 3, 20	1 <u>1</u> April 4, 2010	\$ Change	% Change
		(Dolla	rs in millions)	
North America *	\$ 120	9 \$ 89.8	\$ 31.1	34.7%
Asia **	302	7 332.7	(30.0)	(9.0)%
Europe and the Middle East	49	7 50.2	(0.5)	(1.0)%
Total	\$ 473	3 \$ 472.7	\$ 0.6	0.1%

Primarily the United States.

^{**} Primarily Singapore, China and Taiwan.

The increase in revenues in North America was primarily attributable to increased unit sales of server RAID adapters and semiconductors used in networking product applications. The decrease in revenues in Asia was primarily attributable to decreased unit sales of semiconductors used in storage and networking product applications.

Gross Profit Margin

		Three Mor	nths Endec	<u>1</u>			
	Apr	il 3, 2011	Apr	il 4, 2010	\$ C	hange	% Change
				(Dollars in	millions)		<u></u>
Gross profit	\$	224.2	\$	214.8	\$	9.4	4.4%
% of revenues		47.4%		45.4%			

Gross margins increased primarily due to a favorable shift in product mix as a result of increased revenues from the licensing of our intellectual property, which have higher gross margins than products, and a decrease in amortization of identified intangible assets.

Research and Development

	Apr	il 3, 2011	Apı	il 4, 2010	\$ C	hange	% Change
				(Dollars in	millions)		· · · · · · · · · · · · · · · · · · ·
Research and development	\$	142.3	\$	138.9	\$	3.4	2.4%
% of revenues		30.1%		29.4%			

R&D expenses increased primarily due to higher compensation-related expenses and facility costs as a result of headcount additions in China and India.

Selling, General and Administrative

		Three Mon	ths Ended				
	Apri	il 3, 2011	Apri	l 4, 2010	\$ C	Change	% Change
				(Dollars in	millions)		
Selling, general and administrative	\$	68.9	\$	70.4	\$	(1.5)	(2.1)%
% of revenues		14.6%		14.9%			

SG&A expenses decreased primarily due to decreases in amortization of identified intangible assets and sales commissions.

Restructuring of Operations and Other Items, net

The following table summarizes items included in restructuring of operations and other items, net from continuing operations:

		Three Months Ended				
	April	3, 2011		April 4, 2010		
		ions)				
Lease and contract terminations	\$	1.7(a)	\$	0.9		
Employee severance and benefits		1.6(b)		0.5		
Total restructuring expenses		3.3		1.4		
Other items		(0.5)		0.2		
Total	\$	2.8	\$	1.6		

⁽a) The amount primarily relates to changes in estimates and changes in time value of accruals for previously accrued facility lease exit costs.

Interest Expense, Interest Income and Other, net

The following table summarizes interest expense and components of interest income and other, net:

⁽b) The amount primarily relates to restructuring actions taken during the first quarter of 2011 as we continue to streamline our operations.

	Apri	Three Months Ended April 3, 2011 April 4, 201 (In millions)		
Interest expense	\$		\$	(3.9)
Interest income		3.4		3.6
Other income/(expense), net		0.9		(12.4)
Total	\$	4.3	\$	(12.7)

Interest expense decreased by \$3.9 million for the three months ended April 3, 2011 as compared to the three months ended April 4, 2010 as a result of the repayment of our 4% Convertible Subordinated Notes in May 2010.

Interest income decreased by \$0.2 million for the three months ended April 3, 2011 as compared to the three months ended April 4, 2010 primarily as a result of lower cash balances throughout the first quarter of 2011, offset in part by higher interest rates during the first quarter of 2011 compared to the first quarter of 2010.

The \$12.4 million of other expense, net, for the three months ended April 4, 2010 primarily consisted of \$11.6 million of other than temporary impairment charges for certain non-marketable equity securities.

Benefit from Income Taxes

During the three months ended April 3, 2011 and April 4, 2010, we recorded income tax benefits from continuing operations of \$4.1 million and \$23.1 million, respectively. The income tax benefit from continuing operations for the three months ended April 3, 2011 included a reversal of \$8.2 million in liabilities for uncertain tax positions, which included previously unrecognized tax benefits of \$4.8 million and interest and penalties of \$3.4 million, as a result of the expiration of statutes of limitations in multiple jurisdictions. The income tax benefit from continuing operations for the three months ended April 4, 2010 included a reversal of \$27.9 million in liabilities for uncertain tax positions, which included previously unrecognized tax benefits of \$12.2 million and interest and penalties of \$15.7 million, as a result of the expiration of statutes of limitations in multiple jurisdictions.

In general, we compute our tax provision using an estimated annual tax rate. However, we exclude certain loss jurisdictions from the computation of estimated annual rate when no benefit can be realized on those losses. Excluding certain foreign jurisdictions, we believe that it is more likely than not that the future benefit of deferred tax assets will not be realized.

Discontinued Operations

Net loss from discontinued operations, net of taxes, was \$8.4 million for the three months ended April 3, 2011 as compared to net income of \$8.2 million for the three months ended April 4, 2010. The \$16.6 million decrease primarily represents \$23.8 million of restructuring, asset impairment and other exit charges, offset in part by a \$7.9 million decrease in R&D expenses due to lower compensation-related expenses and facility costs as a result of the decision to sell the external storage systems business during the three months ended April 3, 2011.

FINANCIAL CONDITION, CAPITAL RESOURCES AND LIQUIDITY

Cash, cash equivalents and short-term investments increased to \$682.3 million as of April 3, 2011 from \$676.7 million as of December 31, 2010. The increase was mainly due to cash inflows generated from operating activities, which were substantially offset by cash outflows for financing and investing activities, as described below.

Working Capital

Working capital increased by \$164.7 million to \$943.9 million as of April 3, 2011 from \$779.2 million as of December 31, 2010. The increase was primarily attributable to the following:

- Assets held for sale increased by \$235.8 million primarily as certain assets of the external storage systems business were classified to assets held for sale;
- Accrued salaries, wages and benefits decreased by \$22.1 million primarily as a result of timing differences in the payment of salaries, benefits and performance-based compensation; and

• Cash, cash equivalents and short-term investments increased by \$5.6 million.

These increases in working capital were offset in part by the following:

- Accounts receivable decreased by \$40.5 million primarily as a result of an improvement in collections;
- Inventories decreased by \$31.7 million primarily as a result of the reclassification of the external storage systems business inventory to assets held
 for sale, offset in part by an increase in inventory purchases primarily due to new product introductions and expected increases in product demand in
 future quarters;
- Accounts payable increased by \$24.9 million primarily due to the normal timing of invoice receipts and payments, and increased levels of inventory
 purchases as a result of expected increases in product demand; and
- Prepaid expenses and other current assets decreased by \$4.9 million primarily as a result of decreases in prepaid software maintenance and other receivables.

Working capital increased by \$33.8 million to \$764.9 million as of April 4, 2010 from \$731.1 million as of December 31, 2009. The increase was primarily attributable to the following:

- Cash, cash equivalents and short-term investments increased by \$53.4 million;
- Inventories increased by \$16.4 million primarily as a result of an increase in inventory purchases to meet expected customer demand in the second quarter of 2010;
- Accounts payable decreased by \$12.8 million primarily due to the normal timing of invoice receipts and payments;
- Prepaid expenses and other current assets increased by \$9.4 million primarily due to increases in prepaid software maintenance contracts, other prepaid expenses and receivables; and
- Other accrued liabilities decreased by \$6.8 million primarily attributable to the utilization of restructuring reserves and payment of liabilities, offset in part by increases in accruals for interest on our convertible notes as the interest payment date approached in May 2010 and other accruals related to our operations.

These increases in working capital were offset in part by the following:

- Accounts receivable decreased by \$40.4 million primarily as a result of an improvement in collections; and
- Accrued salaries, wages and benefits increased by \$24.6 million primarily as a result of timing differences in the payment of salaries and benefits and the addition of performance-based compensation accruals, which we reduced in 2009 in response to the global economic downturn.

Cash Provided by Operating Activities

During the three months ended April 3, 2011, we generated \$108.0 million of cash from operating activities as a result of the following:

- Net income adjusted for non-cash items, including depreciation and amortization of \$56.0 million and stock-based compensation expense of \$14.0 million. The non-cash items and other non-operating adjustments are quantified in our condensed consolidated statements of cash flows included in Item 1; and
- A net increase of \$16.0 million in assets and liabilities, including changes in working capital components, from December 31, 2010 to April 3, 2011, as discussed above.

During the three months ended April 4, 2010, we generated \$105.8 million of cash from operating activities as a result of the following:

- Net income adjusted for non-cash items, including depreciation and amortization of \$67.0 million and stock-based compensation expense of \$16.4 million. The non-cash items and other non-operating adjustments are quantified in our condensed consolidated statements of cash flows included in Item 1:
- Offset in part by a net decrease of \$9.7 million in assets and liabilities, including changes in working capital components, from December 31, 2009 to April 4, 2010, as discussed above.

Cash Used in Investing Activities

Cash used in investing activities for the three months ended April 3, 2011 was \$23.8 million. The investing activities during the first quarter of 2011 were the following:

- · Purchases of property, equipment and software, net of proceeds from sales, totaling \$21.2 million; and
- Purchases of available-for-sale debt securities, net of proceeds from maturities and sales, of \$2.6 million.

Cash used in investing activities for the three months ended April 4, 2010 was \$16.0 million. The investing activities during the first quarter of 2010 were the following:

- Purchases of property, equipment and software, net of proceeds from sales, totaling \$27.3 million; and
- Proceeds from maturities and sales of available-for-sale debt securities of \$11.3 million.

We expect capital expenditures to be approximately \$55 million in 2011. In recent years, we have reduced our level of capital expenditures as a result of our focus on establishing strategic supplier alliances with foundry semiconductor manufacturers and with third-party assembly and test operations, which enables us to have access to advanced manufacturing capacity while reducing our capital spending requirements.

Cash Used in Financing Activities

Cash used in financing activities for the three months ended April 3, 2011 was \$79.5 million, as compared to \$22.6 million for the three months ended April 4, 2010. The primary financing activities during the three months ended April 3, 2011 were the use of \$96.8 million to repurchase our common stock, offset in part by the proceeds of \$17.3 million from issuances of common stock under our employee stock plans. On March 9, 2011, our Board of Directors authorized a new stock repurchase program of up to \$750.0 million of our common stock. We expect to fund the repurchases under this authorization from the proceeds of the sale of our external storage systems business, available cash and short-term investments.

The primary financing activities during the three months ended April 4, 2010 were the use of \$26.2 million to repurchase our common stock, offset in part by the proceeds of \$3.6 million from issuances of common stock under our employee stock plans.

We do not currently pay, and do not anticipate paying in the foreseeable future, any cash dividends to our stockholders.

Cash, cash equivalents and short-term investments are our primary source of liquidity. We believe that our existing liquid resources and cash generated from operations will be adequate to meet our operating and capital requirements and other obligations for more than the next 12 months. We may, however, find it desirable to obtain additional debt or equity financing. Such financing may not be available to us at all or on acceptable terms if we determine that it would be desirable to obtain additional financing.

CONTRACTUAL OBLIGATIONS

The following table summarizes our contractual obligations as of April 3, 2011:

	Payments Due by Period											
	Less Than 1 Year 1-3 Years		4-5 Years		After	After 5 Years		Other		Total		
		<u> </u>	·		·	(In millions)						
Operating lease obligations	\$	48.0	\$	51.4	\$	15.5	\$	3.6	\$	_	\$	118.5
Purchase commitments		367.0		4.1		0.1		_		_		371.2
Unrecognized tax positions plus interest												
and penalties		_		_		_		_		81.5**		81.5
Pension contributions		56.2		*		*		*		*		56.2
Total	\$	471.2	\$	55.5	\$	15.6	\$	3.6	\$	81.5	\$ (627.4
			-		_		-					

- * We have pension plans covering substantially all former Agere U.S. employees, excluding management employees hired after June 30, 2003. We also have pension plans covering certain international employees. Although additional future contributions will be required, the amount and timing of these contributions will be affected by actuarial assumptions, the actual rate of return on plan assets, the level of market interest rates, and the amount of voluntary contributions to the plans. The amount shown in the table represents our planned contributions to our pension plans for the remainder of 2011. Because any contributions for 2012 and later will depend on the value of the plan assets in the future and thus are uncertain, we have not included any amounts for 2012 and beyond in the above table.
- ** This amount represents the non-current tax payable obligation. We are unable to make a reasonably reliable estimate as to when cash settlement with a taxing authority may occur.

Operating Lease Obligations

We lease real estate and certain non-manufacturing equipment under non-cancelable operating leases. We also include non-cancellable obligations under certain software licensing arrangements in this category.

Purchase Commitments

We maintain purchase commitments with certain suppliers, primarily for raw materials and manufacturing services and for some non-production items. Purchase commitments for inventory materials are generally restricted to a forecasted time horizon as mutually agreed upon between the parties. This forecasted time horizon can vary for different suppliers.

Uncertain Tax Positions

As of April 3, 2011, we had \$150.8 million of unrecognized tax benefits, for which we are unable to make a reasonably reliable estimate as to when cash settlement with a taxing authority may occur. It is reasonably possible that the total amount of unrecognized tax benefits will increase or decrease in the next 12 months. Such changes could occur based on the normal expiration of statutes of limitations or the possible conclusion of ongoing tax audits in various jurisdictions around the world. If those events occur within the next 12 months, we estimate that the unrecognized tax benefits, plus accrued interest and penalties, could decrease by up to \$18.5 million.

Standby Letters of Credit

As of April 3, 2011 and December 31, 2010, we had outstanding obligations relating to standby letters of credit of \$3.5 million and \$3.9 million, respectively. Standby letters of credit are financial guarantees provided by third parties for leases, customs and certain self-insured risks. If the guarantees are called, we must reimburse the provider of the guarantee. The fair value of the letters of credit approximates the contract amount. The standby letters of credit generally renew annually.

CRITICAL ACCOUNTING POLICIES

There have been no significant changes in our critical accounting estimates or significant accounting policies during the three months ended April 3, 2011 as compared to the discussion in Part II, Item 7 and in Note 1 to our financial statements in Part II, Item 8 of our Annual Report on Form 10-K for the year ended December 31, 2010.

RECENT ACCOUNTING PRONOUNCEMENTS

The information contained in Note 1 to our financial statements in Item 1 under the heading "Recent Accounting Pronouncements" is incorporated by reference into this Item 2.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

There have been no significant changes in the market risk disclosures during the three months ended April 3, 2011 as compared to the discussion in Part II, Item 7A of our Annual Report on Form 10-K for the year ended December 31, 2010.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures: The Securities and Exchange Commission defines the term "disclosure controls and procedures" to mean a company's controls and other procedures that are designed to ensure that information required to be disclosed in the reports that it files or submits under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in the Commission's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed in our reports filed under the Securities Exchange Act is accumulated and communicated to management, including our chief executive officer and chief financial officer, as appropriate, to allow timely decisions regarding required or necessary disclosures. Our chief executive officer and chief financial officer have concluded, based on the evaluation of the effectiveness of the disclosure controls and procedures by our management with the participation of our chief executive officer and chief financial officer, as of the end of the period covered by this report, that our disclosure controls and procedures were effective for this purpose.

Changes in Internal Control: During the first quarter of 2011, we did not make any change in our internal control over financial reporting that materially affected or is reasonably likely to materially affect our internal control over financial reporting.

PART II — OTHER INFORMATION

Item 1. Legal Proceedings

This information is included under the caption "Legal Matters" in Note 13 to our financial statements in Item 1 of Part I.

Item 1A. Risk Factors

Set forth below are risks and uncertainties, many of which are discussed in greater detail in our Annual Report on Form 10-K for the year ended December 31, 2010, that, if they were to occur, could materially adversely affect our business or could cause our actual results to differ materially from the results contemplated by the forward-looking statements in this report and other public statements we make:

As a result of the earthquake and tsunami that affected Japan in early March 2011, and the events at the Fukushima Dai-ichi nuclear plant, it is possible that supplies of some materials we need to make semiconductors, as well as demand from our customers will be affected, which could have an adverse impact on our business.

On March 11, 2011, Japan experienced a 9.0 magnitude earthquake, which triggered a tsunami that led to widespread damage and business interruption. Following the earthquake, the cooling system at the Fukushima Daiichi nuclear generating plant failed and the plant has experienced significant emissions of radiation. As a result of the earthquake, a number of factories in Japan have been forced to shut down and the country has been experiencing rolling blackouts, further affecting industrial production.

We do not maintain significant operations in Japan and do not use semiconductor foundries in Japan. We do, however, obtain materials used in the packaging of some of our semiconductors from suppliers in Japan, the operations of one of which have been curtailed. While we have sufficient supplies of these materials to cover our needs for the near term, we are working with some of our customers and suppliers to obtain alternate sources of supply that are acceptable to our customers. While alternate suppliers and alternate materials may be available, our ability to use these alternate suppliers and materials may be delayed if customers require that the suppliers or materials be qualified before they accept products using materials from the alternate suppliers or using alternate materials. To the extent that our existing supplier's operations continue to be curtailed and we are unable to obtain timely, acceptable alternate sources of supply, our revenues could be adversely affected, beginning in the second half of 2011.

We have been working closely with our customers to understand their needs. We have some customers with significant operations in Japan and demand from these customers may be reduced if the customers' operations are curtailed. Our revenues may also be adversely affected if any of our customers are unable to obtain sufficient parts from other suppliers or if they experience reduced demand from their customers.

- We depend on a small number of customers. The loss of, or a significant reduction in revenue from, any of these customers would harm our results of operations.
- If we fail to keep pace with technological advances, or if we pursue technologies that do not become commercially accepted, customers may not buy our products and our results of operations may be harmed.
- We operate in intensely competitive markets, and our failure to compete effectively would harm our results of operations.
- Customer orders and ordering patterns can change quickly, making it difficult for us to predict our revenues and making it possible that our actual revenues may vary materially from our expectations, which could harm our results of operations and stock price.
- We depend on outside suppliers to manufacture, assemble, package and test our products; accordingly, any failure to secure and maintain sufficient manufacturing capacity at attractive prices or to maintain the quality of our products could harm our business and results of operations.
- · Failure to qualify our products or our suppliers' manufacturing lines with key customers could harm our business and results of operations.
- · Any defects in our products could harm our reputation, customer relationships and results of operations.
- · Our pension plans are underfunded, and may require significant future contributions, which could have an adverse impact on our business.
- We may be subject to intellectual property infringement claims and litigation, which could cause us to incur significant expenses or prevent us from selling our products.
- If we are unable to protect or assert our intellectual property rights, our business and results of operations may be harmed.
- · We are exposed to legal, business, political and economic risks associated with our international operations.
- We use indirect channels of product distribution over which we have limited control.
- · We may engage in acquisitions and strategic alliances, which may not be successful and could harm our business and operating results.
- The semiconductor industry is highly cyclical, which may cause our operating results to fluctuate.
- Our failure to attract, retain and motivate key employees could harm our business.
- Our operations and our suppliers' operations are subject to natural disasters and other events outside of our control that may disrupt our business and harm our operating results.
- We are subject to various environmental laws and regulations that could impose substantial costs on us and may harm our business.
- Our blank check preferred stock and Delaware law contain provisions that may inhibit potential acquisition bids, which may harm our stock price, discourage merger offers or prevent changes in our management.
- Class action litigation due to stock price volatility or other factors could cause us to incur substantial costs and divert our management's attention and resources.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

The following table contains information about our repurchases of our common stock during the quarter ended April 3, 2011.

Issuer Purchases of Equity Securities

Period	Total Number of Shares Purchased	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	Dollar Value of Shares that May Yet Be Purchased Under the Programs
January 1 — January 31, 2011	_	\$ —	_	\$ —
February 1 — February 28, 2011	_	_	_	_
March 1 — April 3, 2011	14,707,638	6.58	14,707,638	653,209,599
Total	14,707,638	\$ 6.58	14,707,638	\$ 653,209,599

On March 9, 2011, our Board of Directors authorized the repurchase of up to \$750 million of our common stock. The repurchases reported in the table above were made pursuant to this authorization.

Item 6. Exhibits

See the Exhibit Index, which follows the signature page to this report.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

LSI CORPORATION (Registrant)

Date: May 12, 2011 By /s/ Bryon Look

Bryon Look Executive Vice President, Chief Financial Officer and Chief Administrative Officer

29

EXHIBIT INDEX

2.1	omitted pursuant to Regulation S-K Item 601(b)(2), but will be provided to the Securities and Exchange Commission upon request.)
31.1	Certification of Chief Executive Officer pursuant to Rule 13a-14(a)
31.2	Certification of Chief Financial Officer pursuant to Rule 13a-14(a)
32.1	Certification of Chief Executive Officer pursuant to 18 U.S.C. 1350
32.2	Certification of Chief Financial Officer pursuant to 18 U.S.C. 1350
101.INS	XBRL instance document
101.SCH	XBRL taxonomy extension schema document
101.CAL	XBRL taxonomy extension calculation linkbase document
101.LAB	XBRL taxonomy extension label linkbase document
101.PRE	XBRL taxonomy extension presentation linkbase document

ASSET PURCHASE AGREEMENT

by and between

LSI CORPORATION

as Seller

and

NETAPP, INC.

as Buyer

dated as of March 9, 2011

TABLE OF CONTENTS

	D (1) 11	Page
1.	Definitions	2
	1.1 Defined Terms	2
	1.2 Additional Defined Terms	12
	1.3 Other Definitional and Interpretive Matters	13
2.	Purchase and Sale of the Engenio Business	14
	2.1 Purchase and Sale of Assets	14
	2.2 Excluded Assets	15
	2.3 Purchase Price	17
	2.4 Assumed Liabilities	17
	2.5 Excluded Liabilities	18
	2.6 Further Assurances; Further Conveyances and Assumptions; Consent of Third Parties	19
	2.7 Intellectual Property and Information	21
	2.8 Bulk Sales Law	21
	2.9 Taxes	22
	2.10 Buyer Designee	23
	2.11 Performance of Warranty Service; Warranty Reimbursement	23
3.	Representations and Warranties of Seller	23
	3.1 Organization and Qualification	23
	3.2 Subsidiaries	24
	3.3 Authorization; Binding Effect	24
	3.4 Non-Contravention; Consents	24
	3.5 Title to Property; Principal Equipment; Sufficiency of Assets	25
	3.6 Permits; Licenses	26
	3.7 Real Estate; Environmental Matters	26
	3.8 Compliance With Laws	28
	3.9 Litigation	28
	3.10 Business Employees	28
	3.11 Contracts	30
	3.12 Financial Information; Absence of Certain Changes	31
	3.13 Intellectual Property	32
	3.14 Product Liability and Recalls	37
	3.15 Product Warranty	37
	3.16 Inventory	37
	3.17 Customer and Suppliers	38
	3.18 Restrictions on the Engenio Business	38
	3.19 Taxes	38
	3.20 Brokers	39
4.	Representations and Warranties of Buyer	39
	4.1 Organization and Qualification	39

		Page
	4.2 Authorization; Binding Effect	39
	4.3 Non-Contravention; Consents	40
	4.4 Brokers	40
	4.5 Sufficiency of Funds	40
5.	Certain Covenants	41
	5.1 Access and Information	41
	5.2 Conduct of the Engenio Business	42
	5.3 Taxes	43
	5.4 Business Employees	44
	5.5 Collateral Agreements; Leased Equipment; Premises	48
	5.6 Regulatory Compliance; Post-Closing Cooperation	48
	5.7 Contacts with Suppliers and Customers	49
	5.8 Use of the Seller Name	49
	5.9 Non-Solicitation or Hiring of Transferred Employees	51
	5.10 No Negotiation or Solicitation	51
	5.11 Non-Competition	51
	5.12 Post Closing Remittances	52
	5.13 Prorations and Adjustments	53
	5.14 Notification of Certain Matters	53
	5.15 Title Insurance	53
	5.16 Closing Statement	54
	5.17 Indian Assets	54
6.	Confidential Nature of Information	54
	6.1 Confidentiality Agreement	54
	6.2 Seller's Proprietary Information	54
	6.3 Buyer's Proprietary Information	55
	6.4 Confidential Nature of Agreements	56
7.	Closing	56
	7.1 Deliveries by Seller or the Subsidiaries	56
	7.2 Deliveries by Buyer or a Buyer Designee	57
	7.3 Closing Date	57
	7.4 Contemporaneous Effectiveness	58
8.	Conditions Precedent to Closing	58
	8.1 General Conditions	58
	8.2 Conditions Precedent to Buyer's Obligations	58
	8.3 Conditions Precedent to Seller's Obligations	59
9.	Indemnification	59
	9.1 Effect of Investigation	59
	9.2 Survival of Representations and Warranties	60
	9.3 General Agreement to Indemnify	60
	9.4 General Procedures for Indemnification	62

62

		Page
10.	Miscellaneous Provisions	64
	10.1 Notices	64
	10.2 Expenses	64
	10.3 Entire Agreement; Modification	64
	10.4 Assignment; Binding Effect; Severability	65
	10.5 Governing Law	65
	10.6 Consent to Jurisdiction	65
	10.7 Waiver of Jury Trial	66
	10.8 Execution in Counterparts	66
	10.9 Public Announcement	66
	10.10 No Third-Party Beneficiaries	66
11.	Termination; Amendment and Waiver	67
	11.1 Termination	67
	11.2 Effect of Termination	67
	11.3 Amendment and Waiver	67
SIGN	NATURE PAGE	69
	iii	

Schedules

Schedules	
Schedule 1.1(a)	Individuals with Knowledge
<u>Schedule 1.1(b)</u>	Engenio Products
<u>Schedule 1.1(c)</u>	ONStor™ Products
<u>Schedule 1.1(d)</u>	Spare Parts Inventory Value
Schedule 2.1(a)	Contracts
Schedule 2.1(h)	Licenses
Schedule 2.1(j)	Governmental Permits
<u>Schedule 2.1(k)</u>	Domain Names
Schedule 2.2(f)	Excluded Contracts
Schedule 2.11	Warranty Reimbursement
Schedule 3.2	Subsidiaries
Schedule 3.4(b)	Required Consents
<u>Schedule 3.7(a)</u>	Leased Premises; Assumed Leases
<u>Schedule 3.7(b)</u>	Transferred Premises
Schedule 3.7(d)	Environmental Matters
Schedule 3.8	Compliance with Laws
Schedule 3.9	Litigation
Schedule 3.10(a)(i)	Business Employees
<u>Schedule 3.10(a)(ii)</u>	Unions and Works Councils
<u>Schedule 3.10(b)</u>	Employee Benefits
<u>Schedule 3.11(a)</u>	Material Contracts
<u>Schedule 3.11(b)</u>	Other Contracts Material to the Engenio Business
<u>Schedule 3.12(a)</u>	Financial Statements
<u>Schedule 3.12(c)</u>	Certain Events
<u>Schedule 3.12(d)</u>	Prepaid Revenues
Schedule 3.13	Intellectual Property
<u>Schedule 3.14(b)</u>	Product Liability and Recalls
<u>Schedule 3.15(a)</u>	Product Warranty Terms
<u>Schedule 3.15(b)</u>	Product Returns
Schedule 3.17	Customers and Suppliers
Schedule 5.3	Entity Level Allocation Statement
<u>Schedule 5.4(b)</u>	Offered Employees Minimum
Schedule 5.4(c)	Tuition Benefits; International Assignees
<u>Schedule 5.4(d)</u>	Severance Benefits
<u>Schedule 7.1(b)</u>	Required Closing Consents
Exhibits	
Exhibit A	Form of Assignment and Bill of Sale and Assumption Agreement
Exhibit B	Form of Intellectual Property Agreement
Exhibit C	Form of Lease Assignment
E 100 E	

Exhibit D
Exhibit E
Exhibit F
Exhibit G Form of Real Estate Deed

Form of Sublease
Form of Transition Services Agreement
Form of Amendment to the Supply Agreement

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT ("Agreement") is made as of March 9, 2011 by and between LSI CORPORATION, a Delaware corporation ("Seller" or "LSI"), and NETAPP, INC., a Delaware corporation ("Buyer") (Buyer and Seller, together, the "Parties").

RECITALS

- A. **WHEREAS**, Seller and its Subsidiaries (as hereinafter defined) are, among other things, engaged through one of its business units (referred to herein as "<u>Engenio Storage Group</u>") in the Engenio Business (as hereinafter defined);
 - B. WHEREAS, the Engenio Business is comprised of certain assets and liabilities that are currently part of Seller and its Subsidiaries;
- C. WHEREAS, Seller and its Subsidiaries desire to sell, transfer and assign to Buyer or a Buyer Designee (as hereinafter defined), and Buyer or a Buyer Designee desires to purchase and assume from Seller and its Subsidiaries, the Purchased Assets (as hereinafter defined), and Buyer or a Buyer Designee is willing to assume, the Assumed Liabilities (as hereinafter defined), in each case as more fully described and upon the terms and subject to the conditions set forth herein; and
- D. **WHEREAS**, Seller and its Subsidiaries and Buyer or a Buyer Designee desire to enter into each Assignment and Bill of Sale and Assumption Agreement, the India Purchase Agreement, the Intellectual Property Agreement, each Lease Assignment, the Real Estate Deed, each Sublease, the Supply Amendment and the Transition Services Agreement (each as hereinafter defined, and collectively, the "<u>Collateral Agreements</u>").

NOW, THEREFORE, in consideration of the mutual agreements and covenants herein contained and intending to be legally bound hereby, the parties hereto hereby agree as follows:

1. Definitions

1.1 Defined Terms

For the purposes of this Agreement the following words and phrases shall have the following meanings:

"Affiliate" of any Person means any Person that controls, is controlled by, or is under common control with such Person. As used herein, "control" (including the terms "controlling", "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities or other interests, by contract or otherwise.

"Assigned Intellectual Property" means all of the Intellectual Property and Information owned by Seller or one of its Affiliates that is used or held for use primarily in the operation or conduct of the Engenio Business and is being assigned to Buyer pursuant to the Intellectual Property Agreement, which will set forth the Assigned Software, the Assigned Technical Information, the Assigned Trademarks, and the Assigned Patents (as each such term is defined in the Intellectual Property Agreement).

"Assignment and Bill of Sale and Assumption Agreement" means each agreement in substantially the form set forth in Exhibit A.

"<u>Assumed Leases</u>" means the Leases identified on <u>Schedule 3.7(a)</u> to be assumed or subleased by Buyer or a Buyer Designee pursuant to a Lease Assignment or Sublease as designated on <u>Schedule 3.7(a)</u>.

"Benefit Plan" means each Pension Plan, Welfare Plan and employment, compensation, bonus, pension, profit sharing, deferred compensation, incentive compensation, stock ownership, stock option, stock appreciation right, stock purchase, phantom stock or other equity compensation, performance, retirement, thrift, savings, stock bonus, excess benefit, supplemental unemployment, paid time off, perquisite, tuition reimbursement, outplacement, fringe benefit, vacation, sabbatical, sick leave, severance, or retention, termination, change in control, redundancy policy, workers' compensation, retirement, cafeteria, disability, death benefit, hospitalization, medical, dental, life insurance, accident benefit, welfare benefit or other plan, program, agreement or arrangement, in each case maintained or contributed to, or required to be maintained or contributed to, by Seller or its Subsidiaries or any ERISA Affiliate, or any plan covering non-United States employees or former employees which if maintained or administered in or otherwise subject to the laws of the United States would be described in this paragraph, in each case for the benefit of any Business Employee.

"Business Day" means a day that is not (i) a Saturday, a Sunday or a statutory or civic holiday in the State of New York, (ii) a day on which banking institutions are required by Law to be closed in the State of New York, or (iii) a day on which the principal offices of Seller or Buyer are closed or become closed prior to 2:00 p.m. local time.

"Business Employees" means the employees of Seller or its Subsidiaries who are principally engaged in performing services for the Engenio Business.

"Business Records" means all books, records, ledgers, tangible data, disks, tapes, other media-storing data and files or other similar information whether in hardcopy or computer format and whether stored in network facilities or otherwise, in each case to the extent used or held for use primarily in the operation or conduct of the Engenio Business, including any advertising, promotional and media materials, training materials, trade show materials and videos, engineering information, manuals and data, including databases for reference designs, product datasheets, sales and purchase correspondence, including price lists, lists of present and former customers, information concerning customer contacts, purchasing history and invoices, technical characteristics and other information reasonably required for ongoing customer relationships, lists of present and former suppliers or vendors, mailing lists, warranty information, catalogs, sales promotion literature, advertising materials, brochures, bids, records of operation, accounting and financial records, personnel and employment records, standard forms of documents, manuals of operations or business procedures, designs, research materials and product testing reports, and any information relating to any Tax imposed on any Purchased Assets or with respect to the Engenio Business, but excluding portions of such items to the extent (i) they are included in, or primarily related to, any Excluded Assets or Excluded Liabilities, or (ii) any applicable Law prohibits the transfer of such information.

"Buyer Designee" means one or more Affiliates of Buyer identified to Seller in accordance with Section 2.10 prior to the Closing Date.

"CERCLA" means the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, 42 U.S.C. §§ 9601 et seq. as amended.

"Closing" means the closing of the transactions described in Article 7.

"Code" means the U.S. Internal Revenue Code of 1986, as amended.

"Confidentiality Agreement" means the Mutual Non-Disclosure Agreement between Seller and Buyer dated January 11, 2011.

"Contracts" means all Third-Party contracts, agreements, leases and subleases, supply contracts, commitments, purchase orders, sales orders, binding offers and instruments, or other written or oral arrangements, used or held for use primarily in the operation or conduct of the Engenio Business, to which Seller or a Subsidiary is a party, including the contracts identified as to be assigned on Schedule 2.1(a) and including any such contracts, agreements, leases and subleases, supply contracts, commitments, purchase orders, sales orders and instruments (i) for the lease of machinery, equipment, motor vehicles, furniture or office equipment, (ii) for the provision of goods or services to the Engenio Business, (iii) for the sale of goods or performance of services by the Engenio Business, (iv) for the sale and distribution of the Engenio Products, and (v) any such contracts, agreements, leases and subleases, supply contracts, commitments, purchase orders, sales orders and instruments referred to in clauses (i) — (iv), inclusive, entered into between the date hereof and the Closing Date by Seller or a Subsidiary to the extent such Contracts entered into after the date hereof are entered into in the ordinary course of business consistent with past practice and outstanding as of the Closing Date, but not the Excluded Contracts.

"Copyrights" means rights in works of authorship, including without limitation copyrights, whether registered or unregistered and whether arising under the laws of the United States or any other jurisdiction anywhere in the world, including moral rights, and all registrations and applications for registration with respect thereto.

"Encumbrance" means any lien, encumbrance, claim, charge, security interest, mortgage, pledge, easement, encroachment, building or use restriction, capital lease, conditional sale or other title retention agreement, covenant or other similar restriction, adverse claims of ownership or use, or other similar restriction or Third Party right affecting the Purchased Assets, but shall not include Permitted Encumbrances.

"Engenio Business" means the worldwide design, engineering, manufacturing, use, marketing, sale and distribution of external storage systems products and related embedded and value-added software and other components for use in external storage markets consisting of open, modular storage products comprised of complete systems and sub-assemblies configured from modular components, as carried on and conducted by the Engenio Storage Group since January 1, 2006, but excluding (a) the RAID adapter business of Seller which develops LSI® MegaRAID® and 3ware® storage controllers and software, (b) the ONStor™ clustered NAS gateway and non-integrated file storage products business of Seller which develops ONStor™ Products; (c) other semiconductor devices supplied by Seller to the Engenio Storage Group, and (d) any billing, order entry, fulfillment, accounting, collections or other corporate centralized functional organizations within, or controlled by, Seller.

"Engenio Product(s)" means all products and components thereof to the extent such components are owned, designed or manufactured by, or on behalf of, Seller or its Affiliates (but not any semiconductor devices supplied by Seller to the Engenio Storage Group) that are used or held for use primarily in the Engenio Business, including without limitation those listed on Schedule 1.1(b), which are under development, produced, marketed or sold by Seller or an Affiliate or Subsidiary of Seller, or have been developed, produced, marketed or sold by Seller or an Affiliate or Subsidiary of Seller since January 1, 2006.

"Environmental Law" means any Law that governs the existence of or provides a remedy for release of Hazardous Substances, the protection of persons, natural resources or the environment, the management of Hazardous Substances, or other activities involving Hazardous Substances including under CERCLA, the Hazardous Materials Transportation Act, 49 U.S.C. § 1801 et seq., the Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq., the Clean Water Act, 33 U.S.C. Section § 1251 et seq., the Clean Air Act, 42 U.S.C. § 7401 et seq., the Toxic Substance Control Act, 15 U.S.C. § 2601 et seq., the Oil Pollution Act of 1990, 33 U.S.C. § 2701 et seq., and the Occupational Safety and Health Act, 29 U.S.C. § 651 et seq., or any other similar Law, as any such Law has been amended or supplemented, and the regulations promulgated pursuant thereto.

"Environmental Liability" means any liability arising in connection with or in any way relating to Seller or a Subsidiary, any property now or previously owned, leased or operated by such Seller or a Subsidiary (or any predecessor thereof), the Engenio Business, or the Purchased Assets which (i) arise under or relate to any Environmental Law and (ii) relate to actions occurring or conditions existing on or before the Closing Date.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means each Subsidiary and any other Person that, together with Seller or any of the Subsidiaries is (or at the relevant time was) treated as a single employer under Section 414(b), (c), (m) or (o) of the Code and the regulations thereunder.

"Excluded Contracts" means those Contracts (i) identified in Schedule 2.2(f), (ii) under which performance by Seller or an Affiliate of Seller has been completed and for which there is no remaining warranty, maintenance, or support obligation and under which performance by the counterparty has been completed and for which there is no remaining payment obligation of such party, (iii) that constitute a General Purchase Agreement, or (iv) primarily related to Excluded Assets or Excluded Liabilities.

"Excluded Taxes" means any liability, obligation or commitment, whether or not accrued, assessed or currently due and payable, with respect to (i) any Taxes of Seller or its Affiliates (including any liability of Seller or its Affiliates for the Taxes of any other Person (other than Buyer or its Affiliates) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law)), as a transferee or successor, by contract or otherwise, (ii) any Taxes relating to, pertaining to, or arising out of, the Engenio Business or the Purchased Assets for any Pre-Closing Tax Period, including all interest, penalties or other amounts with respect thereto accruing in Post-Closing Tax Periods, and (iii) any Taxes required by Law to be paid by Seller or any Subsidiary (or withheld from Seller by Buyer or a Buyer Designee) as a result of their sale of Purchased Assets in any jurisdiction (including any mandatory withholding Taxes) other than (x) any Transfer Taxes to be paid by Buyer or a Buyer Designee under Section 2.9(b) and (y) any Taxes to the extent deducted an withheld by Buyer or a Buyer Designee at Closing pursuant to Section 2.9(a).

"Fixtures and Supplies" means all fixtures, improvements, furniture, furnishings, office and other supplies, vehicles, and other tangible personal property owned by Seller or a Subsidiary and used or held for use primarily in the operation or conduct of the Engenio Business that are located on the Transferred Premises or at the real property which is leased pursuant to the Assumed Leases, including desks, tables, chairs, file cabinets, racks, cubicles and other storage devices and office supplies and any additions, improvements, replacements and alterations thereto between the date hereof and the Closing Date and all warranties and guarantees, if any, express or implied with respect to the foregoing, but excluding any such items primarily related to Excluded Assets or Excluded Liabilities.

"GAAP" means U.S. generally accepted accounting principles.

"General Purchase Agreements" means Third-Party supply contracts or other agreements between Seller or an Affiliate of Seller and a Third Party pursuant to which Seller or an Affiliate purchases products or services from such Third-Party for any of Seller's or an Affiliate's businesses and not used or held for use primarily in the operation or conduct of the Engenio Business.

"Governmental Body" means any legislative, executive or judicial unit of any governmental entity (supranational, national, federal, provincial, state or local) or any

department, commission, board, agency, bureau, official or other regulatory, administrative or judicial authority thereof.

"Governmental Permits" means all governmental permits and licenses, certificates of inspection, approvals or other authorizations issued to Seller or a Subsidiary with respect to the Engenio Business or the Premises and necessary for the operation of the Engenio Business or the Premises as currently conducted under applicable Laws, including those identified on Schedule 2.1(j).

"<u>Hazardous Substance</u>" means (i) any hazardous, toxic or dangerous waste, substance or material defined as such pursuant to any Environmental Law, (ii) asbestos or PCBs and (iii) any other chemical, material or substance, exposure to which is prohibited, limited or regulated by any Governmental Body pursuant to any Environmental Law.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"India Purchase Agreement" means an agreement reasonably satisfactory to both parties for the sale or transfer of Purchased Assets in India.

"Information" means any and all documented and undocumented information (excluding Patents), including any technical information, Trade Secrets and other confidential information, data and drawings of whatever kind in whatever medium, specifications, techniques, know-how, network configurations and architectures, APIs, subroutines, techniques, user interfaces, URLs works of authorship, algorithms, formulae, protocols, schematics, compositions, processes, designs, bills of material, sketches, photographs, graphs, drawings, samples, non-patented inventions, discoveries, developments and ideas, build instructions, Software code (in any form, including source code and executable or object code), build scripts, test scripts, databases and data collections, past and current manufacturing and distribution methods and processes, tooling requirements, current and anticipated customer requirements, price lists, part lists, customer lists, market studies, business plans, database technologies, systems, structures, architectures, improvements, devices, concepts, methods and information, however documented and whether or not embodied in any tangible form, and any and all notes, analysis, compilations, studies, summaries, and other material containing or based, in whole or in part, on any information included in the foregoing, and including all tangible embodiments of any of the foregoing. "Information" does not include any semiconductor integrated circuit design or manufacturing technology.

"Intellectual Property" means all intellectual property rights arising from or associated with any of the following, whether protected, created or arising under the laws of the United States or any other jurisdiction anywhere in the world: (a) Copyrights, (b) Trademarks, (c) Patents, (d) Trade Secrets, (e) mask work rights and other rights protecting integrated circuit or chip topographies or designs, and any applications for registration therefor, (f) rights in databases and data collections (including knowledge databases, customer lists and customer databases), whether registered or unregistered, and any applications for registration therefor; (g) rights in URL and domain name registrations, (h) rights in inventions (whether or not patentable)

and improvements thereto, and (i) any other proprietary, intellectual or industrial property rights of any kind or nature now known or hereafter recognized in any jurisdiction worldwide.

"Intellectual Property Agreement" means the agreement in substantially the form set forth in Exhibit B.

"Inventory" means all inventory, wherever located, including raw materials, work in process, recycled materials, demo and evaluation inventory, finished products, inventoriable supplies, and non-capital spare parts owned by Seller or an Affiliate or Subsidiary and used or held for use primarily in the operation or conduct of the Engenio Business, and any rights of Seller or a Subsidiary to the warranties received from suppliers and any related claims, credits, rights of recovery and set-off with respect to such Inventory, but only to the extent such rights are assignable, but excluding any such items primarily related to Excluded Assets or Excluded Liabilities.

"IRS" means the U.S. Internal Revenue Service.

"knowledge of Seller" or "to Seller's knowledge" means the actual knowledge of the individuals specified on Schedule 1.1(a) after reasonable investigation.

"Law" means any supranational, national, federal, state, provincial or local law, statute, ordinance, rule, regulation, code, order, judgment, injunction or decree of any country.

"Lease" means the lease to be assigned or subleased, as the case may be, together with all amendments, modifications or supplements thereto, for any of the Leased Premises.

"Lease Assignment" means each assignment agreement with respect to a Lease in substantially the form set forth in Exhibit C.

"<u>Leased Equipment</u>" means the vehicles, computers, servers, machinery and equipment and other similar items leased and used or held for use by Seller or a Subsidiary primarily in the operation or conduct of the Engenio Business but excluding any such items primarily related to Excluded Assets or Excluded Liabilities.

"<u>Leased Premises</u>" means the real property identified on <u>Schedule 3.7(a)</u>, together with all rights, easements and privileges appertaining or relating to such real property, and all improvements located thereon, that is leased by Seller or a Subsidiary from Third Parties and used or held for use by Seller or a Subsidiary primarily in the operation or conduct of the Engenio Business.

"<u>Licensed Intellectual Property</u>," means all of the Intellectual Property and Information that is being licensed to Buyer pursuant to the Intellectual Property Agreement, which sets forth the Licensed Software, the Licensed Technical Information and the Licensed Patents (as each such term is defined in the Intellectual Property Agreement).

"<u>Licenses</u>" means all licenses, agreements and other arrangements identified on <u>Schedule 2.1(h)</u> under which Seller or a Subsidiary has the right to use any Intellectual Property or Information of a Third Party which is used or held for use primarily in the operation or conduct

of the Engenio Business but not (i) the Nonassignable Licenses, (ii) Contracts for non-customized Software that is licensed solely in executable or object code form pursuant to a nonexclusive, internal use software license, and not incorporated into, or used directly in, the development, manufacturing or distribution of, any of the Engenio Products or services of the Engenio Business and is generally available to the public on standard, non-negotiated terms ("Standard Software"), or (iii) any such items primarily related to Excluded Assets or Excluded Liabilities.

"Nonassignable Licenses" means those Licenses of Intellectual Property or Information under which Seller or an Affiliate of Seller is the licensee that are (i) not to be assigned as set forth on Schedule 2.1(h) or (ii) related to other businesses of Seller or an Affiliate of Seller and not used or held for use primarily in the operation or conduct of the Engenio Business, including Contracts for Standard Software and corporate wide information technology licenses used to operate Seller's retained businesses.

"ONStor™ Products" means those clustered NAS gateway and non-integrated file storage products listed on Schedule 1.1(c).

"<u>Patents</u>" means patents or patent applications worldwide of any kind or nature (including industrial designs and utility models that are subject to statutory protection), and any renewals, reissues, reexaminations, extensions, continuations, continuations-in-part, divisions and substitutions relating to any of the patents and patent applications, as well as all related counterparts to such patents and patent applications, wheresoever issued or pending anywhere in the world.

"Pension Plan" means each "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) or similar equivalent under applicable Laws other than the United States.

"Permitted Encumbrances" means any (i) statutory lien for Taxes, assessments and other governmental charges or liens of carriers, landlords, warehouseman, mechanics and material men incurred in the ordinary course of business, in each case for sums not yet due and payable, (ii) liens incurred or deposits made in the ordinary course of the Engenio Business in connection with workers' compensation, unemployment insurance and other types of social security or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return of money bonds and similar obligations, (iii) licenses or covenants granted by Seller or an Affiliate of Seller in connection with sales of products or patent licensing activities in the ordinary course of business, (iv) any use or other building restriction set forth in the Leases for the Leased Premises, (v) any Encumbrances that are set forth on the title commitment for the Transferred Premises and (vi) any Encumbrance or minor imperfection in title and minor encroachments, if any, that, individually or in the aggregate, are not material in amount, do not materially interfere with the conduct of the Engenio Business or with the use of the Purchased Assets and do not materially affect the value of the Purchased Assets or the Engenio Business.

"Person" means any individual, corporation, partnership, firm, association, joint venture, joint stock company, trust, unincorporated organization or other entity, or any government or

regulatory, administrative or political subdivision or agency, department or instrumentality thereof.

"Post-Closing Tax Period" means any Tax period beginning after the Closing Date, and, in the case of any Straddle Period, the portion of such Straddle Period beginning the day after the Closing Date.

"Pre-Closing Tax Period" means any Tax period ending on or before the Closing Date and, in the case of any Straddle Period, the portion of such Straddle Period ending on the Closing Date.

"Premises" means the (i) Leased Premises, (ii) the Transferred Premises, and (iii) the premises owned by Seller or a Subsidiary that are not being transferred hereunder, in each case that are used by the Engenio Business.

"Principal Equipment" means all personal property of Seller or a Subsidiary used or held for use by Seller or a Subsidiary primarily in the operation or conduct of the Engenio Business including such personal property located (i) on the Transferred Premises, at the real property which is leased pursuant to the Assumed Leases or at the Bangalore facility or (ii) at any other facility owned, leased or operated by or for the Engenio Business or at any contract manufacturer or original design manufacturer and including without limitation all computers, laptops, cell phone devices, smartphones, servers, printers, copiers, faxes, machinery, equipment (including any related replacement or spare parts, components dies, molds, tools, and tooling), phone or conferencing equipment, network equipment, data processing equipment and peripheral equipment and other similar items of personal property, but not (x) the Leased Equipment, (y) any such items primarily related to Excluded Assets or Excluded Liabilities, or (z) any such item of Principal Equipment abandoned by Buyer, at its sole election, that remains at any facility of Seller or its Subsidiary (other than the Transferred Premises or the real property being leased pursuant to the Assumed Leases) for 30 days following the Closing Date. Principal Equipment includes rights to the warranties received from the manufacturers and distributors of such items and to any related claims, credits, rights of recovery and set-off with respect to such items, but only to the extent that such rights are assignable.

"Real Estate Deed" means a limited warranty deed with respect to the Transferred Premises, in substantially the form set forth in Exhibit D.

"<u>Return</u>" means any return, declaration, report, claim for refund, or information return or statement, and any other document filed or required to be filed in respect of any Tax, including any schedule or attachment thereto or amendment thereof.

"Seller Material Adverse Effect" means any fact, circumstance, change, condition or effect that, individually or when taken together with all other such facts, circumstances, changes, conditions or effects that exist at the date of determination of the occurrence of the Seller Material Adverse Effect, has or is reasonably likely to have a material adverse effect on the business, operations, financial condition or results of operations of the Engenio Business, taken as a whole, or Seller's ability to perform its obligations under this Agreement and the Collateral Agreements or consummate the transactions contemplated hereby or thereby; provided, however,

that no facts, circumstances, changes, conditions or effects (by themselves or when aggregated with any other facts, circumstances, changes, conditions or effects) resulting from, relating to or arising out of the items enumerated in sub-clauses (i) to (vi) below shall be deemed to be or constitute a Seller Material Adverse Effect, and no facts, circumstances, changes, conditions or effects resulting from, relating to or arising out of the following (by themselves or when aggregated with any other facts, circumstances, changes or effects) shall be taken into account when determining whether a Seller Material Adverse Effect has occurred or may, would or could occur: (i) general economic, financial or political conditions in the United States or any other jurisdiction in which the Engenio Business has substantial business or operations, and any changes therein (including any changes arising out of acts of terrorism, war, weather conditions or other force majeure events), to the extent that such conditions do not have a materially disproportionate impact on the Engenio Business, taken as a whole, relative to other external storage businesses of comparable size; (ii) conditions in the storage industry, and any industry-wide changes therein (including any changes arising out of acts of terrorism, war, weather conditions or other force majeure events), to the extent that such conditions do not have a materially disproportionate impact on the Engenio Business, taken as a whole, relative to other external storage businesses of comparable size; (iii) conditions in the financial markets, and any changes therein (including any changes arising out of acts of terrorism, war, weather conditions or other force majeure events), to the extent that such conditions do not have a materially disproportionate impact on the Engenio Business, taken as a whole, relative to other external storage businesses of comparable size; (iv) acts of terrorism or war to the extent that such acts do not have a materially disproportionate impact on the Engenio Business, taken as a whole, relative to other external storage businesses of comparable size; (v) directly from the announcement or pendency of this Agreement and the transactions contemplated hereby, including negative reactions of any OEMs or customers to the sale announcement; or (vi) directly from compliance by Seller or its Subsidiaries with the express terms of this Agreement or the failure by Seller or its Subsidiaries to take any action that is prohibited by this Agreement.

"Software" means any and all (a) computer programs, including any and all software implementations of algorithms, heuristics models and methodologies, whether in source code or object code, (b) testing, validation, verification and quality assurance materials, (c) databases, conversion, interpreters and compilations, including any and all data and collections of data, whether machine readable or otherwise, (d) descriptions, schematics, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, (e) software development processes, practices, methods and policies recorded in permanent form, relating to any of the foregoing, (f) performance metrics, sightings, bug and feature lists, build, release and change control manifests recorded in permanent form, relating to any of the foregoing and (g) all documentation, including user manuals, web materials, and architectural and design specifications and training materials, relating to any of the foregoing.

"Straddle Period" means any Tax period that begins on or before and ends after the Closing Date.

"Sublease" means each sublease with respect to a Lease in substantially the form set forth in Exhibit E.

"Subsidiary" means each entity listed on Schedule 3.2.

"Supply Amendment" means the amendment to that certain Master OEM Agreement effective as of June 17, 2005 between Seller and Buyer in substantially the form set forth in Exhibit G.

"Tax" means a tax of any kind, and all charges, fees, customs, levies, duties, imposts, required deposits or other assessments, whether federal, state, local or foreign, including all net income, capital gains, gross income, gross receipt, property, franchise, sales, use, excise, registration, withholding, payroll, employment, social security, worker's compensation, unemployment, occupation, capital stock, ad valorem, value added, transfer, gains, profits, net worth, asset, transaction, real property, personal property, alternative, add-on minimum, escheat or estimated tax or other tax, including any interest, penalties or additions to tax with respect thereto, whether disputed or not, imposed upon any Person by any taxing or social security authority or other Governmental Body under applicable Law.

"Third Party" means any Person not an Affiliate of the other referenced Person or Persons.

"Trademarks" means trademarks, trade names, corporate names, business names, trade styles, service marks, service names, logos, slogans, 800 numbers, or other source or business identifiers and general intangibles of like nature, together with goodwill associated therewith, whether registered or unregistered and whether arising under the laws of the United States or any state or territory thereof or any other jurisdiction anywhere in the world, and registrations and applications for registration with respect to any of the foregoing.

"Trade Secrets" means all information of any kind or nature, in whatever form and whether or not embodied in a tangible medium, including customer lists, concepts, ideas, methods, processes, know-how, methodologies, designs, plans, schematics, bill of materials, drawings, formulae, technical data, specifications, research and development information, technology and product roadmaps, models, data bases, marketing materials and other proprietary or confidential information, in each case to the extent any of the foregoing derives economic value from not being generally known to other Persons who can obtain economic value from its disclosure or use, excluding any Copyrights or Patents that cover or protect any of the foregoing.

"<u>Transferred Premises</u>" means the real property identified on <u>Schedule 3.7(b)</u>, together with all rights, easements and privileges appertaining or relating to such real property, and all improvements located on such real property.

"Transition Services Agreement" means the agreement in substantially the form set forth in Exhibit F.

"Warranty Cap" means fourteen million five hundred thousand U.S. dollars (\$14,500,000) less the fair market value of spare parts Inventory transferred to Buyer useable to service warranty obligations for Engenio Products listed on Schedule 1.1(b) sold prior to the Closing, which fair market value shall not exceed the dollar amount set forth on Schedule 1.1(d).

"Welfare Plan" means each "employee welfare benefit plan" (within the meaning of Section 3(1) of ERISA) or similar applicable Laws of jurisdictions other than the United States.

1.2 Additional Defined Terms

For purposes of this Agreement, the following terms shall have the meanings specified in the Sections indicated below:

Term	Section
"Accrued Amounts"	Section 5.4(g)
"Agreement"	Preamble
"Allocation"	Section 5.3(c)
"Antitrust Division"	Section 5.6(b)
"ARD Jurisdiction"	Section 5.4(b)
"ARD Regulations"	Section 2.4(d)
"Asset Level Allocation Statement"	Section 5.3(c)
"Assigned Patents"	Section 3.13(a)
"Assigned Registered IP"	Section 3.13(a)
"Assigned Trademarks"	Section 3.13(a)
"Assumed Liabilities"	Section 2.4
"Balance Sheet"	Section 3.12(a)
"Bulk Sales Laws"	Section 2.8
" <u>Buyer</u> "	Preamble
"Buyer Indemnified Party"	Section 9.3(b)
"Buyer Proprietary Information"	Section 6.3
"Buyer Savings Plan"	Section 5.4(f)
"Cap Amount"	Section 9.3(f)
"Closing Date"	Section 7.3
"Closing Statement"	Section 5.16
"Collateral Agreements"	Recital D
"Engenio Intellectual Property"	Section 3.13(b)
"Engenio Storage Group"	Recital A
"Engenio Trade Secrets"	Section 3.13(g)
"Employment Related Liabilities"	Section 2.5(g)
"Entity Level Allocation Statement"	Section 5.3(b)
"Excluded Assets"	Section 2.2
"Excluded Leased Equipment"	Section 5.5(b)
"Excluded Liabilities"	Section 2.5
" <u>FTC</u> "	Section 5.6(b)
"Indemnified Party"	Section 9.3(a)
"Indemnifying Party"	Section 9.4(a)
" <u>Leave Employees</u> "	Section 5.4(b)
"Losses"	Section 9.3(a)
" <u>LSI</u> "	Preamble
"Material Contracts"	Section 3.11(a)
"Nonassignable Assets"	Section 2.6(c)
"Offered Employees"	Section 5.4(b)
"Parties"	Preamble
"Property Taxes"	Section 2.9(c)
" <u>Purchase Price</u> "	Section 2.3

Term	Section
"Purchased Assets"	Section 2.1
"Purchased Leased Equipment"	Section 5.5(b)
"Reasonable Efforts"	Section 5.8(a)(v)
"Required Closing Consents"	Section 7.1(b)
"Required Consents"	Section 3.4(b)
"Securities Act"	Section 5.14(a)
" <u>Seller</u> "	Preamble
"Seller Indemnified Party"	Section 9.3(c)
"Seller Name"	Section 5.8(a)
"Seller Proprietary Information"	Section 6.2
"Shared Contracts"	Section 2.6
" <u>Termination Date</u> "	Section 11.1(e)
" <u>Third-Party Claim</u> "	Section 9.4(a)
"Threshold Amount"	Section 9.3(f)
" <u>Transfer Taxes</u> "	Section 2.9(b)
" <u>Transferred Employees</u> "	Section 5.4(b)
"Unaudited Business Financials"	Section 3.12(a)
" <u>WARN Act</u> "	Section 3.10(j)

1.3 Other Definitional and Interpretive Matters

Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

- (a) <u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the starting reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.
- (b) <u>Gender and Number</u>. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.
- (c) <u>Headings</u>. The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "<u>Section</u>" are to the corresponding Section of this Agreement unless otherwise specified.
- (d) <u>Herein</u>. The words such as "<u>herein</u>," "<u>hereinafter</u>," "<u>hereof</u>," and "<u>hereunder</u>" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.
- (e) <u>Including</u>. The word "<u>including</u>" or any variation thereof means "including, without limitation" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

- (f) Currency. All currency references included herein shall refer to United States dollars.
- (g) <u>Reasonable Commercial Efforts</u>. Reasonable commercial efforts means that the obligated party is required to make a diligent, reasonable and good faith effort to accomplish the applicable objective. Such obligation, however, does not require an expenditure of funds or the incurrence of a liability on the part of the obligated party, nor does it require that the obligated party act in a manner that would be contrary to normal commercial practices in order to accomplish the objective. The fact that the objective is or is not actually accomplished is not, by itself, an indication that the obligated party did or did not in fact utilize its reasonable commercial efforts in attempting to accomplish the objective.
- (h) <u>Schedules and Exhibits</u>. The Schedules and Exhibits attached to this Agreement shall be construed with and as an integral part of this Agreement to the same extent as if the same had been set forth verbatim herein. Any matter disclosed by either party on any one Schedule with respect to any representation, warranty or covenant of such party shall be deemed disclosed for purposes of all other representations, warranties or covenants of such party to the extent that it is reasonably apparent from such disclosure that it also relates to such other representations, warranties or covenants.

2. Purchase and Sale of the Engenio Business

2.1 Purchase and Sale of Assets

Upon the terms and subject to the conditions of this Agreement and in reliance on the representations and warranties contained herein, on the Closing Date, Seller shall, or shall cause one or more of its Subsidiaries, as appropriate, to, grant, bargain, sell, transfer, assign, convey and deliver to Buyer or one or more Buyer Designees, and Buyer or one or more Buyer Designees shall purchase, acquire and accept from Seller or the applicable Subsidiary, all of the right, title and interest in, to and under the Purchased Assets that Seller or the applicable Subsidiary owns, leases, licenses, possesses or uses as the same shall exist on the Closing Date, wherever located, free and clear of Encumbrances. For purposes of this Agreement, "Purchased Assets" means all the assets, properties and rights used or held for use by Seller or the applicable Subsidiary primarily in the operation or conduct of the Engenio Business, whether tangible or intangible, real, personal or mixed, including the asset categories set forth or described in paragraphs (a) through (s) below (except in each case for the Excluded Assets), to the extent used or held for use primarily in the operation or conduct of the Engenio Business, whether or not any of such assets, properties or rights have any value for accounting purposes or are carried or reflected on or specifically referred to in Seller's or the applicable Subsidiary's books or financial statements:

- (a) the Contracts;
- (b) the Assumed Leases and any deposits related thereto;
- (c) the Transferred Premises;
- (d) the Principal Equipment and Purchased Leased Equipment;

- (e) the Fixtures and Supplies;
- (f) the Inventory;
- (g) the Assigned Intellectual Property;
- (h) the Licenses;
- (i) the Business Records;
- (j) the Governmental Permits (but only to the extent that such Governmental Permits are assignable or transferable to Buyer);
- (k) the registered domain names listed on Schedule 2.1(k);
- (l) all prepaid expenses for leased and rented equipment;
- (m) all prepaid deposits for customer orders to be completed after the Closing Date;
- (n) all Seller's lease deposits for the Assumed Leases;
- (o) all Third Party prepaid royalties and maintenance and support for Licenses;
- (p) all rights to the claims, causes of action, rights of recovery, and rights of set-off, made or asserted against any Person on or after the Closing Date relating to the Purchased Assets, whether arising out of actions or conditions occurring prior to, on, or after the Closing Date, including all rights to sue for or assert claims against and seek remedies for past, present and future infringements of any of the Assigned Intellectual Property and rights of priority and protection of interests therein and to retain any and all damages, settlement amounts and other amounts therefrom;
- (q) all guarantees, warranties, indemnities and similar rights in favor of Seller or an applicable Subsidiary related to the items identified in clauses (a) through (h) above;
 - (r) the goodwill of the Engenio Business; and
- (s) in the event Buyer or a Buyer Designee establishes or maintains a healthcare reimbursement or spending account program as of the Closing Date for Transferred Employees then, with respect to each such Transferred Employee who as of the Closing Date has a positive balance in Seller's healthcare reimbursement or spending account program, each such Transferred Employee's balance in Seller's health reimbursement or spending account shall be credited to the healthcare reimbursement or spending account program of Buyer or an applicable Buyer Designee.

2.2 Excluded Assets

Notwithstanding the provisions of Section 2.1, it is hereby expressly acknowledged and agreed that the Purchased Assets shall not include, and neither Seller nor any Subsidiary is granting, bargaining, selling, transferring, assigning, conveying or delivering to Buyer or a Buyer

Designee, and neither Buyer nor any Buyer Designee is purchasing, acquiring or accepting from Seller or any Subsidiary, any of the rights, properties or assets set forth or described in paragraphs (a) through (k) below (the rights, properties and assets expressly excluded by this Section 2.2 or otherwise excluded by the terms of Section 2.1 from the Purchased Assets being referred to herein as the "Excluded Assets"):

- (a) any of Seller's or its Affiliate's receivables, cash, cash equivalents, bank deposits or similar cash items (other than deposits related to the Assumed Leases) or employee receivables;
 - (b) any Intellectual Property or Information of Seller or any Affiliate other than the Assigned Intellectual Property or the Licensed Intellectual Property;
- (c) any (i) confidential personnel records pertaining to any Business Employee to the extent applicable Law prohibits the transfer of such information, or (ii) other books and records that Seller or any Affiliate of Seller is required by Law to retain; <u>provided</u>, <u>however</u>, that Buyer shall have the right, to the extent permitted by Law and subject to reasonable restrictions, to make copies of any portions of such retained confidential personnel records and other books and records that relate to the Engenio Business, the Purchased Assets, the Assumed Liabilities or the Transferred Employees; and (iii) any information management system of Seller or any Affiliate of Seller other than those used or held for use primarily in the operation or conduct of the Engenio Business;
- (d) any claim, right or interest of Seller or any Affiliate of Seller in or to any refund, rebate, abatement or other recovery for Excluded Taxes, together with any interest due thereon or penalty rebate arising therefrom;
 - (e) subject to Section 5.8, any rights to, or the use of, the "LSI" or "LSI Corporation" trademarks;
 - (f) the Excluded Contracts, the Nonassignable Licenses and the Excluded Leased Equipment;
 - (g) any insurance policies or rights of proceeds thereof;
- (h) except as specified in Section 2.1, any of Seller's or any Affiliate's rights, claims or causes of action against Third Parties relating to the assets, properties or operations of the Engenio Business arising out of transactions occurring prior to, and including, the Closing Date;
- (i) except as specifically provided in Section 5.4 or the applicable Assignment and Bill of Sale and Assumption Agreement for any particular jurisdiction, any of the assets of the Benefits Plans;
- (j) all other assets, properties, interests and rights of Seller or any Affiliate relating primarily to (i) the RAID adapter business of Seller which develops LSI® MegaRAID® and 3ware® storage controllers and software and (ii) the ONStor™ clustered NAS gateway and non-integrated file storage products business of Seller which develops ONStor™ Products; and

(k) all other assets, properties, interests and rights of Seller or any Affiliate not related primarily to the Engenio Business.

2.3 Purchase Price

In consideration of the grant, bargain, sale, transfer, assignment, conveyance and delivery by Seller and the Subsidiaries of the Purchased Assets to Buyer or a Buyer Designee, and in addition to assuming the Assumed Liabilities, Buyer and/or a Buyer Designee(s) shall pay to Seller or the applicable Subsidiary at the Closing, an aggregate amount equal to four hundred eighty million dollars (\$480,000,000) (the "Purchase Price") in cash by wire transfer of immediately available funds to an account designated by Seller's written instructions to Buyer at least two (2) Business Days prior to the Closing Date.

2.4 Assumed Liabilities

On the Closing Date, Buyer or one or more Buyer Designee shall execute and deliver to Seller each Assignment and Bill of Sale and Assumption Agreement and one or more Lease Assignments or Subleases pursuant to which Buyer or any such Buyer Designee shall accept, assume and agree to pay, perform or otherwise discharge, in accordance with the respective terms and subject to the respective conditions thereof, the Assumed Liabilities. For purposes of this Agreement, "Assumed Liabilities" means the liabilities and obligations set forth or described in paragraphs (a) through (d) below, whether or not any such liability or obligation has a value for accounting purposes or is carried or reflected on or specifically referred to in either Seller's or the applicable Subsidiary's books or financial statements:

- (a) the liabilities and obligations arising after the Closing Date under the Assumed Leases and the transferred Contracts, Licenses and Governmental Permits, but excluding the liabilities and obligations set forth in Sections 2.5(h), 2.5(i), and 2.5(j);
- (b) with respect to the Engenio Business, any support obligations and product warranty and sales return liabilities arising after the Closing Date from sales in the ordinary course of business after the Closing Date of the Engenio Products listed on <u>Schedule 1.1(b)</u>;
- (c) any support obligations and any product warranty liabilities arising after the Closing Date from sales in the ordinary course of Business of the Engenio Products listed on <u>Schedule 1.1(b)</u> on or before the Closing Date but only to the extent that such support obligations (including personnel costs) and product warranty liabilities exceed the Warranty Cap; and
- (d) except as provided herein, the obligations and liabilities with respect to the Transferred Employees (including European Union Council Directive 77/187 as amended by the European Union Council Directive 2001/23 (Acquired Rights Directive) or any local laws which implement the same (the "ARD Regulations")), the Engenio Business or the Purchased Assets in any of the foregoing cases arising from, or in connection with, the operation or conduct of the Engenio Business or the ownership of the Purchased Assets by Buyer or a Buyer Designee after the Closing Date.

2.5 Excluded Liabilities

Neither Buyer nor any Buyer Designee shall assume or be obligated to pay, perform or otherwise assume or discharge any liabilities or obligations of Seller or any of its Affiliates, whether direct or indirect, known or unknown, absolute or contingent, except for the Assumed Liabilities (all of such liabilities and obligations not so assumed being referred to herein as the "Excluded Liabilities"). For the avoidance of doubt, the parties agree that the Excluded Liabilities include, but are not limited to, any and all liabilities or obligations set forth or described in paragraphs (a) through (n) below, whether or not any such liability or obligation has a value for accounting purpose or is carried or reflected on or specifically referred to in Seller's or the applicable Subsidiary's books or financial statements:

- (a) any liability or obligation that arises from, or in connection with, the operation or the conduct of the Engenio Business or the ownership of the Purchased Assets on or prior to the Closing Date;
 - (b) any Excluded Taxes;
 - (c) any Environmental Liabilities;
 - (d) any liability or obligation arising out of or related to any Excluded Asset;
- (e) any trade payable that arises from, or in connection with, the operation or the conduct of the Engenio Business or the ownership of the Purchased Assets on or prior to the Closing Date;
 - (f) any indebtedness for borrowed money or guarantees thereof of Seller and its Subsidiaries or intercompany obligations of Seller or any Subsidiary;
- (g) except as set forth in Section 5.4(c), any liability or obligation relating to or arising out of (i) the employment and/or any termination of such employment by Seller or any Subsidiary of any employee or former employee of Seller or a Subsidiary on or before the Closing Date, including any and all liability or obligation relating to wages, remuneration, compensation, unreimbursed expenses, benefits, severance, pensions, sabbatical, vacation, personal days, floating holidays or other paid-time-off, working time related benefits, time savings accounts, end of career indemnities, 13th month payment or similar, anniversary bonus, early retirement, old-age part-time (Altersteilzeit) (including any amounts which Seller or any Subsidiary has book-reserved), reconciliation of interests (Interessenausgleich), social plans (Sozialplan), works council negotiation procedure, social security and related costs (together, the "Employment-Related Liabilities") of the employees or former employees of the Seller or any Subsidiary or Seller's or any Subsidiary's obligation to comply with the ARD Regulations, in each case, that are accrued or in the course of accrual or relate to periods prior to and on the Closing Date or that relate on a prorate temporis basis to the period prior and including the Closing Date; (ii) any employee's or former employee's or his/her dependents' rights or obligations under any fringe benefit of employment with Seller or a Subsidiary, including any Benefit Plan; (iii) any retention payments owed to Business Employees pursuant to arrangements entered into on or prior to the Closing Date by Seller or a Subsidiary; and/or (iv) the employment or the termination of employment (whether before, on or after Closing) or the transfer by

operation of Law, in each case as a result of the transaction contemplated by this Agreement, of any person who is not a Transferred Employee but who claims or is deemed to transfer to the Buyer or any Buyer Designee by operation of Law, including, without limitation, liabilities and obligations and Losses arising from, or connected with, any Employment-Related Liabilities;

- (h) any liability and obligation which arises out of or relates to any breach, default or violation by Seller or its Affiliates of the Assumed Leases and the Contracts, Licenses and Governmental Permits occurring on or prior to the Closing Date or which arises out of violation of applicable Law, in each case by Seller or its Affiliates;
- (i) except as set forth in Section 2.4(c), any support obligations (including personnel costs) and any product warranty liabilities arising from sales of Engenio Products listed on <u>Schedule 1.1(b)</u> on or before the Closing Date;
- (j) any support obligations (including personnel costs) and any product warranty and sales return or product return liabilities arising from sales of ONStor™ Products on, before or after the Closing Date;
- (k) any liability or obligation in connection with, or relating to, any actions, suits, claims or proceedings against Seller or any Subsidiary which arise out of, accrue, or relate to (i) the operation or conduct of the Engenio Business or (ii) the ownership of the Purchased Assets in each case on or before the Closing Date:
- (l) any benefit liability or obligation relating to or arising in connection with Section 4980B of the Code (COBRA) or otherwise by operation of applicable Law to provide continuation of health care coverage to employees or former employees of Seller or a Subsidiary or their dependents arising from a qualifying event occurring on or before the Closing Date;
- (m) any liability or obligation arising from any (i) customer rebates or market development funds on or before the Closing Date and (ii) sales returns or product returns arising from sales of Engenio Products listed on <u>Schedule 1.1(b)</u> on or before the Closing Date; and
 - (n) except as set forth in Section 5.4(c), any liability or obligation arising from or relating to any Benefit Plan.

2.6 Further Assurances; Further Conveyances and Assumptions; Consent of Third Parties

- (a) From time to time following the Closing to the extent permitted by applicable Law and subject to reasonable restrictions, Seller shall, or shall cause its Affiliates to, make available to Buyer or a Buyer Designee such confidential data and information in personnel records of Transferred Employees as is reasonably necessary for Buyer to integrate such employees into Buyer's or a Buyer Designee's workforce and comply with its obligations under Section 5.4.
- (b) From time to time following the Closing, Seller and Buyer shall, and shall cause their respective Affiliates to, execute, acknowledge and deliver all such further conveyances, notices, assumptions, releases and acquittances and such other instruments, and shall take such

further actions, as may be necessary or appropriate to assure fully to Buyer and its Affiliates and each of their respective successors or assigns, all of the properties, rights, titles, interests, estates, remedies, powers and privileges intended to be conveyed to Buyer or a Buyer Designee under this Agreement and the Collateral Agreements and to assure fully to Seller and its Affiliates and each of their respective successors and assigns, the assumption of the liabilities and obligations intended to be assumed by Buyer or a Buyer Designee under this Agreement and the Collateral Agreements, and to otherwise make effective the transactions contemplated hereby and thereby (including (i) transferring back to Seller or a Subsidiary any asset or liability not contemplated by this Agreement to be a Purchased Asset or an Assumed Liability, respectively, which asset or liability contemplated by this Agreement to be a Purchased Asset or an Assumed Liability, respectively, which was not transferred to Buyer or a Buyer Designee at the Closing).

(c) Nothing in this Agreement nor the consummation of the transactions contemplated hereby shall be construed as an attempt or agreement to assign any Purchased Asset, including any Contract, Lease, License, Governmental Permit, certificate, approval, authorization or other right, which by its terms or by Law is nonassignable without the consent of a Third Party or a Governmental Body or is cancelable by a Third Party in the event of an assignment ("Nonassignable Assets") (provided that in the event that Buyer or a Buyer Designee notifies Seller that any Purchased Asset should be transferred notwithstanding the right of a Third Party to cancel in the event of an assignment, then such Purchased Asset that is cancelable by a Third Party in the event of assignment shall not be included as a Nonassignable Asset for purposes of this Agreement) unless and until such consents shall have been obtained. Seller shall use all reasonable commercial efforts to obtain such consents and deliver any required notices prior to Closing, and Buyer shall, and shall cause its Affiliates to, cooperate with Seller to obtain such consents promptly. To the extent permitted by applicable Law, in the event consents to the assignment thereof cannot be obtained, Seller and Buyer shall, and shall cause their respective Affiliates to, cooperate in a mutually agreeable arrangement under which (i) Buyer or a Buyer Designee would obtain the benefits and assume the obligations under such Nonassignable Assets in accordance with this Agreement including by sub-contracting, sub-licensing, or sub-leasing to Buyer or a Buyer Designee, or (ii) such Nonassignable Assets would be held, as of and from the Closing Date, by Seller or the applicable Subsidiary in trust for Buyer or a Buyer Designee and the covenants and obligations thereunder would be performed by Buyer or a Buyer Designee in Seller's or such Subsidiary's name and all benefits and obligations existing thereunder would be for Buyer's or the applicable Buyer Designee's account. Seller shall, and shall cause its Affiliates to, also take or cause to be taken at Buyer's or a Buyer Designee's expense such actions in its name or otherwise as Buyer may reasonably request so as to provide Buyer or the applicable Buyer Designee with the benefits of the Nonassignable Assets and to effect collection of money or other consideration that becomes due and payable under the Nonassignable Assets, and Seller or the applicable Subsidiary shall promptly pay over to Buyer or the applicable Buyer Designee all money or other consideration received by it in respect to all Nonassignable Assets. If after the Closing Date any Nonassignable Asset becomes assignable (either because consent for the assignment thereof is obtained or otherwise), Seller shall promptly notify Buyer and assign or transfer such previously Nonassignable Asset to Buyer or the applicable Buyer Designee.

- (d) Buyer and Seller shall, and shall cause their respective Affiliates to, use their respective reasonable commercial efforts to obtain, or to cause to be obtained, any consent, substitution, approval, or amendment required to transfer all rights and obligations under any and all Contracts, Leases, Licenses, Governmental Permits, certificates, approvals, authorizations or other rights or obligations or liabilities that constitute Assumed Liabilities. In the case of any agreements which include obligations with respect to products of Seller other than Engenio Products (the "Shared Contract"), Buyer and Seller shall, and shall cause their respective Affiliates to, obtain consent and amend such agreement to remove any obligations with respect to such products other than Engenio Products. In the event such amendment is not obtained prior to the Closing Date, then such Shared Contract shall be deemed to be a Nonassignable Asset in accordance with this Section 2.6 (provided, however, that, in connection with such Nonassignable Asset, Buyer shall have no obligation to perform any of the obligations with respect to products other than Engenio Products).
- (e) As of and from the Closing Date, Seller on behalf of itself and its Affiliates authorizes Buyer, to the extent permitted by applicable Law and the terms of the Nonassignable Assets, at Buyer's expense, to perform all the obligations and receive all the benefits of Seller or its Affiliates under the Nonassignable Assets and appoints Buyer its attorney-in-fact to act in its name on its behalf or in the name of the applicable Affiliate of Seller and on such Affiliate's behalf with respect thereto.
- (f) Notwithstanding anything in this Agreement to the contrary, unless and until any consent or approval with respect to any Nonassignable Asset is obtained, such Nonassignable Asset shall not constitute a Purchased Asset and any associated liability shall not constitute an Assumed Liability for any purpose under this Agreement.
- (g) As reasonably requested by Buyer, Seller will identify the licenses included in the Nonassignable Assets and shall cooperate with and assist Buyer, at Buyer's reasonable request and expense, to obtain licenses or arrangements to replace the licenses, services and assets provided with respect to any Nonassignable Asset.

2.7 Intellectual Property and Information

Unless expressly set forth in this Agreement, the Intellectual Property Agreement or in any Collateral Agreement, no title, right or license of any kind is granted to Buyer pursuant to this Agreement with respect to the Intellectual Property or Information of Seller or any Affiliate of Seller, either directly or indirectly, by implication, by estoppel or otherwise.

2.8 Bulk Sales Law

Buyer hereby waives compliance by Seller and any Subsidiary with the requirements and provisions of any "<u>bulk-transfer</u>" Laws of any jurisdiction (collectively, the "<u>Bulk Sales Laws</u>"), including Article 6 of the California and Kansas Uniform Commercial Code, in each case that may otherwise be applicable with respect to the sale of any or all of the Purchased Assets to Buyer or a Buyer Designee. Seller agrees that the indemnification obligations set forth in Section 9.3(b)(ii) shall apply to Buyer's waiver of the Bulk Sales Laws.

2.9 Taxes

- (a) Buyer or a Buyer Designee shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to Seller or any Subsidiary such amounts as Buyer or any Buyer Designee is legally required to deduct and withhold under the Code, or any Tax Law with respect to Taxes resulting from Seller's or any Subsidiary's sale of Purchased Assets in any jurisdiction; provided that such right to deduct and withhold shall not apply to any Transfer Taxes allocated to Buyer or a Buyer Designee under Section 2.9(b). Buyer shall provide Seller with written notice of any requirement to so deduct or withhold any amount no less than five (5) days prior to the Closing Date, and shall provide Seller with a receipt from the applicable Governmental Body documenting the remittance of such deduction or withholding under the Code or any such Law as soon as reasonably practicable after the date of such deduction or withholding, but in any event not later than thirty (30) days following any such payment. To the extent that amounts are so withheld, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding was made.
- (b) The Party prescribed by applicable Law as primarily liable for the payment thereof shall be responsible for and timely pay any sales, use, stamp, registration, documentary, filing, recording, transfer, value added or similar fees or Taxes (including all applicable real estate transfer Taxes) ("<u>Transfer Taxes</u>") incurred in connection with the transfer of the Purchased Assets and the assumption of the Assumed Liabilities to and by, respectively, Buyer and Buyer Designees pursuant to this Agreement. In the case and to the extent of value added and similar Taxes incurred in connection with the transactions contemplated hereby that are recoverable by Buyer or a Buyer Designee, such Taxes shall be invoiced by Seller or its Subsidiaries to Buyer or Buyer Designee, as applicable, paid by Buyer or Buyer Designee to Seller or its Subsidiaries, as applicable, and remitted by Seller or its Subsidiaries, to the relevant Governmental Body in accordance with applicable Law, and Buyer or Buyer Designee shall be entitled to such recovery. The Party prescribed by Law as primarily liable for the payment of such Transfer Taxes shall prepare all necessary documents (including all Returns) with respect to all such amounts in a timely manner. The applicable party shall file such Returns and pay such Taxes and shall provide evidence satisfactory to the other party that such Returns have been filed and Transfer Taxes have been paid. Buyer and Seller shall cooperate to minimize the amount of Transfer Taxes.
- (c) All real property Taxes, personal property Taxes and similar ad valorem obligations ("Property Taxes") levied with respect to the Purchased Assets for a Straddle Period shall be apportioned between Seller and Buyer based on the number of days of such Straddle Period, and Seller shall be liable for the proportionate amount of Property Taxes that is attributable to the Pre-Closing Tax Period within such Straddle Period, and Buyer shall be liable for the proportionate amount of Property Taxes that is attributable to the Post-Closing Tax Period within such Straddle Period. Any refund, rebate, abatement or other recovery of Property Taxes attributable to the Pre-Closing Tax Period shall be for the account of Seller, and any refund, rebate, abatement or other recovery Taxes attributable to the Post-Closing Tax Period shall be for the account of Buyer. Upon receipt of any bill (or any refund, rebate, abatement, or other recovery) for such Property Taxes, Buyer or Seller, as applicable, shall present a statement to the other setting forth the amount of reimbursement to which each is entitled under this

Section 2.9(c) together with such supporting evidence as is reasonably necessary to calculate the proration amount. The proration amount shall be paid by the party owing it to the other within ten (10) days after delivery of such statement. In the event that Buyer or Seller makes any payment for which it is entitled to reimbursement under this Section 2.9(c), the applicable party shall make such reimbursement promptly but in no event later than ten (10) days after the presentation of a statement setting forth the amount of reimbursement to which the presenting party is entitled along with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement.

(d) Following the Closing, Buyer and Seller shall cooperate as reasonably requested for the purpose of enabling the requesting party to (i) make any election relating to Taxes, (ii) prepare Returns with respect to the Engenio Business or the Purchased Assets or (iii) to prepare for and defend audits or other Tax-related examinations by a Governmental Body with respect to the Engenio Business and the Purchased Assets. Such cooperation shall be at the expense of the requesting party.

2.10 Buyer Designee

The Parties agree that Buyer may assign the right to purchase certain of the Purchased Assets to one or more Buyer Designees or that one or more Buyer Designees may enter into a Collateral Agreement. Notwithstanding any such assignment or execution of a Collateral Agreement by a Buyer Designee, Buyer shall remain liable for, and any such assignment or execution shall not relieve Buyer of, its obligations hereunder or thereunder. Any reference to Buyer in this Agreement shall to the extent applicable also be deemed a reference to the applicable Buyer Designee, except where in context of this Agreement such use would not be appropriate.

2.11 Performance of Warranty Service; Warranty Reimbursement

Buyer agrees to perform the warranty service arising from sales of Engenio Products listed on <u>Schedule 1.1(b)</u>on, before or following the Closing Date. Seller shall pay to Buyer the Warranty Cap amount as set forth on <u>Schedule 2.11</u>.

3. Representations and Warranties of Seller

Except as set forth in the Schedules attached hereto and delivered by Seller to Buyer prior to the execution of this Agreement, Seller represents and warrants to Buyer that:

3.1 Organization and Qualification

Seller is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware and has all requisite corporate power and authority to carry on the Engenio Business as currently conducted by it and to own or lease and operate the Purchased Assets and conduct the Engenio Business. Seller is duly qualified to do business and is in good standing as a foreign corporation (in any jurisdiction that recognizes such concept) in each jurisdiction where the ownership or operation of the Purchased Assets or the operation or conduct of the Engenio Business requires such qualification, except where the failure to be so

qualified or in good standing, individually or in the aggregate, has not had and could not reasonably be expected to have a Seller Material Adverse Effect.

3.2 Subsidiaries

Schedule 3.2 sets forth a list of each Subsidiary of Seller that has title to any Purchased Asset or any Assumed Liability, together with its jurisdiction of organization. No other Affiliate of Seller owns any assets used or held for use primarily in the operation or conduct of the Engenio Business. Each entity is duly organized and validly existing and in good standing (in any jurisdiction that recognizes such concept) under the Laws of its jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate the Purchased Assets owned by it and to carry on its portion of the Engenio Business as presently conducted. Each Subsidiary of Seller set forth on Schedule 3.2 is duly qualified to do business and is in good standing as a foreign corporation or other entity (in any jurisdiction that recognizes such concept) in each jurisdiction where the ownership or operation of its properties and assets or the operation or conduct of the Engenio Business requires such qualification, except for failures to be so duly organized, validly existing, qualified or in good standing that, individually or in the aggregate, have not had and could not reasonably be expected to have a Seller Material Adverse Effect.

3.3 Authorization; Binding Effect

- (a) (i) Seller has all requisite corporate power and authority to execute, deliver and perform this Agreement and the Collateral Agreements to which it will be a party and to effect the transactions contemplated hereby and thereby, and the execution, delivery and performance of this Agreement and the Collateral Agreements to which it will be a party has been duly authorized by all requisite corporate action.
- (ii) Each Subsidiary that has title to any Purchased Asset or any Assumed Liability has all requisite corporate power and authority to execute, deliver and perform the Collateral Agreements to which it will be a party and to effect the transactions contemplated thereby, and the execution, delivery and performance of the Collateral Agreements to which it will be a party has been duly authorized by all requisite corporate action.
- (b) This Agreement has been duly executed and delivered by Seller and this Agreement is, and the Collateral Agreements to which Seller and each Subsidiary that has title to any Purchased Asset or any Assumed Liability will be a party when duly executed and delivered by Seller or such Subsidiary will be, valid and legally binding obligations of Seller or such Subsidiary, enforceable against Seller or such Subsidiary, as applicable, in accordance with their respective terms, except to the extent that enforcement of the rights and remedies created hereby and thereby may be affected by bankruptcy, reorganization, moratorium, insolvency and similar Laws of general application affecting the rights and remedies of creditors and by general equity principles.

3.4 Non-Contravention; Consents

(a) Assuming that all Required Consents have been obtained, the execution, delivery and performance of this Agreement by Seller and the Collateral Agreements by Seller or any

Subsidiary that is a party thereto and the consummation of the transactions contemplated hereby and thereby do not and will not: (i) result in a breach or violation of, or conflict with, any provision of Seller's or the applicable Subsidiary's charter, by-laws or similar organizational document, (ii) violate or result in a breach of or constitute an occurrence of default under any provision of, result in the acceleration or cancellation of any obligation under, or give rise to a right by any party to terminate or amend its obligations under, any mortgage, deed of trust, conveyance to secure debt, note, loan, indenture, lien, lease, agreement, license, permit, instrument, order, judgment, decree or other arrangement or commitment to which Seller or the applicable Subsidiary is a party or by which it is bound and which relates to the Engenio Business or the Purchased Assets or (iii) violate any applicable Law, order, judgment, decree, rule or regulation of any court or any Governmental Body having jurisdiction over Seller, a Subsidiary, the Engenio Business or the Purchased Assets, other than in the case of clauses (ii) and (iii), any such violations, breaches, defaults, accelerations or cancellations of obligations or rights that, individually or in the aggregate, are not and could not reasonably be expected to be material to the Engenio Business, taken as a whole.

(b) No consent, approval, order or authorization of, or registration, declaration or filing with, any Person is required to be obtained by Seller or a Subsidiary in connection with the execution, delivery and performance of this Agreement and the Collateral Agreements to which Seller or such Subsidiary will be a party or for the consummation of the transactions contemplated hereby or thereby by Seller or such Subsidiary, except for (i) any filings required to be made under the HSR Act and any applicable filings required under foreign antitrust Laws, (ii) consents or approvals of Governmental Bodies or other Third Parties that are required to transfer or assign to Buyer or a Buyer Designee any Purchased Assets or assign the benefits of or delegate performance with regard thereto in any material respect, which are set forth in Schedule 3.4(b) (items (i) and (ii) being referred to herein as the "Required Consents") and (iii) such consents, approvals, orders, authorizations, registrations, declarations or filings the failure of which to be obtained or made, individually or in the aggregate, are not and could not reasonably be expected to be material to the Engenio Business, taken as a whole.

3.5 Title to Property; Principal Equipment; Sufficiency of Assets

- (a) Seller or a Subsidiary has and at the Closing will have good and valid title to, or a valid and binding leasehold interest or license in, all real and personal tangible Purchased Assets free and clear of any Encumbrance except for Permitted Encumbrances.
- (b) Each material item of Principal Equipment is in good operating condition and repair, subject to normal wear and tear, suitable for the purposes for which it is currently being used, but is otherwise being transferred on a "where is" and, as to condition, "as is" basis.
- (c) Except for (i) the assets that will be used in connection with providing services under the Transition Services Agreement, (ii) the assets and Business Employees not transferred to Buyer or a Buyer Designee at Buyer's written request and (iii) the Excluded Assets (other than those set forth in Section 2.2(k)), the Purchased Assets and the Transferred Employees and the other rights to be acquired or licensed under this Agreement and the Collateral Agreements (including the services to be provided pursuant to the Transition Services Agreement) constitute (x) all property, assets, personnel and rights that are used or held for use by Seller or a Subsidiary

primarily in the operation or conduct of the Engenio Business and (y) all property, assets and rights that are necessary for the operation or conduct of the Engenio Business as currently conducted. In the event this Section 3.5(c) is breached because Seller or a Subsidiary has in good faith failed to identify and transfer any asset or property or provide any service used or held for use primarily in the Engenio Business, such breach shall be deemed cured if Seller or the applicable Subsidiary promptly transfers such properties or assets or provides such services to Buyer or a Buyer Designee at no additional cost to Buyer or a Buyer Designee.

3.6 Permits; Licenses

Except as set forth in Schedule 2.1(j), there are no material Governmental Permits necessary for or used by Seller or a Subsidiary to operate the Engenio Business as now being operated or to use or occupy the Premises, which Governmental Permits are required by currently effective Laws. Seller or one of its Subsidiaries owns, holds or possesses in their own name, all Governmental Permits necessary to own or lease, operate and use the Purchased Assets or own, use or occupy the Premises and to carry on and conduct the Engenio Business and its operations as presently conducted, except for such Governmental Permits, the absence of which, individually or in the aggregate, is not material to the Engenio Business. The Governmental Permits held, owned or possessed by Seller or a Subsidiary are valid and in full force and effect and no proceeding is recorded, pending or, to Seller's knowledge, threatened seeking the suspension, modification, limitation or revocation of any such Governmental Permit. Neither Seller nor any Subsidiary is in material violation of or default under any such Governmental Permits.

3.7 Real Estate; Environmental Matters

- (a) <u>Schedule 3.7(a)</u> contains a complete and accurate list of the Leased Premises and the Assumed Leases. Buyer has been provided with a complete and correct copy of each Assumed Lease. Except as set forth in <u>Schedule 3.7(a)</u>, each Assumed Lease is in full force and effect and neither Seller nor any Subsidiary has violated, and, to Seller's knowledge, the landlord has not violated or waived, any of the material terms or conditions of any Assumed Lease and all the material covenants to be performed by Seller or a Subsidiary, and to Seller's knowledge, the landlord under each Assumed Lease prior to the date hereof have been performed in all material respects.
- (b) <u>Schedule 3.7(b)</u> contains a complete and accurate list of the Transferred Premises. Seller or a Subsidiary has good and marketable title to the Transferred Premises in accordance with Kansas Law. The Transferred Premises are in good operating condition and repair, subject to ordinary wear and tear, suitable for the purposes for which it is currently being used, but is otherwise being transferred on a "where is" and, as to condition, "as is" basis. Except as set forth in <u>Schedule 3.7(b)</u>, none of such Transferred Premises are subject to any Encumbrance except for Permitted Encumbrances. No Third Party is in possession of any of the Transferred Premises or the Leased Premises (or any portion thereof).
- (c) The use of any Premises, as presently used by the Engenio Business, does not violate in any material respect any local zoning or similar land use or other applicable Laws. Neither Seller nor any Subsidiary is in violation of or in noncompliance with any covenant,

condition, restriction, order or easement affecting any Premises except where such violation or noncompliance, individually or in the aggregate, is not or could not reasonably be expected to be material to the Engenio Business, taken as a whole. There is no pending or, to Seller's knowledge, threatened condemnation or similar proceeding affecting any Premises.

- (d) Except as set forth in Schedule 3.7(d), in respect of the Engenio Business and the Premises:
 - (i) the operations of the Engenio Business and the Premises comply in all material respects with all applicable Environmental Laws;
- (ii) Seller and each Subsidiary has obtained all environmental, health and safety Governmental Permits required by or related to any Environmental Law and necessary for its operations, and all such Governmental Permits are in good standing, and Seller and each Subsidiary is in compliance with all terms and conditions of such permits except where the failure to obtain, maintain in good standing or be in compliance with, such permits, individually or in the aggregate, is not or could not reasonably be expected to be material to the Engenio Business, taken as a whole;
- (iii) since January 1, 2008, none of Seller, any Subsidiary or any of the Premises or the operations of the Engenio Business, have been subject to any on-going or previous investigation by, order from or agreement with any Person respecting (A) any Environmental Law, or (B) any remedial action arising from the release or threatened release of a Hazardous Substance into the environment;
- (iv) neither Seller nor any Subsidiary is subject to any judicial or administrative proceeding, order, judgment, decree or settlement alleging or addressing a violation of or liability under any Environmental Law;
- (v) Seller or each applicable Subsidiary has filed all notices required to be filed under any Environmental Law indicating past or present treatment, storage or disposal of a Hazardous Substance or reporting a spill or release of a Hazardous Substance into the environment except where the failure to file any such notices, individually or in the aggregate, has not had and could not reasonably be expected to have a Seller Material Adverse Effect;
- (vi) Seller and its Subsidiaries have provided or made available to Buyer all material reports, assessments, compliance reports or audits, remedial actions plans or similar documents relating to any material environmental conditions of the Premises that are in Seller's possession;
- (vii) To Seller's knowledge, there is no asbestos containing material or lead based paint containing materials in at, on, under or within the Transferred Premises;
- (viii) neither Seller nor any Subsidiary has received any written notice, or to Seller's knowledge, other claim to the effect that it is or may be liable to any Person as a result of the release or threatened release of a Hazardous Substance; and

(ix) there have been no releases or to Seller's knowledge, threatened releases of any Hazardous Substances into, on or under any of the Premises by Seller or its Affiliates or, to Seller's knowledge, any other Person, in any case in such a way as to create any liability (including the costs of investigation and remediation) under any applicable Environmental Law.

3.8 Compliance With Laws

- (a) Except as set forth on Schedule 3.8, with respect to the Purchased Assets, the Licensed Intellectual Property, the Engenio Products and the Engenio Business, Seller and each Subsidiary is in compliance in all material respects with all applicable Laws and all decrees, orders, judgments, writs, injunctions, permits and licenses of or from Governmental Bodies by which the Engenio Business, the Licensed Intellectual Property, the Engenio Products or the Purchased Assets are bound or affected.
- (b) Without limiting the generality of the foregoing, neither Seller nor any of its Subsidiaries, nor, to Seller's knowledge, any agent, employee or other Person associated with or acting on behalf of Seller or its Subsidiaries, has, directly or indirectly, used any corporate funds for any unlawful contribution, gift, entertainment or other unlawful expense relating to political activity, made any unlawful payment to any foreign or domestic government official or employee or to any foreign or domestic political party or campaign from corporate funds, violated any provision of the Foreign Corrupt Practices Act of 1977, as amended or any money laundering laws, or similar legislation in applicable jurisdictions or made any bribe, rebate, payoff, influence payment, kickback or other similar unlawful payment.

3.9 Litigation

Except as set forth on <u>Schedule 3.9</u>, there is no action, suit, consent decree, proceeding, arbitration or governmental investigation pending or, to Seller's knowledge, threatened by, against or involving Seller or any Subsidiary, the Engenio Business, the Purchased Assets, the Assumed Liabilities or the Transferred Employees (i) which seeks to restrain or enjoin the consummation of the transactions contemplated hereby or (ii) with respect to the Engenio Business, the Purchased Assets, the Assumed Liabilities or the Transferred Employees that, individually or in the aggregate, has been or could reasonably be expected to be material to the Engenio Business, taken as a whole. To Seller's knowledge, there is no basis for any such action, suit, decree, proceeding, arbitration or investigation not disclosed on <u>Schedule 3.9</u>.

3.10 Business Employees

(a) <u>Schedule 3.10(a)(i)</u> contains a complete and accurate list of all the Engenio Business Employees as of March 2, 2011, showing for each Business Employee, the name, title, location, service date, leave status (active or inactive), annual base salary or wages, annual incentive/bonus or commission opportunity and 2011 salary increase. Except as set forth on <u>Schedule 3.10(a)(ii)</u>, (i) no Business Employee is covered by any union, collective bargaining agreement or other similar labor agreement; (ii) to Seller's knowledge, are there no pending union, works council or similar labor organizing activities or arrangements; and (iii) in the three years prior to the date hereof, there has been no labor dispute, other than routine individual grievances, or any activity or proceeding by a labor union or representative thereof to organize

the Business Employees, or any lockouts, strikes, slowdowns, work stoppages or threats thereof by or with respect to Business Employees. No unfair labor practice, labor dispute or labor charge or complaint is pending or, to the knowledge of Seller, threatened with respect to any Business Employee.

- (b) Except as set forth in <u>Schedule 3.10(b)</u>, neither Seller nor any Subsidiary currently maintains, contributes to or has any liability under any Benefit Plan. With respect to each of the Benefit Plans identified on <u>Schedule 3.10(b)</u>, Seller has made available to Buyer true and complete copies of the most recent plan or summary or other written description describing all material terms thereof.
- (c) Each Benefit Plan that is intended to be qualified under Section 401(a) of the Code has received a favorable determination letter, or has pending or has time remaining in which to file an application for such determination, from the IRS, and to the knowledge of Seller, no fact exists indicating that any such determination letter should be revoked or not issued or reissued.
- (d) Other than as set forth in Schedule 3.10(b), no Benefit Plan is, and Seller, any Subsidiary or any ERISA Affiliate does not sponsor or maintain or has previously sponsored, maintained, contributed to, incurred an obligation to contribute to, or is or was required to contribute to: (i) any "multiemployer plan" as defined in Section 3(37) or 4011(a)(3) of ERISA, or (ii) any pension plan subject to Title IV of ERISA, Part 3 of Title I of ERISA or Section 412 of the Code.
- (e) No Benefit Plan provides for retiree or post-employment health, disability or life benefits to any Business Employee, and Seller, any Subsidiary or any ERISA Affiliate has not promised to or contracted with any Business Employee (either individually or to Business Employees as a group) with retiree health or other retiree employee welfare benefits.
- (f) Neither the execution nor the delivery of this Agreement or the Collateral Agreements or the consummation of the transactions contemplated hereby, either alone or in combination with another event, will (i) entitle any Business Employee to any payment from Seller or any Subsidiary; (ii) increase the amount of compensation or benefits due from Seller or any Subsidiary to any such employee; (iii) accelerate the vesting, funding or time of payment of any compensation, equity award or other benefits from Seller or any Subsidiary; or (iv) result in the payment of any "excess parachute payment" within the meaning of Section 280G of the Code or any similar provisions of foreign, state or local Law.
- (g) Neither Seller nor any Subsidiary maintains or sponsors any "nonqualified deferred compensation plan" within the meaning of Section 409A(d)(1) of the Code in which a Business Employee participates and may incur an additional tax under Section 409A of the Code.
- (h) Each Benefit Plan that covers any Business Employee outside of the United States or is otherwise not subject to ERISA or the Code has been maintained in substantial compliance with its terms and with the requirements prescribed by any and all applicable Laws (including without limitation any special provisions relating to the tax status of contributions to, earnings of

or distributions from such Benefit Plans where each such Benefit Plan was intended to have such tax status).

- (i) With respect to the Engenio Business, there is not presently pending or existing, and to Seller's knowledge there is not threatened, (i) any strike, slowdown, picketing, or work stoppage, (ii) any application for certification of a collective bargaining agent, (iii) any controversies or disputes pending, or to Seller's knowledge, threatened between Seller or any Subsidiary and any of its employees, or (iv) any claims, litigation or disputes by a works council or other applicable Government Body, which controversies, claims, litigation or disputes, individually or in the aggregate, are or could reasonably be expected to be material to the Engenio Business, taken as a whole.
- (j) Except for the defined benefit plan set forth on Schedule 3.10(b) with respect to Business Employees located in Germany, no Business Employee has a contractual right to any defined benefit pension or rights under an occupational pension plan.
- (k) In the twelve (12) months prior to the date hereof, no Business Employee located in the European Union has transferred to the Engenio Business from another business unit of Seller or any of its Subsidiaries or pursuant to the ARD Regulations.

3.11 Contracts

- (a) Schedule 3.11(a) contains a complete and accurate list of all existing Contracts of Seller or a Subsidiary that:
- (i) involve or could reasonably be expected to involve payments by or to Seller or a Subsidiary either of more than \$250,000 per year or more than \$500,000 in the aggregate over the full term thereof;
- (ii) contain any provision or covenant prohibiting or limiting the ability of Seller or a Subsidiary to (A) engage in any activity relating to or involving the Engenio Business (including geographical restrictions), (B) to compete in any line of business, directly or indirectly, with any Person as to the Engenio Business;
 - (iii) provide for "most favored nation" terms, including such terms for pricing, and that is material to the Engenio Business;
- (iv) create or obligate Seller or a Subsidiary to participate in any joint venture or similar arrangement with respect to or affecting the Engenio Business or the Purchased Assets:
- (v) contain material maintenance, warranty, support or similar obligations, other than as set forth on the standard terms and conditions of sale included in <u>Schedule 3.15</u>;
- (vi) for any distributor, original equipment manufacturer, reseller, value added reseller, sales, agency or manufacturer's representative relationships that is material to the Engenio Business; and

- (vii) constitute any other agreement, commitment, arrangement or plan not made in the ordinary course of business that is material to the Engenio Business (clauses (i) through (vii) collectively, the "Material Contracts").
- (b) <u>Schedule 3.11(b)</u> contains a complete and accurate list of all contracts, agreements, commitments, purchase orders, and instruments that are material to the Engenio Business but are not included in the Purchased Assets.
- (c) Each Material Contract is valid, binding and enforceable against Seller or the applicable Subsidiary and, to Seller's knowledge, the other parties thereto in accordance with its terms and is in full force and effect. Neither Seller nor any Subsidiary is in default under or in breach of or is otherwise delinquent in performance under any Material Contract (and neither Seller nor any Subsidiary has received any notice alleging any such default, breach or delinquency). To Seller's knowledge, each of the other parties thereto has performed in all material respects all obligations required to be performed by it under, and is not in material default under, any Material Contract and no event has occurred that, with notice or lapse of time, or both, would constitute such a material default. Seller or a Subsidiary has made available to Buyer true and complete copies of all Material Contracts.

3.12 Financial Information; Absence of Certain Changes

- (a) <u>Schedule 3.12(a)</u> contains true and complete copies of the following unaudited financial statements of the Engenio Business (the "<u>Unaudited Business</u> Financials"):
 - (i) unaudited balance sheet of the Engenio Business as of December 31, 2010 (the "Balance Sheet");
 - (ii) unaudited balance sheet of the Engenio Business as of December 31, 2009; and
 - (iii) unaudited statements of operations of the Engenio Business for the years ended December 31, 2010, 2009 and 2008.
- (b) The Unaudited Business Financials were prepared on the basis of the books and records (which are accurate and complete in all material respects) of the Engenio Business (in each case, as of the date of such Unaudited Business Financials) and in accordance with GAAP consistently applied throughout the periods covered, except for the omission of footnotes and for normal year-end adjustments. The Unaudited Business Financials present fairly, in all material respects, the financial position of the Engenio Business as of the dates thereof and the results of its operations and cash flows for each of the periods then ended in conformity with GAAP except for (i) the omission of footnotes and normal year-end adjustments; and (ii) any potential adjustments relating to the tax provision in the statements of operations or to any line items in the balance sheets affected by tax related adjustments. The Closing Statement will be prepared on the basis of the books and records of the Engenio Business and each line item thereof will be prepared in accordance with GAAP applied consistently with the Unaudited Business Financials.

- (c) Except as set forth in <u>Schedule 3.12(c)</u>, since the date of the Balance Sheet, the Engenio Business has been conducted by Seller and the Subsidiaries in the ordinary course consistent with past practices and there has not been:
- (i) any event, occurrence, development or state of circumstances or facts which, individually or in the aggregate, has had or could reasonably be expected to have a Seller Material Adverse Effect;
- (ii) any creation or other incurrence of any Encumbrance on any Purchased Asset other than in the ordinary course of business consistent with past practices;
 - (iii) failure to timely pay when due any material obligation related to the Engenio Business;
- (iv) any material damage, destruction or other casualty loss (whether or not covered by insurance) affecting the Engenio Business or any Purchased Asset;
- (v) any transaction or commitment made, or any contract or agreement entered into, by Seller or a Subsidiary relating to the Engenio Business or any Purchased Asset (including the acquisition or disposition of any assets) or any relinquishment by Seller or a Subsidiary of any contract or other right, in either case, material to the Engenio Business, other than transactions and commitments other than in the ordinary course of business consistent with past practices and those contemplated by this Agreement and the Collateral Agreements;
 - (vi) any change in any method of accounting or accounting practice by Seller or a Subsidiary with respect to the Engenio Business;
- (vii) any (i) employment, retention, bonus, deferred compensation, severance, retirement or other similar agreement entered into with any Business Employee (or any amendment to any such existing agreement), (ii) change in compensation or other benefits payable to any Business Employee pursuant to any severance or retirement plans or policies thereof, or (iii) grant of any severance or termination pay to any Business Employee;
- (viii) any labor dispute, other than routine individual grievances, or any activity or proceeding by a labor union or representative thereof to organize the Business Employees, or any lockouts, strikes, slowdowns, work stoppages or threats thereof by or with respect to Business Employees; or
- (ix) any shipments or sales of quantities of Engenio Products to customers, including distributors, other than in the ordinary course consistent with their past requirements.
- (d) Except as set forth in <u>Schedule 3.12(d)</u>, Seller has not received or booked any prepaid revenues for the Engenio Business applicable to performance due after the Closing Date.

3.13 Intellectual Property

(a) Seller or one of its Affiliates owns exclusively all right, title and interest in and to the Assigned Intellectual Property, free and clear of all Encumbrances other than Permitted

Encumbrances. Seller or one of its Affiliates owns or has a valid right to grant the licenses to the Licensed Intellectual Property that it is licensing to Buyer pursuant to the Intellectual Property Agreement. Seller and its Affiliates have not received any notice or claim challenging Seller's or any of its Affiliates' ownership of any Assigned Intellectual Property or suggesting that any other Person has any claim of legal or beneficial ownership or exclusive rights with respect thereto, nor to Seller's knowledge is there a reasonable basis for any claim that Seller or its Affiliates, as applicable, does not so own any of such Assigned Intellectual Property. Schedule A of Appendix F to the Intellectual Property Agreement contains a complete and accurate list of all registered Trademarks and pending applications for registration of Trademarks that are to be assigned to Buyer or a Buyer Designee (the "Assigned Trademarks"). Schedule A of Appendix G to the Intellectual Property Agreement contains a complete and accurate list of all Patents that are to be assigned to Buyer or a Buyer Designee (the "Assigned Patents" and, together with the Assigned Trademarks, the "Assigned Registered IP"), and there are no registered Copyrights and pending applications for registration of Copyrights being assigned to Buyer of a Buyer Designee. The appendices to the Intellectual Property Agreement attached hereto identify as of the date of this Agreement (i) each item of Assigned Registered IP in which Seller has or purports to have an ownership interest of any nature (whether exclusively, jointly with another Person, or otherwise), (ii) the jurisdiction in which such item of Assigned Registered IP has been registered or filed and the applicable application, registration, or serial or other similar identification number, and (iii) any other Person that has an ownership interest in such item of Assigned Registered IP and the nature of such ownership interest. The parties acknowledge that the appendices to the Intellectual Property Agreement may be updated by mutual written agreement of the parties prior to the Closing to ensure that any lists accurately reflect the Assigned Registered IP to be transferred pursuant to the terms and conditions of this Agreement. Seller will provide to Buyer at Closing (i) complete and accurate copies of all applications, material correspondence with Governmental Bodies or registration organizations, and other material documents related to each such item of Assigned Registered IP in Seller's possession, and (ii) a listing of all actions, filings and payment obligations due to be made to any Governmental Body within one hundred and eighty (180) days following the date of Closing with respect to each item of Assigned Registered IP. Seller has not received any notice or claim challenging the validity or enforceability of any of the Assigned Registered IP or indicating an intention on the part of any Person to bring a claim that any of the Assigned Registered IP is invalid or unenforceable, nor to Seller's knowledge is there a reasonable basis for any claim that any of the Assigned Registered IP is either invalid or unenforceable. All Assigned Registered IP has been registered or obtained in accordance with all applicable legal requirements, and Seller has timely paid all filing, examination, issuance, post registration and maintenance fees and annuities associated with or required with respect thereto. To Seller's knowledge, none of the Assigned Registered IP has been or is now involved in any interference, reissue, reexamination, opposition, cancellation or similar proceeding and no such action is or has been threatened. Seller has not taken any action or failed to take any action that would result in the abandonment, cancellation, forfeiture, relinquishment, invalidation or unenforceability of any Assigned Registered IP.

- (b) Except as set forth in Schedule 3.13(b), in connection with the operation of the Engenio Business,
- (i) To Seller's knowledge, none of Seller or any of its Affiliates has infringed, misappropriated or otherwise violated any Intellectual Property rights of any Third Party;

- (ii) there is no suit, or proceeding pending against or, to Seller's knowledge, threatened against or a written or, to Seller's knowledge, oral claim affecting, the Engenio Business (x) based upon, or challenging or seeking to deny or restrict, the rights of Seller or any of its Affiliates in any of the Assigned Intellectual Property or the Licensed Intellectual Property (collectively, the "Engenio Intellectual Property"), (y) alleging that the use of the Engenio Intellectual Property or any services provided, processes used, or products manufactured, used, imported, offered for sale or sold with respect to the Engenio Business conflict with, misappropriate, infringe or otherwise violate any Intellectual Property of any Third Party, or (z) alleging that Seller or any of its Affiliates infringed, misappropriated, or otherwise violated any Intellectual Property of any Third Party in connection with the operation of the Engenio Business; and
- (iii) (A) the Engenio Intellectual Property constitutes all the Intellectual Property and Information owned by or licensed (to the extent Seller has a right to license or sublicense Buyer thereunder without payment of a fee) to Seller or one of its Affiliates that is used or held for use primarily in the operation or conduct of the Engenio Business; (B) there exist no restrictions on the disclosure, use, license or transfer of the Engenio Intellectual Property (other than the restrictions imposed in the Intellectual Property Agreement or by applicable Law); (C) the consummation of the transactions contemplated by this Agreement will not alter, impair or extinguish any of the Engenio Intellectual Property; (D) the Licenses set forth on Schedule 2.1(h) (the "Inbound License Agreements") constitute all Licenses of Seller or one of its Subsidiaries to the Intellectual Property or Information of Third Parties that are used or held for use primarily in the operation or conduct of the Engenio Business, other than licenses for Standard Software, non-disclosure agreements and other similar agreements that are not material to the Engenio Business; and (E) the Assigned Intellectual Property and Licensed Intellectual Property constitute all of the Intellectual Property and Information owned by Seller that are necessary for the operation or conduct of the Engenio Business as currently conducted (provided that this subsection (E) shall not be interpreted as a representation regarding non-infringement, which is addressed in subsection (b)(i) above). No loss or expiration of Seller's or one of its Subsidiary's rights to use any Intellectual Property or Information licensed to Seller or any of its Subsidiaries under any Inbound License Agreement is pending or to the knowledge of Seller, threatened.
- (c) <u>Schedule 3.13(c)</u> lists the companies that design, manufacture, market, sell or distribute external storage systems products and related embedded and value-added software and other components (other than semiconductor devices) with which Seller or its Subsidiaries have entered into Contracts by which the Seller or its Subsidiaries are bound, and which Assigned Registered IP may be subject to, containing any covenant or other provision that in any way limits or restricts the ability of Seller or a Subsidiary to use, assert, enforce, or otherwise exploit any Assigned Registered IP anywhere in the world. Neither Seller nor its Subsidiaries have (i) transferred ownership of (whether a whole or partial interest), or granted any exclusive right to use, any Assigned Intellectual Property to any Person; (ii) transferred ownership of (whether a whole or partial interest) or granted any exclusive right to use any improvements to or derivative works of any Assigned Intellectual Property; or (iii) granted rights to any Person to create improvements to or derivative works of any material Assigned Intellectual Property that is or would be owned by such Person or exclusively licensed to such Person.

- (d) At the Closing, Seller will assign to Buyer free and clear of any Encumbrance the Assigned Intellectual Property and will license to Buyer the Licensed Intellectual Property, in each case in accordance with the Intellectual Property Agreement.
- (e) None of the Assigned Intellectual Property has been adjudged invalid or unenforceable in whole or part and, to Seller's knowledge, all Assigned Intellectual Property is valid and enforceable.
- (f) Seller and its Affiliates have taken reasonable actions to maintain and protect the Assigned Intellectual Property, including payment of applicable maintenance fees and filing of applicable statements of use other than certain foreign applications which Seller or an Affiliate thereof, in its reasonable business judgment, has abandoned in the ordinary course of business.
- (g) Seller and its Affiliates have taken reasonable steps to maintain the confidentiality of all Trade Secrets relating to the Engenio Business ("Engenio Trade Secrets") and other information that at any time constituted a Trade Secret relating to the Engenio Business, including taking steps to ensure that any Engenio Trade Secrets disclosed by Seller or any of its Affiliates to a Third Party are subject to the confidentiality undertakings set forth in an applicable non-disclosure agreement. To Seller's knowledge, there has been no misappropriation of any material Engenio Trade Secrets. Seller and its Affiliates have not disclosed, nor is Seller or any of its Affiliates under any contractual or other obligation to disclose, to another Person any Engenio Trade Secrets, except pursuant to an enforceable confidentiality agreement or undertaking, and, to Seller's knowledge, no Person has materially breached any such agreement or undertaking. Without limiting the generality of the foregoing, Seller has and enforces in a commercially reasonable manner a policy requiring each Business Employee and independent contractor who has participated in the creation of any Engenio Intellectual Property or have had access to any Engenio Trade Secrets to enter into non-disclosure and invention assignment agreements substantially in Seller's standard forms (which have previously been provided to Buyer).
- (h) To Seller's knowledge, no Business Employee or independent contractor of Seller or any of its Affiliates who is employed in connection with the Engenio Business is obligated under any agreement or subject to any judgment, decree or order of any court or Governmental Body, or any other restriction that could reasonably be expected to materially interfere with such Business Employee or independent contractor carrying out his or her duties for Seller or such Affiliate, as applicable, or that could reasonably be expected to materially conflict with the Assigned Intellectual Property, the Licensed Intellectual Property or the Engenio Business as presently conducted.
- (i) To Seller's knowledge, none of the Software (other than Software code currently under development) used or held for use primarily in the operation or conduct of the Engenio Business and owned, developed, marketed, distributed, licensed, sold, or otherwise made available to any Person by Seller or a Subsidiary as used in the Engenio Business (collectively, "Company Software") (i) contains any bug, defect, or error that materially and adversely affects the use, functionality, or performance of such Company Software or any product or system containing or used in conjunction with such Company Software or (ii) fails to comply in any material respect with any applicable warranty or other contractual commitment relating to the

use, functionality, or performance of such Company Software, or any product or system containing or used in conjunction with such Company Software.

- (j) To Seller's knowledge, no Company Software contains any "back door," "drop dead device," "time bomb," "Trojan horse," "virus," "worm," "spyware" or "adware" (as such terms are commonly understood in the software industry) or any other code designed or intended to have, or capable of performing or facilitating, any of the following functions: (i) disrupting, disabling, harming, or providing unauthorized access to, a computer system or network or other device on which such code is stored or installed or (ii) compromising the privacy or data security of a user or damaging or destroying any data or file without the user's consent (collectively, "Malicious Code"). Seller implements commercially reasonable measures designed to prevent the introduction of Malicious Code into Company Software, including firewall protections and regular virus scans.
- (k) Other than as set forth in Schedule 3.13(k), (i) no source code for any Company Software has been delivered, licensed, or made available to any escrow agent or other Person who is not, as of the date of this Agreement, an employee of Seller; (ii) Seller does not have any duty or obligation (whether present, contingent, or otherwise) to deliver, license, or make available the source code for any Company Software to any escrow agent or other Person; and (iii) no event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time) will, or could reasonably be expected to, result in the delivery, license, or disclosure of any source code for any Company Software to any other Person who is not, as of the date of this Agreement, an employee of Seller.
- (l) No material Company Software is subject to any "copyleft" or other obligation or condition (including any obligation or condition under any "open source" license such as the GNU Public License, Lesser GNU Public License, or Mozilla Public License) that (i) could require, or could condition the use or distribution of such material Company Software or portion thereof on, (A) the disclosure, licensing, or distribution of any source code for any portion of such material Company Software or (B) the granting to licensees of the right to make derivative works or other modifications to such material Company Software or portions thereof or (ii) could otherwise impose any limitation, restriction, or condition on the right or ability of Sellers to use, distribute or charge for any material Company Software.
- (m) There are no march-in or reversion rights to any Governmental Body or any public or private university, college, or other educational or research institution for any Assigned Intellectual Property. To Seller's knowledge, the Assigned Intellectual Property is not subject to a compulsory licensing scheme due to Seller's participation as a member or promoter of, or a contributor to, any industry standards body or similar organization. Neither Seller nor any Subsidiary has entered into a Contract with any other Person that licenses any Assigned Registered IP in accordance with any compulsory licensing scheme due to Seller's participation as a member or promoter of, or a contributor to, any industry standards body or similar organization.
- (n) Neither the execution, delivery, or performance of this Agreement nor the consummation of any of the transactions or agreements contemplated by this Agreement will, with or without notice or the lapse of time, result in, or give any other Person the right or option

to cause or declare, (i) a loss of, or Encumbrance on, any material Assigned Intellectual Property; (ii) the material release, disclosure, or delivery of the source code for any material Company Software, or of any Information, by or to any escrow agent or other Person; or (iii) the grant, assignment, or transfer by Seller to any other Person of any license or other right or interest under, to, or in any material Assigned Intellectual Property.

3.14 Product Liability and Recalls

- (a) Each Product produced or sold by Seller or a Subsidiary in connection with the Engenio Business is, and at all times up to and including the sale thereof has been, in compliance in all material respects with all applicable Laws. To Seller's knowledge, there is no material design or manufacturing defect that has been established or is being investigated with respect to any such Product.
- (b) Except as set forth in <u>Schedule 3.14(b)</u>, since January 1, 2008, there has been no action, suit, claim, inquiry, proceeding or investigation in any case by or before any court or Governmental Body pending or, to Seller's knowledge, threatened against or involving the Engenio Business relating to any Product alleged to have been designed, manufactured or sold by the Engenio Business and alleged to have been defective or improperly designed or manufactured, nor, to Seller's knowledge, has there been any pattern of product failure relating to any Product designed, manufactured or sold by the Engenio Business.
- (c) Since January 1, 2008, there has been no pending, or to Seller's knowledge, threatened recall or investigation of any Product sold by Seller or a Subsidiary in connection with the Engenio Business.

3.15 Product Warranty

- (a) <u>Schedule 3.15(a)</u> includes copies of the standard terms and conditions of sale for the Engenio Products (containing applicable guaranty, warranty and indemnity provisions and support obligations). Except as set forth in <u>Schedule 3.15(a)</u>, the products manufactured by the Engenio Business have been sold by the Engenio Business in accordance with the standard terms and conditions of sale.
- (b) <u>Schedule 3.15(b)</u> sets forth a complete and accurate listing of any Engenio Products for which one percent or more have either been returned to Seller by customers or for which Seller has received return requests from customers, since December 31, 2008.

3.16 Inventory

The Inventory is, and as of the Closing Date will be, valued in accordance with GAAP of quality and quantity usable and saleable in the ordinary course of the Engenio Business consistent with past practice, except in each case for excess, obsolete items and items of below-standard quality that have been reserved for or written down to estimated net realizable value in accordance with GAAP applied on a basis consistent with past practices as set forth in the Balance Sheet.

3.17 Customer and Suppliers

Schedule 3.17 contains a list setting forth the five (5) largest customers of the Engenio Business, by dollar amount, over the twelve (12) months ended December 31, 2010 (and the amount of sales with respect to each such customer during such twelve month period), and the five (5) largest suppliers of the Engenio Business, by dollar amount, over the twelve (12) months ended December 31, 2010 (and the amounts paid to each such supplier during such twelve month period). Seller has no knowledge of, and has not received written notice of the intention of any of such customers or suppliers to cease doing business with Seller. All purchase and sale orders and other commitments for purchases and sales made by Seller or any Subsidiary in connection with the Engenio Business have been made in the ordinary course of business in accordance with past practices, and no payments have been made to any supplier or customers or any of their respective representatives other than payments to such suppliers or their representatives for the payment of the invoiced price of supplies purchased or goods sold in the ordinary course of business.

3.18 Restrictions on the Engenio Business

Except for this Agreement, there is no agreement, judgment, injunction, order or decree materially affecting (i) Seller's or a Subsidiary's conduct of the Engenio Business as currently conducted, or (ii) to Seller's knowledge, Buyer's ability to conduct the Engenio Business after the Closing as currently conducted by Seller.

3.19 Taxes

There are no liens for Taxes upon any of the Purchased Assets other than Permitted Encumbrances. No action, proceeding or, to Seller's knowledge, investigation has been instituted against Seller or any Subsidiary (to the extent related to the Engenio Business or the Purchased Assets). Seller and each Subsidiary has duly and timely filed all Returns that it was required to file; all such Returns were correct and complete in all material respects; and all Taxes of Seller or its Subsidiaries owed or shown as due on any Return have been paid. Seller and its Subsidiaries (to the extent related to the Engenio Business or the Purchased Assets) have withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing by them to any employee, independent contractor, creditor, stockholder or other third party except where the failure to make such payment, individually or in the aggregate, is not or could not reasonably be expected to be material to the Engenio Business, taken as a whole. Neither Seller nor any Subsidiary has received any claim in writing in the last two (2) years from a Governmental Body or social security administration in a jurisdiction where any Seller or Subsidiary (to the extent related to the Engenio Business or the Purchased Assets) does not file Returns that such Seller or Subsidiary is or may be subject to taxation by that jurisdiction. None of the Purchased Assets (a) is property required to be treated as owned by another person pursuant to the provisions of Section 168(f)(8) of the U.S. Internal Revenue Code of 1954 and in effect immediately before the enactment of the Tax Reform Act of 1986, (b) constitutes "tax-exempt use property" or "tax-exempt bond financed property" within the meaning of Section 168 of the Code, (c) secures any debt the interest of which is tax-exempt under Section 103(a) of the Code, or (d) is subject to a 467 rental agreement as defined in Section 467 of the Code. Seller

(and not any of Seller's Affiliates) is the beneficial owner of the Assigned Intellectual Property and the Licensed Intellectual Property.

3.20 Brokers

Other than Goldman Sachs & Co., as to which Seller shall have full responsibility and for which Buyer shall not have any liability, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Seller or any Affiliate of Seller.

4. Representations and Warranties of Buyer

Except as set forth in Schedules attached hereto and delivered by Buyer to Seller prior to the execution of this Agreement, Buyer represents and warrants to Seller that:

4.1 Organization and Qualification

Each of Buyer and any Buyer Designee is a corporation, limited partnership, or other legal entity duly organized, validly existing and in good standing (in any jurisdiction in which such concept exists) under the Laws of the jurisdiction of its incorporation or organization and each of Buyer and any Buyer Designee has all requisite legal power and authority to carry on its business as currently conducted by it and to own or lease and operate its properties. Each of Buyer and any Buyer Designee is duly qualified to do business and is in good standing as a foreign corporation (in any jurisdiction that recognizes such concept) in each jurisdiction where the ownership or operation of its assets or the conduct of its business requires such qualification, except where the failure to be so qualified or in good standing, individually or in the aggregate, has not had and could not reasonably be expected to have a material adverse effect on Buyer or on Buyer's or any Buyer Designee's ability to consummate the transactions under this Agreement and the Collateral Agreements.

4.2 Authorization; Binding Effect

- (a) Each of Buyer and any Buyer Designee has all requisite corporate power and authority to execute, deliver and perform this Agreement and the Collateral Agreements to which it will be a party, as the case may be, and to effect the transactions contemplated hereby and thereby and the execution, delivery and performance of this Agreement and the Collateral Agreements by Buyer has been duly authorized by all requisite corporate action and, to the extent not completed on the date hereof by a Buyer Designee, will be duly authorized by all requisite corporate action.
- (b) This Agreement has been duly executed and delivered by Buyer and this Agreement is, and the Collateral Agreements to which Buyer or a Buyer Designee will be a party when duly executed and delivered by Buyer or such Buyer Designee will be, valid and legally binding obligations of Buyer or such Buyer Designee enforceable against Buyer or such Buyer Designee in accordance with their respective terms, except to the extent that enforcement of the rights and remedies created hereby and thereby may be affected by bankruptcy, reorganization,

moratorium, insolvency and similar Laws of general application affecting the rights and remedies of creditors and by general equity principles.

4.3 Non-Contravention; Consents

(a) Assuming that the consents specified in Section 4.3(b) below have been obtained, the execution, delivery and performance of this Agreement and the Collateral Agreements by Buyer and any Buyer Designee and the consummation of the transactions contemplated hereby and thereby do not and will not: (i) result in a breach or violation of any provision of Buyer's or any Buyer Designee's charter or by-laws or similar organizational document, (ii) violate or result in a breach of or constitute an occurrence of default under any provision of, result in the acceleration or cancellation of any obligation under, or give rise to a right by any party to terminate or amend its obligations under, any mortgage, deed of trust, conveyance to secure debt, note, loan, indenture, lien, lease, agreement, instrument, order, judgment, decree or other arrangement or commitment to which Buyer or any Buyer Designee is a party or by which it or its assets or properties are bound, or (iii) violate any applicable Law, order, judgment, injunction, decree, rule or regulation of any court or any Governmental Body having jurisdiction over Buyer or any Buyer Designee or any of their respective properties, other than in the case of clauses (ii) and (iii), any such violations, breaches, defaults, accelerations or cancellations of obligations or rights that, individually or in the aggregate, have not had and could not be reasonably expected to have a material adverse effect on Buyer or on Buyer's or any Buyer Designee's ability to consummate the transactions under this Agreement and the Collateral Agreements.

(b) No consent, approval, order or authorization of, or registration, declaration or filing with, any Person is required to be obtained by Buyer or any Buyer Designee in connection with the execution, delivery and performance of this Agreement or the Collateral Agreements or for the consummation of the transactions contemplated hereby or thereby, except for (i) any filings required to be made under the HSR Act and any applicable filings required under foreign antitrust Laws, and (ii) such consents, approvals, orders, authorizations, registrations, declarations or filings the failure of which to be obtained or made, individually or in the aggregate, have not had and could not reasonably be expected to have a material adverse effect on Buyer or on Buyer's or any Buyer Designee's ability to consummate the transactions under this Agreement and the Collateral Agreements.

4.4 Brokers

No broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Buyer or any Affiliate of Buyer.

4.5 Sufficiency of Funds

Buyer: (i) has sufficient funds available to pay the Purchase Price and any expenses incurred by Buyer in connection with the transactions contemplated by this Agreement or the Collateral Agreements; (ii) has the resources and capabilities (financial or otherwise) to perform its obligations hereunder and under the Collateral Agreements; and (iii) has not incurred any

obligation, commitment, restriction or liability of any kind, absolute or contingent, present or future, which would impair or adversely affect its ability to perform its obligations hereunder and under the Collateral Agreements.

5. Certain Covenants

5.1 Access and Information

- (a) Seller shall give, or cause its Subsidiaries to give, to Buyer and its Affiliates, and their respective officers, employees, accountants, counsel and other representatives reasonable access during Seller's or the applicable Subsidiary's normal business hours throughout the period prior to the Closing to all of Seller's or the applicable Subsidiary's properties, books, contracts, commitments, reports of examination and records relating to the Engenio Business, the Transferred Employees, the Purchased Assets and the Assumed Liabilities (subject to any limitations that are reasonably required to preserve any applicable attorney-client privilege or legal or contractual Third-Party confidentiality obligation). Seller shall assist, and cause its Subsidiaries to assist, Buyer and its Affiliates in making such investigation and shall cause its counsel, accountants, engineers, consultants and other non-employee representatives to be reasonably available to any of them for such purposes.
- (b) After the Closing Date, Seller and Buyer shall provide, and shall cause their respective Affiliates to provide, to each other and to their respective officers, employees, accountants, counsel and other representatives, upon request (subject to any limitations that are reasonably required to preserve any applicable attorney-client privilege or legal or contractual Third-Party confidentiality obligation), reasonable access for inspection and copying of all Business Records, Governmental Permits, Licenses, Contracts and any other information existing as of the Closing Date and relating to the Engenio Business, the Purchased Assets, the Assumed Liabilities or the Transferred Employees and shall make their respective personnel reasonably available for interviews, depositions and testimony in any legal matter concerning transactions contemplated by this Agreement, the operations or activities relating to the Engenio Business, the Purchased Assets, the Assumed Liabilities or the Transferred Employees and as otherwise may be necessary or desirable to enable the party requesting such assistance to: (i) comply with any reporting, filing or other requirements imposed by any Governmental Body; (ii) assert or defend any claims or allegations in any litigation or arbitration or in any administrative or legal proceeding other than claims or allegations that one party to this Agreement has asserted against the other; or (iii) subject to clause (ii) above, perform its obligations under this Agreement. The party requesting such information or assistance shall reimburse the other party for all reasonable and necessary out-of-pocket costs and expenses, if any, incurred by such party in providing such information and in rendering such assistance. The access to files, books and records contemplated by this Section 5.1(b) shall be during normal business hours and upon reasonable prior notice and shall be subject to such reasonable limitations as the party having custody or control thereof may impose to preserve the confidentialit
- (c) Buyer agrees to preserve all Business Records, Licenses and Governmental Permits in accordance with its corporate policies related to preservation of records. Buyer further agrees that, to the extent Business Records, Licenses or Governmental Permits are placed in

storage, they will be kept in such a manner as to make individual document retrieval possible in a reasonably expeditious manner.

5.2 Conduct of the Engenio Business

From and after the date of this Agreement and until the Closing Date, except as otherwise contemplated by this Agreement or as set forth in the Schedules hereto or as Buyer shall otherwise consent to in writing, Seller and its Subsidiaries, with respect to the Engenio Business:

- (a) will carry on the Engenio Business in the ordinary course consistent with past practice and consistent therewith use its reasonable commercial efforts to keep intact the Engenio Business, keep available the services of the Business Employees and preserve the relationships of the Engenio Business with customers, suppliers, licensors, licensees, distributors and others that have a business relationship with the Engenio Business;
- (b) in the ordinary course consistent with past practice maintain the Purchased Assets in good operating condition and repair or restore such assets as necessary for the operation of the Engenio Business in the ordinary course of business;
- (c) will not permit, other than as may be required by Law or a Governmental Body, all or any of the Purchased Assets (real or personal, tangible or intangible) presently and actively used or held for use primarily in the operation or conduct of the Engenio Business to be transferred, sold, licensed, disposed of, or subjected to any Encumbrance, other than sales of Inventory in the ordinary course of business consistent with past practice and Section 5.2(c);
- (d) will not sell Inventory outside of the ordinary course of business consistent with past practice, including with respect to pricing, discounting practices, bundling, sales volume and services levels, and will maintain Inventory sufficient to meet expected customer requirements, consistent with past practice, including sufficient raw materials, capacity and work in process in light of anticipated demand and customary cycle times and sufficient finished goods Inventory for satisfaction of customer orders on hand at Closing and Inventory will be in an amount that at the Closing shall have a value of no less than \$35 million in the aggregate as calculated in accordance with the Balance Sheet;
 - (e) will not acquire any asset that will be a Purchased Asset except in the ordinary course of business consistent with past practice;
 - (f) will not fail to pay when due any material obligation related to the Engenio Business;
- (g) will not enter into, terminate or materially extend, amend, modify or waive any right with respect to any Material Contract except for purchase orders entered in the ordinary course of business consistent with past practice;
- (h) will not sell, lease, license, abandon, permit to lapse, or otherwise transfer, or create or incur any Encumbrance on, any of the assets, securities, properties, or interests of the Engenio Business (including the Assigned Intellectual Property), including not taking any action to abandon, disclose, misuse, or misappropriate the Assigned Intellectual Property in any manner

(other than non-exclusive licenses (other than patent portfolio licenses or cross licenses) in the ordinary course of business consistent with past practice) or assert or threaten any claims with respect to the Assigned Intellectual Property;

- (i) will not incur or assume any liabilities, obligations or indebtedness for borrowed money, other than in the ordinary course of business consistent with past practice or that will constitute Excluded Liabilities;
- (j) will not increase the salaries (except for the 2011 salary increase reflected on <u>Schedule 3.10(a)(i)</u>), wage rates, other compensation or fringe benefits of, or grant any severance or termination payment (other than as required by Law) to, any Business Employee;
 - (k) fail to comply in any material respect with all Laws applicable to the Engenio Business or the Purchased Assets;
- (l) will not make, change or revoke any Tax election; file any amended Return; enter into any closing agreement; settle or compromise any Tax claim or assessment; or consent to any extension or waiver of the limitation period applicable to any claim or assessment with respect to Taxes, in each case to the extent such action could reasonably be expected to adversely affect the Purchased Assets or the Engenio Business;
- (m) will not do any other act which would cause any representation or warranty of Seller in this Agreement to be or become untrue in any material respect or intentionally omit to take any action necessary to prevent any such representation or warranty from being untrue in any material respect;
- (n) will use reasonable efforts to continue the construction of laboratories and facilities at the Transferred Premises in the manner and according to the schedule as currently contemplated by Seller immediately prior to the date hereof; and
 - (o) will not enter into any agreement or commitment with respect to any of the foregoing.

5.3 Taxes

- (a) Seller and Buyer acknowledge and agree that (i) Seller will be responsible for and will perform all Tax withholding, payment and reporting duties with respect to any wages and other compensation paid by Seller or a Subsidiary to any Business Employee in connection with the operation or conduct of the Engenio Business for any Pre-Closing Tax Period, and (ii) Buyer will be responsible for and will perform all Tax withholding, payment and reporting duties with respect to any wages and other compensation paid by Buyer or a Buyer Designee to any Transferred Employee with respect any Post-Closing Tax Period. For the avoidance of doubt, nothing in this paragraph is intended to modify or adjust the substantive liability of Buyer and Seller under this Agreement with respect to the Taxes described in this paragraph.
- (b) The allocation of the Purchase Price between the Seller and each Subsidiary shall be as set forth on <u>Schedule 5.3</u> hereto, as adjusted pursuant to this Section 5.3 (the "Entity Level

Allocation Statement."). No later than five (5) days prior to the Closing Date, Buyer shall deliver to Seller an updated Entity Level Allocation Statement.

- (c) Not later than sixty (60) days after the Closing Date, Buyer shall prepare and deliver to Seller (i) a final Entity Level Allocation Statement, and (ii) an allocation of the Purchase Price among the Purchased Assets (including a separate allocation for each separate purchase reflected on the final Entity Level Allocation Statement) in accordance with Section 1060 of the Code and the Treasury regulations promulgated thereunder (and any similar provision of state, local or foreign law, as appropriate) (the "Asset Level Allocation Statement"). Seller and Buyer shall work in good faith to resolve any disputes relating to the Asset Level Allocation Statement. If Seller and Buyer are unable to resolve any such dispute, such dispute shall be resolved promptly by a nationally recognized accounting firm acceptable to Buyer and Seller, the costs of which shall be borne equally by Buyer and Seller. The Parties agree that they will not, and will not permit any of their respective Affiliates to, take a position (except as required pursuant to any order of a Governmental Body) on any Return or in any audit or examination before any Governmental Body that is in way inconsistent with the final Entity Level Allocation Statement or the Asset Level Allocation Statement (the final Entity Level Allocation Statement, together, the "Allocation"). If the Purchase Price is adjusted pursuant to Section 9.3(d), the Allocation shall be adjusted in a manner consistent with the procedures set forth in this Section 5.3(c).
- (d) Seller shall promptly notify Buyer in writing upon receipt by Seller of notice of any pending or threatened Tax audits or assessments relating to the income, properties or operations of Seller that reasonably may be expected to relate to or give rise to a lien on the Purchased Assets or the Engenio Business. Each of Buyer and Seller shall promptly notify the other in writing upon receipt of notice of any pending or threatened Tax audit or assessment challenging the Allocation.
- (e) Seller shall deliver to Buyer at the Closing a properly executed affidavit prepared in accordance with Treasury Regulations section 1.1445-2(b) certifying Seller's non-foreign status.

5.4 Business Employees

- (a) Prior to the Closing, Seller shall update the information provided in Schedule 3.10(a)(i) as of the Closing Date.
- (b) As of the Closing Date, Buyer shall make offers of employment to at least the number of Business Employees of Seller set forth on Schedule 5.4(b) whom shall be specifically identified by Buyer prior to the Closing and whom shall include all Business Employees located in a country that has adopted the ARD Regulations in the event Buyer has made an offer of employment to one or more Business Employees in such country (such country, an "ARD Jurisdiction"), but shall not otherwise include any Business Employee located in any country that is an ARD Jurisdiction (the "Offered Employees"). Seller and any applicable Subsidiary shall cooperate and assist in facilitating Buyer's or a Buyer Designee's offers and will not take any action, or cause any of the Subsidiaries to take any action, which would impede, hinder, interfere or otherwise compete with Buyer's or a Buyer Designee's effort to hire any Business Employees.

Promptly after the date hereof, Seller will provide to Buyer all information not provided in Schedule 3.10(a) required to be disclosed by applicable Law of the jurisdiction in which the Business Employee is located in connection with the sale of the Engenio Business. To the extent permitted by applicable Law, including data privacy and data protection Laws, Seller agrees to provide Buyer with such information reasonably requested by Buyer to assist it with complying with the terms of this Section 5.4 and to assist Buyer with determining the wages paid to the Transferred Employees (as defined below) with respect to the period beginning on January 1, 2011 and ending on the Closing Date. Seller shall be responsible for any employment action related to any Business Employee who is not an Offered Employee. To the extent any notification or consultation requirements or works council negotiation procedures are imposed by applicable Law with regard to the transfer of Business Employees to Buyer or any of its Affiliates, Seller and Buyer agree to cooperate to ensure that such notification or consultation requirements or works council negotiation procedures are timely completed. Without limiting the foregoing, each Party shall comply with all applicable Laws in connection with the transfer of the Offered Employees to Buyer or a Buyer Designee, including with respect to notice, consultation and other procedural requirements. The parties will enter into an Assignment and Bill of Sale and Assumption Agreement or other appropriate documentation for relevant jurisdictions outside the United States where necessary or appropriate for the transfer of such Offered Employees and shall cooperate to complete all requisite consultation and related objection periods prior to the Closing Date. Any Offered Employee who accepts Buyer's offer of employment and commences employment with Buyer or a Buyer Designee shall be referred to as a "Transferred Employee." Employment of the Transferred Employees with Buyer or a Buyer Designee shall be effective as of the day following the close of business on the Closing Date; provided, that with respect to Offered Employees employed outside the European Union who, as of the Closing Date, are on Seller-approved leave (the "Leave Employees"), employment with Buyer or a Buyer Designee shall be effective as of (i) with respect to Leave Employees absent due to leave that is not protected under applicable Law, within 90 days after the Closing Date or (ii) with respect to Leave Employees absent due to protected leave under applicable Law, no later than the first Business Day following the end of the protected leave period.

(c) Where terms are not dictated by applicable Law, Buyer or a Buyer Designee shall provide, or shall cause to be provided, to Transferred Employees, until at least March 31, 2012 during their employment with Buyer or a Buyer Designee, at a minimum, the same base salaries or, as applicable, base wage rates, offered by Seller or the applicable Subsidiary immediately prior to the Closing Date (but taking into account the 2011 salary increases) as set forth on Schedule 3.10(a)(i). Buyer or a Buyer Designee shall provide, or shall cause to be provided, to Transferred Employees either (at Buyer's discretion) employee benefits that are no less favorable in the aggregate than either (i) those benefits provided to similarly situated employees of Buyer or the applicable Buyer Designee (taking into account employee's seniority and service with Seller or Buyer or their respective Affiliates, as applicable) or (ii) the employee benefits that they were offered by Seller or the applicable Subsidiary immediately prior to the Closing Date as set forth on Schedule 3.10(b). Except as expressly set forth in this Section 5.4, no Benefit Plans or assets of any Benefit Plan shall be transferred to Buyer or any Affiliate of Buyer. Buyer will take all action necessary to ensure that, to the extent permitted under applicable Buyer or Buyer Designee Benefit Plans, such Benefit Plans shall recognize (i) for purposes of satisfying any deductibles, co-pays and out-of-pocket maximums during the coverage period that includes the Closing Date, any payment made by any Transferred Employee towards deductibles, co-pays and

out-of-pocket maximums in any health or other insurance plan of Seller or a Subsidiary during the coverage period that includes the Closing Date and (ii) for purposes of determining eligibility to participate and vesting and, in the case of any Buyer severance plan or program (if any), benefit accruals, all service with Seller or a Subsidiary prior to the Closing, including service with predecessor employers that was recognized by Seller or a Subsidiary, provided that such service shall not be recognized to the extent such recognition would result in a duplication of benefits. Buyer or the applicable Buyer Designee will continue to provide tuition assistance to those Transferred Employees who are receiving such benefits as of the Closing Date for the current academic session, in each case as set forth on Schedule 5.4(c). Buyer or the applicable Buyer Designee will honor the terms and conditions of Seller's international assignee program, including repatriation upon completion of assignment, completion bonuses, Tax equalization and Tax return preparation, with respect to Transferred Employees who are on international assignment as of the Closing Date, in each case as set forth on Schedule 5.4(c), except that these costs shall be allocated between the parties based on the portion of the international assignment occurring before or on the Closing Date (which shall be Seller's or the applicable Subsidiary's obligation) and after the Closing Date (which shall be Buyer's or the applicable Buyer Designee's obligation).

- (d) Seller and Buyer intend that the transactions contemplated by this Agreement shall not constitute a severance of employment, under the terms of any Benefit Plan of Seller of any Subsidiary, of any Transferred Employee prior to or upon the consummation of the transactions contemplated hereby and that such employees will have continuous and uninterrupted employment immediately before and immediately after the Closing Date. Notwithstanding anything to the contrary in this Agreement, Buyer shall provide, at a minimum, severance benefits substantially equivalent to the benefits contained in the plans listed or as described on Schedule 5.4(d) to Transferred Employees whose employment is terminated involuntarily by Buyer on or before March 31, 2012 other than terminations in circumstances that would not require payments of severance benefits under Seller's severance plan.
- (e) To the extent permitted under applicable Buyer Benefit Plans, (i) Buyer shall use commercially reasonable efforts to waive any pre-existing condition exclusion (to the extent such exclusion was waived under applicable health and Welfare Plans offered to the Transferred Employees by Seller or a Subsidiary) and proof of insurability, and (ii) the medical and dental plans maintained by Buyer and any Affiliate of Buyer shall recognize as dependents of the Transferred Employees the dependents recognized by Seller's or the applicable Subsidiary's medical and dental plans.
- (f) As soon as practicable following the Closing Date, Buyer shall cause one or more defined contribution savings plans intended to qualify under sections 401(a) and 401(k) of the Code (the "Buyer Savings Plan") to provide for the receipt of Transferred Employees' lump sum cash distributions, in the form of an eligible rollover distribution from the LSI Corporation 401(k) Plan, provided such rollovers are made at the election of the Transferred Employees and in accordance with the terms of the Buyer Savings Plan. Seller shall cause the LSI Corporation 401(k) Plan to fully vest Transferred Employees in their accounts immediately prior to the Closing and permit the Transferred Employees to elect a lump sum cash distribution of benefits accrued through the Closing Date in accordance with the Code.

- (g) Seller shall make and be responsible for incentive compensation payments, if any, earned by the Transferred Employees for the period from January 1, 2011 to and including the Closing Date under the applicable incentive plans in effect for any such period (including any pro rata amount with respect to such period under a plan or program ending or vesting on or after the Closing Date). Buyer shall not assume or otherwise become liable for, and Seller shall not transfer to Buyer, any liabilities of Seller with respect to accrued but unused vacation and sabbatical leave (collectively, the "Accrued Amounts"). At the Closing, Seller shall pay to each Transferred Employee the Accrued Amount with respect to such Transferred Employee; provided, that with respect to Transferred Employees in the European Union, Seller shall pay to Buyer each such Transferred Employee's applicable Accrued Amount, including for the avoidance of doubt, any applicable employee's and employer's social contributions, in each case to the extent required pursuant to applicable Law.
- (h) As soon as practicable following the Closing, Seller shall provide Buyer with a schedule setting forth the number of employees and the work location of each employee of Seller or any Subsidiary in the United States who terminated employment within the ninety (90) day period prior to the Closing Date.
- (i) Notwithstanding anything herein to the contrary, nothing in this Agreement shall require Buyer or a Buyer Designee to employ any Business Employees, or to employ any Transferred Employee on anything other than an at-will basis, terminable at any time with or without cause unless required otherwise under applicable Law. Nothing in this Section 5.4, expressed or implied, shall confer upon any employee or former employee of Seller or any Subsidiary or related entities (including, without limitation, the Transferred Employees) any rights or remedies (including, without limitation, any right to employment or continued employment for any specified period) of any nature or kind whatsoever, under or by reason of this Section 5.4. It is expressly agreed that the provisions of this Section 5.4 are not intended to be for the benefit of or otherwise be enforceable by, any third party, including, without limitation, any Transferred Employees. No provision of this Section 5.4 shall create any rights in any such persons in respect of any benefits that may be provided under any Benefit Plan or any plan or arrangement which may be established or maintained by Buyer, shall be construed to establish, amend, or modify an Benefit Plan or any other benefit plan, program, agreement or arrangement nor shall require Seller, Buyer or any Affiliate of Seller or Buyer to continue or amend any particular benefit plan and any such plan may be amended or terminated in accordance with its terms and applicable Law.
- (j) Seller or a Subsidiary shall use reasonable commercial efforts to cause each Transferred Employee located in India to enter into a general release of claims against Seller in customary form.
- (j) The Seller and its Subsidiaries shall execute and perform all such deeds, documents, and acts as may be reasonably required to continue to make available to Transferred Employees in India their respective account balances in the Provident Fund Plan and the Gratuity Fund following their transfer to Buyer or a Buyer Designee.
 - (k) None of the Business Employees in India are "workmen" as defined under Indian Law.

5.5 Collateral Agreements; Leased Equipment; Premises

- (a) On or prior to the Closing Date, Buyer or a Buyer Designee shall execute and deliver to Seller, and Seller or the applicable Subsidiary shall execute and deliver to Buyer or a Buyer Designee, the Collateral Agreements. In addition, on or prior to the Closing Date Buyer and Seller shall negotiate in good faith an arm's-length customer and technical support agreement from Buyer to Seller related to Engenio Products currently used by Seller in its information technology infrastructure.
- (b) Promptly after the date hereof, Seller shall provide Buyer with the costs and other terms applicable to the Leased Equipment and Buyer shall decide whether such Leased Equipment will (a) transfer to Buyer or a Buyer Designee as of the Closing Date by Buyer or a Buyer Designee assuming the leases for such equipment, (b) become the property of Buyer or a Buyer Designee as of the Closing Date by Buyer or a Buyer Designee paying for the costs of purchasing such equipment pursuant to the leases (the "Purchased Leased Equipment"), or (c) remain the property of Seller or a Subsidiary as of the Closing Date (the "Excluded Leased Equipment").
- (c) Prior to the Closing Date, parties agree to negotiate in good faith to demarcate in a manner reasonably acceptable to Buyer and Seller (i) the portion of the Transferred Premises that will be leased on a transition basis to Seller pursuant to the Transition Services Agreement, and (ii) the portion of the Assumed Leases to be subleased by Seller or its Subsidiary to Buyer or a Buyer Designee.

5.6 Regulatory Compliance; Post-Closing Cooperation

(a) Subject to Section 5.6(b), upon the terms and subject to the conditions set forth in this Agreement, each of the parties agrees to use its reasonable commercial efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the transactions contemplated by this Agreement, including using reasonable commercial efforts to accomplish the following: (i) the taking of all acts necessary to cause the conditions to Closing to be satisfied as promptly as practicable; (ii) the obtaining of all necessary actions or nonactions, waivers, consents and approvals from Governmental Bodies and the making of all necessary registrations and filings (including filings with Governmental Bodies, if any) and the taking of all steps as may be necessary to obtain an approval or waiver from, or to avoid an action or proceeding by any Governmental Body; (iii) the obtaining of all necessary consents, approvals or waivers from Third Parties; (iv) the defending of any lawsuits or other legal proceedings, whether judicial or administrative, challenging this Agreement or the Collateral Agreements or the consummation of the transactions contemplated hereby or thereby, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Body vacated or reversed; and (v) the execution and delivery of any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement and the Collateral Agreements.

(b) Seller and Buyer shall timely and promptly make all filings which may be required by each of them in connection with the consummation of the transactions contemplated hereby under the HSR Act and any state, foreign or multinational antitrust legislation or by any other foreign or multinational governmental authority, and the parties shall respectively use all reasonable commercial efforts to cause the receipt of approval of, or prompt termination or expiration of the applicable waiting period under such Laws. Seller and Buyer agree that their respective initial filings under the HSR Act and in respect of any foreign antitrust approval shall be made no later than the fifth (5th) Business Day following the date of this Agreement. Each party shall furnish to the other such necessary information and assistance as the other party may reasonably request in connection with the preparation of any necessary filings or submissions by it to any U.S. federal or state or foreign or multinational governmental agency, including any filings necessary under the provisions of the HSR Act. Each party shall provide the other party the opportunity to make copies of all correspondence, filings or communications (or memoranda setting forth the substance thereof) between such party or its representatives, on the one hand, and the Federal Trade Commission (the "FTC"), the Antitrust Division of the United States Department of Justice (the "Antitrust Division") or any state, foreign or multinational Governmental Body or members of their respective staffs, on the other hand, with respect to this Agreement or the transactions contemplated hereby. Each party agrees to inform promptly the other party of any communication made by or on behalf of such party to, or received by or on behalf of such party from, the FTC, the Antitrust Division or any other state, foreign or multinational Governmental Body regarding any of the transactions contemplated hereby.

5.7 Contacts with Suppliers and Customers

Prior to the Closing, Seller and Buyer agree to cooperate to prepare a communications plan for business partners of the Engenio Business, and in contacting any suppliers to, or customers of, the Engenio Business in connection with or pertaining to any subject matter of this Agreement or the Collateral Agreements and to facilitate the transition of the Engenio Business, including the preparation of letters to all customers, suppliers, distributors and other business partners of the Engenio Business to notify them of the Closing and provide information regarding the transition of the Engenio Business to Buyer. The Seller will be responsible for contacting parties to any Contracts for which consent is required in connection with their assignment pursuant to this Agreement. Notwithstanding anything to the contrary contained herein, this Agreement shall not affect Seller's continuing right to contact customers and suppliers in connection with the operation or conduct of the Engenio Business nor Buyer's continuing right to contact customers and suppliers in connection with the operation or conduct of its business.

5.8 Use of the Seller Name

- (a) Buyer and Seller agree as follows:
- (i) Except as provided below, immediately after the Closing Date, Buyer and any Buyer Designee shall cease using "<u>LSI</u>," or "<u>LSI Corporation</u>" or other similar mark (the "<u>Seller Name</u>") and any other trademark, design or logo previously or currently used by Seller or any of its Affiliates (other than those that are transferred pursuant to the Intellectual Property

Agreement) in all invoices, letterhead, advertising and promotional materials, office forms or business cards;

- (ii) Except as provided below, within three (3) months after the Closing Date, Buyer shall (A) remove any other trademark, design or logo previously or currently used by Seller or any of its Affiliates from all buildings, signs and vehicles of the Engenio Business; and (B) cease using the Seller Name and any other trademark, design or logo previously or currently used by Seller or any of its Affiliates (other than those that are transferred pursuant to the Intellectual Property Agreement) in electronic databases, web sites, product instructions, packaging and other materials, printed or otherwise.
- (iii) Buyer and Buyer Designees shall not be required at any time to remove the Seller Name and any other trademark, design or logo previously or currently used by Seller or any of its Affiliates from Inventory of the Engenio Business that is in existence as of the Closing Date, nor shall Buyer nor Buyer Designees be required at any time to remove such Seller Name and any such other trademark, design or logo from schematics, plans, manuals, drawings, machinery, tooling including hand tools, and the like of the Engenio Business in existence as of the Closing Date to the extent that such instrumentalities are used in the ordinary internal operation or conduct of the Engenio Business and are neither generally observed by the public nor intended for use as means to effectuate or enhance sales;
- (iv) Buyer and Buyer Designees shall have the right to sell existing Inventory and to use existing packaging, labeling, containers, supplies, advertising materials, technical data sheets and any similar materials bearing the Seller Name or any other trademark, design or logo previously or currently used by Seller or any of its Affiliates until the earlier of (A) one year after the Closing Date or (B) the depletion of existing Inventory;
- (v) Buyer and Buyer Designees shall use Reasonable Efforts (as defined below) to remove the Seller Name and any other trademark, design or logo previously or currently used by Seller or any of its Affiliates (other than those that are transferred pursuant to the Intellectual Property Agreement) from those assets of the Engenio Business (such as, but not limited to, tools, molds, and machines) used in association with the Engenio Products or otherwise reasonably used in the operation or conduct of the Engenio Business after the Closing. For the purposes of this Section 5.8(a)(v), "Reasonable Efforts" means Buyer and Buyer Designees shall remove the Seller Name from such assets but only at such time when such asset is not operated or otherwise is taken out of service in the normal course of business due to regular maintenance or repair (but only for such repairs or maintenance where such removal could normally be undertaken, for example, repair or maintenance of a mold cavity) whichever occurs first; provided that, in no event shall Buyer or any Buyer Designee use the Seller Name after the date which is one (1) year from the Closing Date. Buyer and Buyer Designees shall not be required to perform such removal on such assets that are not or are no longer used to manufacture the Engenio Products or other parts, or if discontinuance of use of such assets is reasonably anticipated during such time period, or from assets stored during that period provided that such marks are removed upon such asset's return to service or prior to their sale or other disposition.

- (vi) Seller hereby grants to Buyer and Buyer Designees a limited right to use the Seller Name and associated trademarks, designs and logos as specified in, and during the periods, if any, specified in clauses (i) (v) above.
- (vii) Buyer and its Affiliates shall also have the right to use (in a factual manner that constitute fair use pursuant to applicable Law) the Seller Name solely to the extent necessary to communicate that the Engenio Products were formerly owned by Seller.
 - (b) In no event shall Buyer or any Affiliate of Buyer advertise or hold itself out as LSI or an Affiliate of LSI after the Closing Date.

5.9 Non-Solicitation or Hiring of Transferred Employees

None of Seller, any of its representatives or any of its Affiliates will at any time prior to the date that is one year after the date hereof, directly or indirectly, solicit the employment of or hire any Transferred Employee without Buyer's prior written consent. The term "solicit the employment" shall not be deemed to include generalized searches for employees through media advertisements, employment firms or otherwise that are not focused on or directed to Transferred Employees. This restriction shall not apply to any employee whose employment was involuntarily terminated other than for cause by Buyer a Buyer Designee, or their respective successors, after the Closing.

5.10 No Negotiation or Solicitation

Prior to the Closing Date, Seller and its Affiliates will not (and Seller will cause each of its employees, officers, representatives and agents or advisors not to and shall cause its Affiliates to cause employees, officers, representatives and agents or advisors not to) directly or indirectly (a) solicit, initiate, entertain, encourage or accept the submission of any proposal, offer or any discussions relating to or that might reasonably be expected to lead to or result in any proposal or offer from any Person relating to the direct or indirect acquisition of the Engenio Business or any portion of the Purchased Assets (other than purchases of Engenio Products or services from the Engenio Business in the ordinary course of business consistent with past practice), or (b) participate in any discussions or negotiations regarding the Engenio Business, furnish any information with respect thereto, or assist or participate in, or facilitate or encourage in any other manner any effort or attempt by any Person to do or seek any of the foregoing. Seller will notify Buyer if any Person makes any proposal, offer, inquiry or contact with respect to any of the foregoing within two (2) Business Days after receipt of any such offer or proposal, including the identity of the Person making such proposal, offer, inquiry or contact and all material terms thereof.

5.11 Non-Competition

(a) Seller agrees that, as part of the consideration for the payment of the Purchase Price, for a period of three (3) years immediately following the Closing Date, neither Seller nor any of its Affiliates will, directly or indirectly, as a principal, stockholder, joint venturer or otherwise, operate, perform or have any ownership interest in any business that designs, develops, manufactures, markets, sells, installs or distributes products in competition with the Engenio Business, except that Seller may (i) continue the activities of its RAID adapter business

which develops LSI® MegaRAID® and 3ware® storage controllers and software and the ONStor™ clustered NAS gateway and non-integrated file storage products business of Seller which develops ONStor™ Products, and (ii) purchase or otherwise acquire by merger, purchase of assets, stock (including investing as a minority shareholder), controlling interest or otherwise any Person or business or engage in any similar merger and acquisition activity with any Person the primary business of which is not in competition with the Engenio Business, provided that Seller may not provide any such business access or license to any of the Assigned Intellectual Property for use in any business or product line that competes with the Engenio Business. For the purposes of this Section 5.11(a), ownership of securities of a company whose securities are publicly traded under a recognized securities exchange not in excess of 5% of any class of such securities shall not be considered to be competition with the Engenio Business, and a Person shall not be considered to be in the "primary business" of competing with the Engenio Business if such Person derives less than 20% of its revenues from products that compete with the Engenio Business. For the avoidance of doubt, the parties agree that the agreements and limitations set forth in this Section 5.11 shall not apply to any entity that acquires all or part of Seller in any transaction.

- (b) Seller acknowledges that the restrictions set forth in Section 5.11(a) constitute a material inducement to Buyer's entering into and performing this Agreement. Seller further acknowledges, stipulates and agrees that a breach of such obligation could result in irreparable harm and continuing damage to Buyer for which there may be no adequate remedy at Law and further agrees that in the event of any breach of said obligation, Buyer may be entitled to injunctive relief and to such other relief as is proper under the circumstances.
- (c) If any provision contained in this Section shall for any reason be held invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provisions of this Section 5.11, but this Section 5.11 shall be construed as if such invalid, illegal or unenforceable provision had never been contained herein. It is the intention of the parties that if any of the restrictions or covenants contained herein is held to cover a geographic area or to be for a length of time which is not permitted by applicable Law, or in any way construed to be too broad or to any extent invalid, such provision shall not be construed to be null, void and of no effect, but to the extent such provision would be valid or enforceable under applicable Law, a court of competent jurisdiction shall construe and interpret or reform this Section 5.11 to provide for a covenant having the maximum enforceable geographic area, time period and other provisions (not greater than those contained herein) as shall be valid and enforceable under such applicable Law.

5.12 Post Closing Remittances

If on or after the Closing Date, either Party receives a payment from a Third Party (including a customer of the Engenio Business) that, pursuant to the terms hereof, should have been paid to the other Party, the Party who receives the payment agrees to hold in trust and remit such payment to the Party entitled thereto within five (5) Business Days of such receipt.

5.13 Prorations and Adjustments

- (a) Except as otherwise expressly provided herein, all ordinary course expenses for (i) rents and other charges or amounts payable included in the Purchased Assets and transferred to Buyer hereunder and (ii) gas, electricity, water, sewer, rent and telephone charges at the Leased Premises and the Transferred Premises, in each case, for the period prior to the Closing Date, will be for the account of Seller and for the period on and after the Closing Date shall be for the account of Buyer. If any Party actually makes any payments that are, in whole or in part, designated as payments for the period allocated to the other Party under this Section 5.13, such other Party shall promptly reimburse such amounts to the Party so making such payments.
- (b) For purposes of calculating prorations, Buyer shall be deemed to own the Purchased Assets, and, therefore be responsible for the expense thereof, as of 12:01 a.m. local time on the day after the Closing Date. All prorations shall be made on the basis of the actual number of days of the month that shall have elapsed as of the Closing Date and based upon a 365-day year. The amount of the prorations shall be subject to adjustment after the Closing, as and when complete and accurate information becomes available, and the Parties agree to cooperate and use their good faith efforts to make such adjustments.

5.14 Notification of Certain Matters

Seller shall give prompt written notice to Buyer of (a) the occurrence or non-occurrence of any event, the occurrence or non-occurrence of which could reasonably be expected to cause any representation or warranty of Seller in this Agreement to be untrue or inaccurate at or prior to the Closing in any material respect and (b) any failure of Seller in any material respect to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder, and Buyer shall give prompt written notice to Seller of (x) the occurrence or non-occurrence of any event, the occurrence or non-occurrence of which is could reasonably be expected to cause any representation or warranty of Buyer in this Agreement to be untrue or inaccurate at or prior to the Closing in any material respect and (y) any failure of Buyer in any material respect to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder; provided, however, that the delivery of any notice pursuant to this Section 5.14 shall not limit or otherwise affect any remedies available to the Party receiving such notice.

5.15 Title Insurance

Seller will cooperate with Buyer in Buyer's obtaining title insurance policies with respect to such of the Transferred Premises and Leased Premises as Buyer shall determine. In addition, Seller will provide an affidavit to Buyer's title insurance company in reasonable and customary form sufficient for such title insurance company to remove the "standard exceptions," including any exception for mechanics' or material suppliers' liens and any exception for "parties in possession," and sufficient for the title insurance company to insure any "gap" period prior to recording the instrument of transfer due to Seller's actions.

5.16 Closing Statement

Within ten Business Days after the Closing Date, Seller shall deliver to Buyer a schedule calculated as of the Closing Date of those line items of the Purchased Assets and Assumed Liabilities that are being transferred or assumed, as applicable, pursuant to this Agreement and that would be required to be set forth on a balance sheet prepared in accordance with GAAP (the "Closing Statement"). At least five Business Days prior to the Closing Date, Seller shall deliver a good faith estimate of the amounts to be set forth on the Closing Statement.

5.17 Indian Assets

Buyer and Seller will take all actions reasonably required to de-bond and bond Purchased Assets located in India to the extent required by, or otherwise beneficial under, applicable Law.

6. Confidential Nature of Information

6.1 Confidentiality Agreement

Buyer agrees that the Confidentiality Agreement shall apply to (a) all documents, materials and other information that it shall have obtained regarding Seller or its Affiliates during the course of the negotiations leading to the consummation of the transactions contemplated hereby (whether obtained before or after the date of this Agreement), any investigations made in connection therewith and the preparation of this Agreement and related documents and (b) all analyses, reports, compilations, evaluations and other materials prepared by Buyer or its counsel, accountants or financial advisors that contain or otherwise reflect or are based upon, in whole or in part, any of the provided information; provided, however, that subject to Section 6.2(a), the Confidentiality Agreement shall terminate as of the Closing and shall be of no further force and effect thereafter with respect to information of Seller or its Affiliates the ownership of which is transferred to Buyer or a Buyer Designee.

6.2 Seller's Proprietary Information

(a) Except as provided in Section 6.2(b), after the Closing and for a period of five (5) years following the Closing Date, Buyer agrees that it will keep confidential all of Seller's and its Affiliates' Information that is received from, or made available by, Seller in the course of the transactions contemplated hereby, and marked or identified at the time of disclosure as the proprietary or confidential information of Seller ("Seller Proprietary Information"), including, for purposes of this Section 6.2, information about the Engenio Business's business plans and strategies, marketing ideas and concepts, especially with respect to unannounced products and services, present and future product plans, pricing, volume estimates, financial data, product enhancement information, business plans, marketing plans, sales strategies, customer information (including customers' applications and environments), market testing information, development plans, specifications, customer requirements, configurations, designs, plans, drawings, apparatus, sketches, software, hardware, data, prototypes, connecting requirements or other technical and business information, except for such Seller Proprietary Information the ownership of which is transferred to Buyer or a Buyer Designee as part of the Purchased Assets.

- (b) Notwithstanding the foregoing, such Seller Proprietary Information shall not be deemed confidential and Buyer shall have no obligation with respect to any such Seller Proprietary Information that:
- (i) at the time of disclosure was already known to Buyer other than as a result of this transaction, free of restriction as evidenced by documentation in Buyer's possession;
 - (ii) is or becomes publicly known through publication, inspection of a product, or otherwise, and through no negligence or other wrongful act of Buyer;
 - (iii) is received by Buyer from a Third Party without similar restriction and without breach of any agreement;
 - (iv) to the extent it is independently developed by Buyer; or
 - (v) is, subject to Section 6.2(c), required to be disclosed under applicable Law or judicial process.
- (c) If Buyer (or any of its Affiliates) is requested or required (by oral question, interrogatory, request for information or documents, subpoena, civil investigative demand or similar process) to disclose any Seller Proprietary Information, Buyer will promptly notify Seller of such request or requirement and will cooperate with Seller such that Seller may seek an appropriate protective order or other appropriate remedy. If, in the absence of a protective order or the receipt of a waiver hereunder, Buyer (or any of its Affiliates) is in the opinion of Buyer's counsel compelled to disclose the Seller Proprietary Information or else stand liable for contempt or suffer other censure or penalty, Buyer (or its Affiliate) may disclose only so much of the Seller Proprietary Information to the party compelling disclosure as is required by Law. Buyer will exercise its (and will cause its Affiliates to exercise their) reasonable commercial efforts to obtain a protective order or other reliable assurance that confidential treatment will be accorded to such Seller Proprietary Information.

6.3 Buyer's Proprietary Information

(a) Except as provided in Section 6.3(b), after the Closing Date and for a period of five (5) years thereafter, Seller agrees that it will keep confidential all of Seller's and its Affiliates' Information that is received from, or made available by, Buyer in the course of the transactions contemplated hereby and marked or identified at the time of disclosure as the proprietary or confidential information of Buyer, or the ownership of which or exclusive use of which is transferred to Buyer as part of the Purchased Assets (collectively, "Buyer Proprietary Information"), including, for purposes of this Section 6.3, information about the Engenio Business's business plans and strategies, marketing ideas and concepts, especially with respect to unannounced products and services, present and future product plans, pricing, volume estimates, financial data, product enhancement information, business plans, marketing plans, sales strategies, customer information (including customers' applications and environments), market testing information, development plans, specifications, customer requirements, configurations, designs, plans, drawings, apparatus, sketches, software, hardware, data, prototypes, connecting requirements, other technical and business information and information regarding Business Employees.

- (b) Notwithstanding the foregoing, such Buyer Proprietary Information regarding the Engenio Business shall not be deemed confidential and Seller shall have no obligation with respect to any such Buyer Proprietary Information that:
 - (i) is or becomes publicly known through publication, inspection of a product, or otherwise, and through no negligence or other wrongful act of Seller;
 - (ii) is received by Seller after the Closing Date from a Third Party without similar restriction and without breach of any agreement; or
 - (iii) is, subject to Section 6.3(c), required to be disclosed under applicable Law or judicial process.
- (c) If Seller (or any of its Affiliates) is requested or required (by oral question, interrogatory, request for information or documents, subpoena, civil investigative demand or similar process) to disclose any Buyer Proprietary Information regarding the Engenio Business, Seller will promptly notify Buyer of such request or requirement and will cooperate with Buyer such that Buyer may seek an appropriate protective order or other appropriate remedy. If, in the absence of a protective order or the receipt of a waiver hereunder, Seller (or any of its Affiliates) is in the opinion of Seller's counsel compelled to disclose the Buyer Proprietary Information or else stand liable for contempt or suffer other censure or penalty, Seller (or its Affiliate) may disclose only so much of the Buyer Proprietary Information to the party compelling disclosure as is required by Law. Seller will exercise its (and will cause its Affiliates to exercise their) reasonable commercial efforts to obtain a protective order or other reliable assurance that confidential treatment will be accorded to such Buyer Proprietary Information.

6.4 Confidential Nature of Agreements

Except to the extent that disclosure thereof is required under accounting, stock exchange or federal securities or labor relations Laws disclosure obligations or pursuant to legal process, both parties agree that the terms and conditions of this Agreement, the Collateral Agreements and all Schedules, attachments and amendments hereto and thereto shall be considered confidential or proprietary information protected under this Article 6. Notwithstanding anything in this Article 6 to the contrary, in the event that any such Information is also subject to a limitation on disclosure or use contained in another written agreement between Buyer and Seller or either of their respective Affiliates that is more restrictive than the limitation contained in this Article 6, then the limitation in such agreement shall supersede this Article 6.

7. Closing

At the Closing, the following transactions shall take place:

7.1 Deliveries by Seller or the Subsidiaries

On the Closing Date, Seller shall, or shall cause a Subsidiary to, execute and deliver to Buyer or a Buyer Designee the following:

(a) the Collateral Agreements;

- (b) the consents, waivers or approvals identified by Buyer on Schedule 7.1(b) (the "Required Closing Consents");
- (c) a certificate of an appropriate officer of Seller, dated the Closing Date, certifying to the fulfillment of the conditions set forth in Sections 8.2(a) and (b), and an incumbency certificate of an Assistant Secretary of Seller, dated the Closing Date, in customary form; and
- (d) all such other bills of sale, assignments and other instruments of assignment, transfer or conveyance as Buyer or a Buyer Designee may reasonably request or as may be otherwise necessary to evidence and effect the sale, transfer, assignment, conveyance and delivery of the Purchased Assets to Buyer or a Buyer Designee or to put Buyer or a Buyer Designee in actual possession or control of the Purchased Assets; provided that all information (including documents) capable of electronic transmission will be transmitted to Buyer or the applicable Buyer Designee in such manner, in which case such information shall not be transferred in any tangible form, and any inadvertent transfer of a tangible manifestation of such information shall promptly be returned to Seller or the applicable Selling Subsidiary upon discovery of Buyer's or such Buyer Designee's receipt thereof.

7.2 Deliveries by Buyer or a Buyer Designee

On the Closing Date, Buyer shall, or shall cause a Buyer Designee to, execute and deliver to Seller or a Subsidiary the following:

- (a) the Purchase Price;
- (b) the Collateral Agreements;
- (c) a certificate of an appropriate officer of Buyer, dated the Closing Date, certifying to the fulfillment of the conditions set forth in Sections 8.3(a) and (b), and an incumbency certificate of an Assistant Secretary of Seller of Buyer, dated the Closing Date, in customary form; and
- (d) all such other documents and instruments as Seller or a Subsidiary may reasonably request or as may be otherwise necessary or desirable to evidence and effect the assumption by Buyer or a Buyer Designee of the Assumed Liabilities.

7.3 Closing Date

The Closing shall take place at the offices of LSI, 1621 Barber Lane, Milpitas, California, at 10:00 a.m. local time within three (3) Business Days following the date on which the last of the conditions specified in Article 8 to be satisfied or waived has been satisfied or waived (other than conditions which can only be satisfied on the Closing Date, but subject to the satisfaction or waiver of such conditions), or at such other place or time or on such other date as Seller and Buyer may agree upon in writing (such date and time being referred to herein as the "Closing Date"); provided, however, that notwithstanding the foregoing, in no event shall the Closing Date occur prior to May 6, 2011.

7.4 Contemporaneous Effectiveness

All acts and deliveries prescribed by this Article 7, regardless of chronological sequence, will be deemed to occur contemporaneously and simultaneously on the occurrence of the last act or delivery, and none of such acts or deliveries will be effective until the last of the same has occurred.

8. Conditions Precedent to Closing

8.1 General Conditions

The respective obligations of Buyer and Seller to effect the Closing of the transactions contemplated hereby are subject to the fulfillment, prior to or at the Closing, of each of the following conditions:

- (a) <u>Legal Proceedings</u>. No Governmental Body shall have enacted, issued, promulgated, enforced or entered any statute, rule, regulation, non-appealable judgment, decree, injunction or other order which is in effect on the Closing Date and which prohibits, restricts or delays consummation of the transactions contemplated by this Agreement or the Collateral Agreements and there shall be no pending lawsuit, claim or legal action relating to the transactions contemplated by this Agreement or any of the Collateral Agreements which seeks to prohibit or restrict the transactions contemplated by this Agreement.
- (b) <u>Antitrust Laws</u>. Any applicable waiting period or approvals of a Governmental Body under the HSR Act or applicable foreign antitrust Laws, respectively, relating to the transactions contemplated by this Agreement or the Collateral Agreements shall have expired or been terminated.

8.2 Conditions Precedent to Buyer's Obligations

The obligations of Buyer to effect the Closing of the transactions contemplated hereby are subject to the fulfillment, prior to or at the Closing, of each of the following conditions, any of which may be waived in writing by Buyer:

(a) Representations and Warranties of Seller True and Correct at Closing. The representations and warranties of Seller contained in this Agreement or in any certificate delivered pursuant to the provisions of this Agreement that are qualified by the words "material," "Seller Material Adverse Effect" and similar phrases shall be true and correct in all respects at and as of date of this Agreement and at and as of the Closing Date, except to the extent that such representations and warranties are made as of a specified date, in which case such representations and warranties shall be true and correct in all respects as of the specified date, and the representations and warranties of Seller contained in this Agreement or in any Schedule, certificate or document delivered pursuant to the provisions hereof that are not so qualified shall be true and correct in all material respects at and as of date of this Agreement and at and as of the Closing Date, except to the extent that such representations and warranties are made as of a specified date, in which case such representations and warranties shall be true and correct in all material respects as of the specified date.

- (b) <u>Performance by Seller</u>. Seller and/or the applicable Subsidiary shall have delivered all of the documents required under Section 7.1 and shall have otherwise performed in all material respects all obligations and agreements and complied in all material respects with all covenants required by this Agreement to be performed or complied with by it prior to or at the Closing, including executing the Collateral Agreements.
 - (c) Seller Material Adverse Effect. There shall not have occurred a Seller Material Adverse Effect from the date hereof to the Closing Date.

8.3 Conditions Precedent to Seller's Obligations

The obligations of Seller to effect the Closing of the transactions contemplated hereby are subject to the fulfillment, prior to or at the Closing, of each of the following conditions, any of which may be waived in writing by Seller:

- (a) Representations and Warranties of Buyer True and Correct at Closing. The representations and warranties of Buyer contained in this Agreement or in any certificate delivered pursuant to the provisions of this Agreement that are qualified by the words "material," "material adverse effect" and similar phrases shall be true and correct in all respects at and as of the date of this Agreement and at and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date, except to the extent that such representations and warranties are made as of a specified date, in which case such representations and warranties shall be true and correct in all respects as of the specified date, and the representations and warranties of Buyer contained in this Agreement or in any Schedule, certificate or document delivered pursuant to the provisions hereof or in connection with the transactions contemplated hereby that are not so qualified shall be true and correct in all material respects at and as of the date of this Agreement and at and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date, except to the extent that such representations and warranties are made as of a specified date, in which case such representations and warranties shall be true and correct in all material respects as of the specified date.
- (b) <u>Performance by Buyer</u>. Buyer and/or the applicable Buyer Designee shall have delivered all of the documents required under Section 7.2 and shall have otherwise performed in all material respects all obligations and agreements and complied in all material respects with all covenants required by this Agreement to be performed or complied with by it prior to or at the Closing, including executing the Collateral Agreements.

9. Indemnification

The rights and obligations of Buyer and Seller under this Agreement shall be subject to the following terms and conditions:

9.1 Effect of Investigation

The right to indemnification or other remedy of Buyer or its Affiliates hereunder based on the representations, warranties, covenants and agreements herein will not be affected by any investigation conducted with respect to, or any knowledge acquired (or capable of being

acquired) at any time, by the Buyer or its Affiliates, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy or inaccuracy of or compliance with, any such representation, warranty, covenant or agreement.

9.2 Survival of Representations and Warranties

The representations and warranties of Buyer and Seller contained in this Agreement shall survive the Closing solely for purposes of this Article 9 and such representations and warranties shall terminate at the close of business on the date that is eighteen (18) months after the Closing Date; provided, however, that (i) the representations and warranties in Section 3.7(d) with respect to environmental matters and the representations and warranties relating to Tax and ERISA matters shall survive the Closing and shall terminate at the close of business on the 120th day following the expiration of the applicable statute of limitations with respect to the environmental, Tax or ERISA liabilities in question (giving effect to any waiver, mitigation or extension thereof), (ii) the representations and warranties in Section 3.13 shall terminate at the close of business on the date that is twenty-four (24) months after the Closing Date and (iii) the representations and warranties in Section 3.2 with respect to authorization and Section 3.5(a) with respect to title shall survive indefinitely. Neither Seller nor Buyer shall have any liability whatsoever with respect to any such representations or warranties after the applicable expiration date; provided, however, that, notwithstanding anything herein to the contrary, the obligations of Buyer or Seller to indemnify and hold harmless any Indemnified Party shall not terminate with respect to any claim or right to indemnification as to which such Indemnified Party shall have made in good faith and with reasonable specificity under the circumstances before the applicable expiration date, provided notice to the Indemnifying Party in accordance with this Article 9 and, in such case, such claim or right to indemnification shall survive indefinitely until such claim has been finally resolved.

9.3 General Agreement to Indemnify

(a) Seller and Buyer shall indemnify, defend and hold harmless the other party hereto, and Affiliates thereof, and any director, officer, employee or agent of such other party or Affiliates thereof (each an "Indemnified Party") from and against any and all claims, actions, suits, proceedings, liabilities, obligations, losses, and damages, amounts paid in settlement, interest, costs and expenses (including reasonable attorney's fees, court costs and other out-of-pocket expenses incurred in investigating, preparing or defending the foregoing) (collectively, "Losses") incurred or suffered by any Indemnified Party to the extent that the Losses arise by reason of, or result from (i) subject to Section 9.2, any breach or any failure of any representation or warranty of such party contained in this Agreement or any certificate delivered in connection with this Agreement to have been true when made and at and as of the Closing Date, or (ii) the breach by such party of any covenant or agreement of such party contained in this Agreement to the extent not waived by the other party.

(b) Seller further agrees to indemnify and hold harmless Buyer and Affiliates thereof, and any director, officer, employee or agent of Buyer or Affiliates thereof (each a "Buyer Indemnified Party") from and against any Losses incurred by Buyer or any Buyer Indemnified Party arising out of, resulting from, or relating to: (i) the Excluded Liabilities; (ii) Buyer's waiver of, or noncompliance with, any applicable Bulk Sales Laws; and (iii) any claim, demand or

liability for Taxes relating to, pertaining to, or arising out of the Engenio Business or the Purchased Assets for any Pre-Closing Tax Period.

- (c) Buyer further agrees to indemnify and hold harmless Seller and Affiliates thereof, and any director, officer, employee or agent of Seller or Affiliates thereof (each a "Seller Indemnified Party") from and against any Losses incurred by Seller or any Seller Indemnified Party arising out of, resulting from, or relating to: (i) any failure of Buyer to discharge any of the Assumed Liabilities; and (ii) any claim, demand or liability for Taxes relating to, pertaining to, or arising out of the Engenio Business or the Purchased Assets for any Post-Closing Tax Period.
- (d) Amounts payable in respect of the parties' indemnification obligations shall be treated as an adjustment to the Purchase Price for Tax purposes and shall be treated as such by Buyer and Seller on their Returns to the extent permitted by law. Whether or not the Indemnifying Party (as defined below) chooses to defend or prosecute any Third-Party Claim (as defined below), both parties hereto shall cooperate in the defense or prosecution thereof and shall furnish such records, information and testimony, and attend such conferences, discovery proceedings, hearings, trials and appeals, as may be reasonably requested in connection therewith or as provided in Section 5.1.
- (e) The indemnification obligations of each party hereto under this Article 9 shall inure to the benefit of the directors, officers and Affiliates of the other party hereto on the same terms as are applicable to such other party.
- (f) The Indemnifying Party's liability for all claims made under Section 9.3(a)(i) shall be subject to the following limitations: (i) the Indemnifying Party shall have no liability for such claims until the aggregate amount of the Losses incurred shall exceed five tenths of one percent (0.5%) of the Purchase Price (the "Threshold Amount") in which case the Indemnifying Party shall be liable for the entire amount of such Losses (subject to the limit set forth in clause (ii)) and not just the amount of the Losses that exceeds the Threshold Amount, and (ii) the Indemnifying Party's aggregate liability for all such claims shall not exceed ten percent (10%) of the Purchase Price (the "Cap Amount"); provided, however, in no event shall the limitations in clauses (i) and (ii) of this Section 9.3(f) apply to (a) Losses resulting from fraud or intentional misrepresentation or (b) Losses arising out of breaches of the representations and warranties set forth in Sections 3.3, 3.5(a) and 3.20. The Indemnified Party may not make a claim for indemnification under Section 9.3(a)(i) after the expiration of the applicable survival period specified in Section 9.2; provided, however, that, notwithstanding anything herein to the contrary, so long as such Indemnified Party shall have, before the applicable expiration date, provided notice of a claim (made in good faith and with reasonable specificity under the circumstances) before the applicable expiration date to the Indemnifying Party in accordance with this Article 9, then, in such case, such claim or right to indemnification shall survive indefinitely until such claim has been finally resolved. Notwithstanding anything herein to the contrary, for purposes of this Article 9, all "materiality", "Seller Material Adverse Effect" and similar qualifications in the representations and warranties contained in this Agreement (or contained, incorporated or referenced in any certificate delivered pursuant to this Agreement) shall be disregarded for purposes of calculating the amount of s

contained in this Agreement (or contained, incorporated or referenced in any certificate delivered pursuant to this Agreement) has occurred.

- (g) The indemnification provided in this Article 9 shall be the sole and exclusive remedy after the Closing Date for damages available to the parties to this Agreement for breach of any of the terms, conditions, covenants, representations or warranties contained herein or any right, claim or action arising from the transactions contemplated by this Agreement; <u>provided</u>, <u>however</u>, this exclusive remedy for damages does not preclude a party from bringing an action for (i) specific performance or other equitable remedy to require a party to perform its obligations under this Agreement or any Collateral Agreement or (ii) fraud or intentional misrepresentation.
- (h) Notwithstanding anything contained in this Agreement to the contrary, no party shall be liable to the other party for indirect, special, punitive, exemplary or consequential loss or damage arising out of this Agreement, <u>provided</u>, <u>however</u>, the foregoing shall not be construed to preclude recovery by the Indemnified Party in respect of Losses directly incurred from Third Party Claims. Both parties shall mitigate their damages.
- (i) The rights to indemnification under this Section 9.3 shall not be subject to set-off for any claim by the Indemnifying Party against any Indemnified Party, whether or not arising from the same event giving rise to such Indemnified Party's claim for indemnification.

9.4 General Procedures for Indemnification

(a) The Indemnified Party seeking indemnification under this Agreement shall promptly notify the party against whom indemnification is sought (the "Indemnifying Party") of the assertion of any claim, or the commencement of any action, suit or proceeding by any Third Party, in respect of which indemnity may be sought hereunder and shall give the Indemnifying Party such information with respect thereto as the Indemnifying Party may reasonably request, but failure to give such notice shall not relieve the Indemnifying Party of any liability hereunder (unless and to the extent that the Indemnifying Party has suffered material prejudice by such failure). If the Indemnifying Party acknowledges in writing its obligation to indemnify the Indemnified Party, then the Indemnifying Party shall have the right, but not the obligation, exercisable by written notice to the Indemnified Party within twenty (20) days of receipt of notice from the Indemnified Party of the commencement of or assertion of any claim, action, suit or proceeding by a Third Party in respect of which indemnity may be sought hereunder (a "Third-Party Claim"), to assume the defense of such Third-Party Claim only if such Third-Party Claim (i) involves (and continues to involve) solely money damages or (ii) involves (and continues to involve) claims for both money damages and equitable relief against the Indemnified Party that cannot be severed, where the claims for money damages are the primary claims asserted by the Third Party and the claims for equitable relief are incidental to the claims for money damages; provided further, that in the case where Seller is the Indemnifying Party, as additional requirements in addition to the foregoing, the Seller shall only be permitted to assume the defense of such Third-Party Claim in the event that the monetary damages that are sought (or that would reasonably be expected to be sought) in connection with such Third-Party Claim do not exceed the Cap Amount or such lesser remaining amount after deducting therefrom the

this Article 9. If the Indemnifying Party has not acknowledged in writing its obligation to indemnify the Indemnified Party, then the Indemnified Party shall have the right to assume and control the defense against such Third Party Claim. In the event that any party exercises its right to undertake any such defense against any Third Party Claim as provided above, then the other parties shall cooperate in such defense and make available at such cooperating party expense all witnesses, pertinent records, materials and information in such party's possession and control relating thereto as is reasonably required to by the party conducting the defense.

- (b) The Indemnifying Party or the Indemnified Party, as the case may be, shall have the right to participate in (but not control), at its own expense, the defense of any Third-Party Claim that the other is defending, as provided in this Agreement.
- (c) The Indemnifying Party, if it has assumed the defense of any Third-Party Claim as provided in this Agreement, shall not consent to a settlement of, or the entry of any judgment arising from, any such Third-Party Claim without the Indemnified Party's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed) unless such settlement or judgment relates solely to monetary damages and, in addition, in the case where Seller is the Indemnifying Party, such monetary damages do not exceed the Cap Amount or such lesser remaining amount after deducting therefrom the amount of all other previously paid and outstanding unpaid and/or unresolved claims pursuant to this Article 9. The Indemnifying Party shall not, without the Indemnified Party's prior written consent, enter into any compromise or settlement that (i) commits the Indemnified Party to take, or to forbear to take, any action, or (ii) does not provide for a complete release by such Third Party of the Indemnified Party. The Indemnified Party shall have the sole and exclusive right to settle any Third-Party Claim, on such terms and conditions as it deems reasonably appropriate, to the extent such Third-Party Claim involves equitable or other non-monetary relief against the Indemnified Party, and shall have the right to settle any Third-Party Claim involving money damages for which the Indemnifying Party has not assumed the defense pursuant to this Section 9.4 with the written consent of the Indemnifying Party, which consent shall not be unreasonably withheld, conditioned or delayed.
- (d) In the event an Indemnified Party shall claim a right to payment pursuant to this Agreement, such Indemnified Party shall send written notice of such claim to the Indemnifying Party; but failure to give such notice shall not relieve the Indemnifying Party of any liability hereunder (unless and to the extent that the Indemnifying Party has suffered material prejudice by such failure). Such notice shall specify the basis for such claim, the amount thereof, if known, and the method of computation thereof, all with reasonable particularity and shall contain a reference to the provisions of this Agreement in respect of which such a claim shall be incurred. Such notice shall be given promptly after the Indemnified Party becomes aware of the basis for each such a claim. The Indemnifying Party shall, within thirty (30) days after receipt of such notice of an indemnified Loss, and subject to the limitations set forth in Section 9.3, (i) pay or cause to be paid to the Indemnified Party the amount of such Loss specified in such notice which the Indemnifying Party does not contest, or (ii) notify the Indemnified Party if it wishes to contest the existence or amount of part or all of such a Loss by stating with particularity the basis upon which it contests the existence or amount thereof.

10. Miscellaneous Provisions

10.1 Notices

All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given upon receipt if (i) mailed by certified or registered mail, return receipt requested, (ii) sent by Federal Express or other express carrier, fee prepaid, (iii) sent via facsimile with receipt confirmed, or (iv) delivered personally, addressed as follows or to such other address or addresses of which the respective party shall have notified the other.

If to Seller, to: LSI Corporation

Attn: Chief Financial Officer

1621 Barber Lane Milpitas, CA 18109 United States of America Facsimile: (408) 433-7306

With a copy to: LSI Corporation

Attn: Vice President — Law 1110 American Parkway NE Allentown, PA 18109 United States of America Facsimile: (610) 712-5712

If to Buyer, to: NetApp, Inc.

Attention: General Counsel 495 East Java Drive Sunnyvale, CA 94089 United States of America Facsimile: (408) 822-4501

With a copy to: Latham & Watkins LLP

140 Scott Drive

Menlo Park, California 94025 United States of America

Attention: Peter Kerman and Tad Freese

Facsimile: (650) 463-2600

10.2 Expenses

Except as otherwise provided herein, each party to this Agreement will bear all of the fees, costs and expenses incurred by it in connection with the transactions contemplated hereby, whether or not such transactions are consummated.

10.3 Entire Agreement; Modification

The agreement of the parties, which consists of this Agreement, the Schedules and Exhibits hereto and the documents referred to herein, sets forth the entire agreement and

understanding between the parties and supersedes any prior agreement or understanding, written or oral, relating to the subject matter of this Agreement, including the Confidentiality Agreement. No amendment, supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the party to be bound thereby, and in accordance with Section 11.3.

10.4 Assignment; Binding Effect; Severability

This Agreement may not be assigned by any party hereto without the other party's written consent; provided, that Buyer may transfer or assign in whole or in part to one or more Buyer Designee its right to purchase all or a portion of the Purchased Assets, but no such transfer or assignment will relieve Buyer of its obligations hereunder. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the successors, legal representatives and permitted assigns of each party hereto. The provisions of this Agreement are severable, and in the event that any one or more provisions are deemed illegal or unenforceable the remaining provisions shall remain in full force and effect unless the deletion of such provision shall cause this Agreement to become materially adverse to either party, in which event the parties shall use reasonable commercial efforts to arrive at an accommodation that best preserves for the parties the benefits and obligations of the offending provision.

10.5 Governing Law

THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS PERFORMED ENTIRELY WITHIN THAT STATE, IRRESPECTIVE OF THE CHOICE OF LAWS PRINCIPLES OF THE STATE OF NEW YORK, AS TO ALL MATTERS, INCLUDING MATTERS OF VALIDITY, CONSTRUCTION, EFFECT, ENFORCEABILITY, PERFORMANCE AND REMEDIES.

10.6 Consent to Jurisdiction

Each of Buyer and Seller irrevocably submits to the exclusive jurisdiction of (i) the Supreme Court of the State of New York, New York County, and (ii) the United States District Court for the Southern District of New York, for the purposes of any suit, action or other proceeding arising out of this Agreement or any transaction contemplated hereby (and each agrees that no such action, suit or proceeding relating to this Agreement or any transaction contemplated hereby shall be brought by it or any of its Affiliates except in such courts). Buyer further agrees, and Seller further agrees, that service of any process, summons, notice or document by U.S. registered mail to such Person's respective address set forth in Section 10.1 above shall be effective service of process for any action, suit or proceeding in New York with respect to any matters to which it has submitted to jurisdiction as set forth above in the immediately preceding sentence. Each of Buyer and Seller irrevocably and unconditionally waives (and agrees not to plead or claim), any objection to the laying of venue of any action, suit or proceeding arising out of this Agreement or the transactions contemplated hereby in (i) the Supreme Court of the State of New York, New York County, or (ii) the United States District Court for the Southern District of New York or that any such action, suit or proceeding brought in any such court has been brought in an inconvenient forum.

10.7 Waiver of Jury Trial

Each party hereby waives, and agrees to cause each of its Affiliates to waive, to the fullest extent permitted by applicable Law, any right it may have to a trial by jury in respect of any litigation directly or indirectly arising out of, under or in connection with this Agreement. Each party (i) certifies that no representative of any other party has represented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing waiver and (ii) acknowledges that it and the other parties hereto have been induced to enter into this Agreement by, among other things, the mutual waivers and certifications in this Section 10.7.

10.8 Execution in Counterparts

This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

10.9 Public Announcement

Prior to the signing of this Agreement, Seller and Buyer shall prepare a mutually agreeable release announcing the transaction contemplated hereby. Except for such press release, neither Seller nor Buyer shall, without the approval of the other, make any press release or other announcement concerning the existence of this Agreement or the terms of the transactions contemplated by this Agreement, except as and to the extent that any such party shall be so obligated by Law, in which case the other party shall be advised and the parties shall use their reasonable commercial efforts to cause a mutually agreeable release or announcement to be issued; provided, however, that the foregoing shall not preclude, or require notification to the other Party of, communications or disclosures necessary to comply with accounting, stock exchange or federal securities or labor relations Law disclosure obligations.

10.10 No Third-Party Beneficiaries

Nothing in this Agreement, express or implied, is intended to or shall (a) confer on any Person other than the Parties and their respective successors or assigns (other than the right to indemnification of Indemnified Parties hereunder) any rights (including Third-Party beneficiary rights), remedies, obligations or liabilities under or by reason of this Agreement or (b) constitute the parties hereto as partners or as participants in a joint venture. This Agreement shall not provide Third Parties with any remedy, claim, liability, reimbursement, cause of action or other right in excess of those existing without reference to the terms of this Agreement. Nothing in this Agreement shall be construed as giving to any Business Employee, or any other individual, any right or entitlement to employment or continued employment or any right or entitlement under any Benefit Plan, policy or procedure maintained by Seller, except as expressly provided in such Benefit Plan, policy or procedure. No Third Party shall have any rights under Section 502, 503 or 504 of ERISA, any comparable applicable Law of any other jurisdiction, or any regulations thereunder because of this Agreement that would not otherwise exist without reference to this Agreement. No Third Party shall have any right, independent of any right that exists irrespective of this Agreement, under or granted by this Agreement, to bring any suit at Law or equity for any matter governed by or subject to the provisions of this Agreement.

11. Termination; Amendment and Waiver

11.1 Termination

This Agreement may be terminated at any time prior to the Closing Date by:

- (a) Mutual Consent. The mutual written consent of Buyer and Seller;
- (b) <u>Failure of Buyer Condition</u>. Buyer upon written notice to Seller if any of the conditions to the Closing set forth in Section 8.2 shall have become incapable of fulfillment by the Termination Date and shall not have been waived in writing by Buyer;
- (c) <u>Failure of Seller Condition</u>. Seller upon written notice to Buyer if any of the conditions to the Closing set forth in Section 8.3 shall have become incapable of fulfillment by the Termination Date and shall not have been waived in writing by Seller;
- (d) <u>Court or Administrative Order</u>. Buyer or Seller if (i) there shall be in effect a final, non-appealable order of a court or government administrative agency of competent jurisdiction prohibiting the consummation of the transactions contemplated hereby or (ii) if there shall be any Law that makes consummation of the transactions contemplated hereby illegal or otherwise prohibited; or
- (e) <u>Delay</u>. Buyer or Seller if the Closing shall not have occurred by July 31, 2011 (the "<u>Termination Date</u>"); <u>provided</u>, <u>however</u>, that the party seeking termination pursuant to clause (b), (c) or (e) is not then in breach in any material respect of any of its representations, warranties, covenants or agreements contained in this Agreement.

11.2 Effect of Termination

In the event of the termination of this Agreement in accordance with Section 11.1, this Agreement shall become void and have no effect, without any liability on the part of any party or its directors, officers or stockholders, except for the obligations of the parties hereto as provided in Article 6 relating to the obligations of Buyer and Seller to keep confidential certain information, Section 10.2 relating to certain expenses, Section 10.9 relating to publicity and this Section 11.2. Nothing in this Section 11.2 shall be deemed to release either party from any liability for any knowing, willful and material breach of any representation, warranty, covenant or agreement hereunder prior to termination.

11.3 Amendment and Waiver

The Agreement may be amended with respect to any provision contained herein at any time prior to the Closing Date by action of the parties hereto taken by their Boards of Directors or by their duly authorized officers or employees, whether before or after such party's action. Any term or condition hereof may be waived and at any time prior to the Closing Date by the party hereto which is entitled to the benefits thereof by action taken by its Board of Directors or its duly authorized officer or employee, whether before or after the action of such party. Any such amendment or waiver shall be evidenced by a written instrument duly executed on behalf of each party by its duly authorized officer or employee. The failure of either party to enforce at

any time any provision of this Agreement shall not be construed to be a waiver of such provision nor shall it in any way affect the validity of this Agreement or the right of such party thereafter to enforce each and every such provision. No waiver of any breach of this Agreement shall be held to constitute a waiver of any other or subsequent breach.

IN WITNESS WHEREOF, each party has caused this Agreement to be duly executed on its behalf by its duly authorized officer as of the date first written above.

LSI CORPORATION

By: /s/ Abhi Talwalkar

Name: Abhi Talwalkar

Title: President and Chief Executive Officer

NETAPP, INC.

By: /s/ Thomas Georgens

Name: Thomas Georgens

Title: President and Chief Executive Officer

-69-

CERTIFICATION OF CHIEF EXECUTIVE OFFICER AS ADOPTED PURSUANT TO SECTION 302(a) OF THE SARBANES-OXLEY ACT OF 2002

I, Abhijit Y. Talwalkar, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of LSI Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 12, 2011

By: /s/ Abhijit Y. Talwalkar

Name: Abhijit Y. Talwalkar

Title: President and Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER AS ADOPTED PURSUANT TO SECTION 302(a) OF THE SARBANES-OXLEY ACT OF 2002

I, Bryon Look, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of LSI Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 12, 2011

By: /s/ Bryon Look

Name: Bryon Look

Title: Executive Vice President, Chief Financial Officer and Chief Administrative Officer

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Abhijit Y. Talwalkar, certify pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of LSI Corporation on Form 10-Q for the quarterly period ended April 3, 2011 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of LSI Corporation.

By: /s/ Abhijit Y. Talwalkar

Name: Abhijit Y. Talwalkar

Title: President and Chief Executive Officer

Date: May 12, 2011

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Bryon Look, certify pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of LSI Corporation on Form 10-Q for the quarterly period ended April 3, 2011 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of LSI Corporation.

By: /s/ Bryon Look

Name: Bryon Look

Title: Executive Vice President, Chief Financial Officer

and Chief Administrative Officer

Date: May 12, 2011